ON THE DEVELOPMENTAL PROFILE OF HORMONES IN SPODOPTERA MAURITIA BOISD. (LEPIDOPTERA : NOCTUIDAE)

Thesis submitted to the University of Calicut in partial fulfilment of the requirements for the Degree of DOCTOR OF PHILOSOPHY in Zoology

MONA. P.M., M.Sc.

DEPARTMENT OF ZOOLOGY, UNIVERSITY OF CALICUT KERALA, 673 635, INDIA

DEPARTMENT OF ZOOLOGY

UNIVERSITY OF CALICUT

Fax : 0494 400269 Grams : UNICAL E-mail : nair@unical.ac.in

Dr. V. S. K. NAIR, Ph.D. PROFESSOR



Off : 0494 401144 Phone Extn.604,419,420 Resi : 0494 400031

CALICUT UNIVERSITY P. O 673 635, KERALA, INDIA

CERTIFICATE

This is to certify that this is an authentic record of the research work carried out by Mrs. P.M. Mona, M.Sc. from October, 1997 to August, 2001 as a part-time student in partial fulfilment of the requirements for the Degree of DOCTOR OF PHILOSOPHY under the faculty of Science of the University of Calicut under my supervision and guidance. No part of this thesis has been presented before for any other degree. I also certify that she has passed the Ph.D. qualifying examination of the University of Calicut, held in December, 1999.

Φρηγημα, αυτοπογιστικό τηλός 36, 244 Ω. αυτοπο ματοπογιστικό τη μαριστικό καγ μαριστικό την προσφοριας Κημαριστικό την Ναθέδ.

.8.01

(V.S. KRISHNAN NAIR)

c.°

DECLARATION

I do hereby declare that the present work is original and it has not previously formed the basis for the award of any degree or diploma.

Calicut University, 13 August, 2001.

MONA P.M.

ACKNOWLEDGEMENTS

I wish to express my gratitude and indebtedness to Dr.V.S. Krishnan Nair, Professor, Department of Zoology, for his supervision and invaluable guidance throughout the period of my work.

Funding for this study has been provided by the United States Department of Agriculture (USDA), Washington in the form of a major research project sanctioned to Dr. V.S. Krishnan Nair. I worked as a Junior Research Fellow in this project. I thank USDA and Dr. Krishnan Nair for allowing me to present the data collected during the investigation for my Ph.D. thesis.

I am equally indebted to Dr. Dale B. Gelman (Collaborating Scientist), Insect Physiologist, USDA, BARC-EAST, Beltsville, Maryland, U.S.A. for helping me in the radioimmunoassay (RIA) of tissue samples. I am also obliged to Dr. Gelman for the stimulating discussions and constructive criticisms at each and every stage of this study.

I record my sincere thanks to Dr. U.V.K. Mohamed, Professor and Head of the Department of Zoology, University of Calicut for providing the necessary laboratory facilities.

I am also grateful to Dr. N. Raju and his colleagues of the Department of Statistics, University of Calicut for helping me in the statistical calculations.

I express my sincere thanks to my colleagues Dr. T.M. Benny, Miss E.M. Manogem, Miss A. Sindhu, Mr. K. Safarulla and Mrs. K.V. Nisha Suresh Babu for their help and co-operation at various phases of the present work. I would also like to thank Mr. V. Lohithakshan, our laboratory assistant, for his help in the rearing of the insect colony and Mr. K. Balu for carefully typing the manuscript.

Mona P.M.

CONTENTS

CHAPTER 1. GENERAL INTRODUCTION

1.	Insect hormones : A review				
1.1	Prothoracicotropic hormone (PTTH)				
1.1.1	Control of PTTH release				
1.2	The moulting hormone				
1.2.1	Ecdysteroid titres during postembryonic development				
1.2.2	Factors	regulating the secretory activity of prothoracic glands	11		
1.3	Juvenile hormones (JHs)				
1.3.1	Control of CA activity				
1.4	Testes as a source of ecdysteroids				
1.4.1	Ecdysteroid titres of the testes in relation to				
	haemoly	ymph titres	27		
1.4.2	Factors	regulating synthesis of testicular ecdysteroids	28		
1.4.3	Insect growth regulators				
1.5	Objectives of the investigation				
CHAF	TER 2	MATERIALS AND METHODS			
2.1	Pest sta	tus of S. mauritia	37		
2.2	Rearing	; and maintenance of the larvae of <i>S. mauritia</i>	38		
2.3	Biology of S. mauritia				
2.4	Experimental animals				
2.5	General experimental techniques				
CHAI	CHAPTER 3 DEVELOPMENTAL PROFILE OF HORMONES DURING METAMORPHOSIS AND THE EFFECTS OF HORMONE MIMICS				
3.1	Introdu	ction	51		
3.2	Materia	ls and Methods	54		
3.3	Results				
3.3.1	Haemol	ymph ecdysteroid titres of larvae, pupae and	55		
227	Dotormi	ination of head critical period (HCP) during			
J.J.Z	larval	unal transformation	58		
222	Effocte (of ligations of sixth instar larvae on metamorphosis			
9.9.9	and haemolymph ecdysteroid titres				
	and hat	morympri cenyserom mues			

3.3.4	Effects of treatments of JHA on metamorphosis and haemolymph ecdysteroid titres			
225	Effects of treatments of adductoreid a seriet DLI 5000 or			
5.5.5	Effects of freatments of ecuysteroid agonist RH 5992 on			
	metam	orphosis and naemolymph ecdysteroid fitres	69	
3.4	Discussion			
3.5	Summary			
CHAI	TER 4	DEVELOPMENTAL PROFILE OF TESTICULAR ECDYSTEROIDS AND THE EFFECTS OF HORMONE MIMICS		
4.1	Introdu	iction	113	
4.2	Materials and Methods			
4.3	Results	6	116	
4.3.1	Testicu	lar ecdysteroid titres during larval-pupal-adult		
	develop	oment	116	
4.3.2	Effects of ligations of sixth instar larvae on testicular			
	ecdyste	roid titres	118	
4.3.3	Effects	of treatments of JHA on testicular ecdysteroid titres	121	
4.3.4	Effects	of treatments of ecdysteroid agonist RH 5992 on		
	testicul	ar ecdysteroid titres	126	
4.3.5	Comparison of testicular and haemolymph ecdysteroid			
	titres du	uring larval-pupal and pupal-adult development	131	
4.4	Discuss	Discussion		
4.5	Summary			
REFEI	RENCES		163	

CHAPTER 1

GENERAL INTRODUCTION

1. INSECT HORMONES: A REVIEW

Insect metamorphosis is typified by abrupt morphological transitions which are in reality smooth continua of precisely regulated events controlled by titres of several hormones. The endocrine nature of the processes and factors responsible for the regulation of moulting and metamorphosis have been understood only within the last few decades. Based on the pioneering studies by Kopec (1917, 1922); Wigglesworth (1934, 1936, 1940); Fukuda (1940, 1944); Williams (1948, 1952) and others, a basic model for the endocrine control of insect postembryonic development emerged by early 1950s. According to this basic model, the brain in response to appropriate environmental stimuli synthesises and releases via neurohaemal organs (corpora cardiaca or corpora allata) the brain hormone. This brain hormone now referred to as prothoracicotropic hormone (PTTH), stimulates the prothoracic glands or analogous structures (ring gland, ventral glands) to synthesise and secrete moulting hormone (MH), which initiates the events generically referred to as moult. However the character of the moult, i.e., whether it is larval-larval, larval-pupal or pupal-adult, is dictated by the relative titre of juvenile hormone (JH) which is synthesised by the corpora allata (CA) and released at or before the moult. The basic model hypothesised that presence of a high titre of JH would result in a larval- larval moult,

whereas presence of reduced titre of JH would result in larval-pupal moult. Absence of JH would result in a pupal-adult moult.

Although earlier studies demonstrated that exogenous ecdysone administered to insects could initiate moulting and metamorphosis (Karlson, 1980), one of the implicit assumptions of the basic model that MH titres fluctuated during insect postembryonic development, was not supported by direct experimental evidence until the early to mid-1960s. At this time Burdette (1962) and Shaaya and Karlson (1965a,b) using the *Calliphora* bioassay, succeeded in demonstrating that MH titres varied during larvalpupal-adult development of *Calliphora erythrocephala* and *Bombyx mori*. In accordance with the basic model, their data suggested that an increase or peak in ecdysteroid activity preceded or occurred concomitantly with the moult.

1.1. PROTHORACICOTROPIC HORMONE (PTTH)

Site of synthesis

PTTH is synthesised primarily by four large neurosecretory cells (the prothoracicotropes), two per hemisphere, whose soma reside in the dorsolateral region of the brain (Agui *et al.*, 1979; Mizoguchi *et al.*, 1990). PTTH is then transported down the axons of these neurons to the corpora

cardiaca in most of the insects whereas in lepidopterans, the axons terminate in corpora allata which serve as neurohaemal organs.

Chemistry

It has been confirmed that PTTH is a protein or peptide. Purification of PTTH from brains of *Manduca sexta* established the presence of at least one PTTH of approximately 22,000 MW and possibly a second hormone of approximately 1,000 MW (Bollenbacher and Gilbert, 1981; Gilbert *et al.*, 1981). In *Bombyx*, PTTH exists as several peptides (Ishizaki and Ichikawa, 1967). Kawakami *et al.* (1990) report that in *Bombyx* larva, the PTTH is first synthesised as a 224 amino acid polypeptide containing 109 amino acid PTTH subunits. Later studies confirmed that the mature, secreted PTTH is a homodimer containing inter and intra monomer disulphide bonds, the latter being necessary for retention of prothoracicotropic activities (Ishibashi *et al.*, 1994).

PTTH synthesis and release

At specific times during development, PTTH is released into the haemolymph (Agui *et al.*, 1980). A preliminary titre of PTTH in the haemolymph of fourth and fifth (last) instar larvae of *Manduca sexta* has been determined using the *in vitro* PTTH assay (Gilbert *et al.*, 1981). The data from this study demonstrates that the haemolymph titre of PTTH is elevated

just prior to, or concomitant with the increase in the haemolymph titre of ecdysteroids. These periods of elevated PTTH titre in the haemolymph is found to correspond temporally to the time of PTTH release (head critical periods,HCPs) as predicted by classical head ligation studies. However, direct measurement of haemolymph PTTH titres in fifth (last) instar *Bombyx* larvae shows that PTTH is released every day (Mizoguchi, 1995).

In many lepidopterans, the release of PTTH is a gated event, occurring only at certain times during the light-dark cycle. In *Manduca* and *Antheraea*, PTTH release is controlled by a light-sensitive circadian pacemaker (Truman and Riddiford, 1974; Fujishita and Ishizaki, 1982). PTTH release occurs in the larvae which have attained a critical weight before or during the time of photoperiodically determined gate that is "open" only once in every 24 hours.

One of the most important events in the timing of PTTH release during larval-pupal development is that the JH titre, levels off to an undetectable level at the halfway stage of feeding in the last instar while it remains at a detectable level in the penultimate instar prior to ecdysis. This pattern of changes in JH titre has been described in *M. sexta* (Nijhout and Williams, 1974; Fain and Riddiford, 1975; Sedlak *et al.*, 1983), *B. mori* (Sakurai, 1983) and *Trichoplusia ni* (Jones *et al.*, 1986). In *M. sexta* the precommitment drop in the JH titre is permissive for the gated release of PTTH (Rountree and Bollenbacher, 1986).

1.1.1. Control of PTTH release

The mechanism of control of PTTH release varies among the different insects. Both extrinsic and intrinsic stimuli are involved and the mediators of these stimuli can be either neural and/or endocrine.

Proprioceptive and mechanoreceptive stimuli

Studies on the mechanisms by which nervous stimuli controls PTTH release led to establishing the timing of PTTH release in insects. In Rhodnius prolixus, proprioceptive stimuli controlled PTTH release. Proprioception being a nervous phenomenon, the effect of abdominal stretching appears to be the cause for the release of PTTH from the brain (Wigglesworth, 1933). Stretch stimulated moulting mediated by proprioception also was demonstrated in Locusta migratoria (Clarke and Langley, 1963) and Oncopeltus fasciatus (Nijhout, 1979). In Manduca and other lepidopterans, proprioceptive control of PTTH release involves a form of allometry; i.e., PTTH release occurs when the brain receives information that the body has attained a certain critical weight (Nijhout, 1975). In Manduca, allometry may be controlling PTTH release by inactivation of the CA (Nijhout and Williams, 1974).

 $\mathbf{5}$

Photoperiodic stimuli

Photoperiodicity is a major source of external control of insect development and its apparently direct effect on the release of PTTH is expressed in either a circadian or seasonal manner. In *Manduca*, release of PTTH is photosensitive during larval development upto the second HCP of the last larval instar (Truman, 1972; Truman and Riddiford, 1974). Photoperiod does not directly control PTTH release, but acts as a primary oscillator, establishing a gate or time of day, when release can occur. The effect of seasonal photoperiodism on PTTH release is a well-established phenomenon and cues the induction and duration of diapause in insects. It is suggested that the photoreceptor and the photoperiodic mechanism in general may reside in the lateral NSC (Bollenbacher and Bowen, 1983).

Temperature stimuli

Temperature is an important external cue that regulates PTTH release, and this effect is expressed either in a circadian or seasonal context. Temperature can either act alone or in concert with a particular photoperiod, and can affect the rate of development and the induction, reversal or termination of diapause (Tauber and Tauber, 1976; Beck, 1980).

Endocrine regulators

Endocrine systems function as the chemical mediators for the environmental stimuli controlling development. In several insect species, JH appears to be a feedback inhibitor of PTTH release (Fukaya and Mitsuhashi, 1961; Fukaya and Kobayashi, 1966; Yagi and Fukaya, 1974; Hiruma *et al.*, 1978), inducing and/or maintaining larval diapause (Yin and Chippendale, 1973; Takeda, 1978; Chippendale and Yin, 1979; Sieber and Benz, 1980; Chippendale and Turunen, 1981). Based on observations of the hormonal effects of 20-hydroxyecdysone on cerebral NSC, ecdysteroids may also regulate the release and/or synthesis of PTTH (Marks *et al.*, 1972; Steel, 1975; Agui and Hiruma, 1977a,b). The control of PTTH release is thus the cumulative effect of a complex and interrelated series of environmental and proprioceptive-mechanoreceptive stimuli acting *via* the nervous and endocrine systems.

1.2. THE MOULTING HORMONE

Site of synthesis

Ecdysteroids are primarily secreted by the prothoracic glands. These organs do not store significant amounts of ecdysteroids (Agui *et al.*, 1972; Chino *et al.*, 1974; King *et al.*, 1974; Bollenbacher *et al.*, 1976). Prothoracic glands from several species were cultured *in vitro* and the gland products

secreted into the culture medium were characterised and quantified by radioimmunoassay, bioassay and physico-chemical techniques. The results of these experiments established that the larval prothoracic glands of *Manduca sexta* (King *et al.*, 1974), *Bombyx mori* (Chino *et al.*, 1974), *Leucophaea maderae* (Borst and Engelmann, 1974) and Sarcophaga bullata (Bollenbacher *et al.*, 1976) synthesise and secrete ecdysone. Other tissues such as oenocytes, epidermis, gut, ovary and testis also seem to be the secondary sources of ecdysteroids (Delbecque *et al.*, 1990).

Chemistry

The first two moulting hormones have been isolated from Bombyx pupae (Butenandt and Karlson, 1954; Karlson, 1956) and are now designated as ecdysone and 20-hydroxyecdysone respectively. Both MHs proved to be polyhydroxylated ketosteroids generically referred to as ecdysteroids (Goodwin *et al.*, 1978). Ecdysone is secreted as a prohormone (Granger and Bollenbacher, 1981) which undergoes hydroxylation into the actual moulting hormone, 20-hydroxyecdysone. This process is catalysed by ecdysone 20-monoxygenase, a cytochrome P-450 mixed function oxidase present in tissues peripheral to the prothoracic glands (Smith *et al.*, 1980). Ecdysteroids possess the full C₂₇ carbon skeleton of cholesterol, exhibit a cis fusion of the A and B ring, have two functional moieties in the B ring and posses hydroxyl groups at several positions. To date, 20 or so ecdysteroids have been tentatively identified in insects or crustaceans. In addition to these zooecdysteroids, atleast 60 structurally distinct ecdysteroids (phytoecdysteroids) have been isolated from numerous phytogenetically diverse families of plants, in most cases at much higher concentrations than in arthropods (Hetru and Horn, 1980).

1.2.1 Ecdysteroids titres during postembryonic development

Ecdysteroid titres have been measured in numerous species of insects during various stages of postembryonic development. The ecdysteroid titre appears to fluctuate considerably during postembryonic development, varying from basal levels (~ 10^{-8} M) by a factor of from 20 to 30 (e.g. in *Drosophila melanogaster*) upto approximately 10^2 - 10^3 (e.g. in *Manduca* and *Schistocerca gregaria*, Dean *et al.*, 1980; Richards, 1981).

During larval-larval and larval-pupal development (larval-adult in hemimetabolous insects), the major peak in the ecdysteroid titre is usually present during the latter half of each stadium. Temporally and quantitatively, this titre peak is characterised by a rapid rise from basal levels before apolysis; maximal levels either slightly before, at, or slightly after apolysis; and an equally rapid drop back to low or basal levels by the time of ecdysis. The peaks preceding larval-pupal and larval-adult metamorphic moults are of slightly greater magnitude and/or duration than peaks preceding the non-metamorphic larval-larval moults.

During larval-larval development, there appears only one surge of ecdysteroid titre just preceding each ecdysis in Galleria mellonella (Sehnal et al., 1981, 1986), Manduca sexta (Bollenbacher et al., 1981), Trichoplusia ni (Jones et al., 1981) and Mamestra brassicae (Agui and Hiruma, 1982). During larval-pupal transformation of Lepidoptera, the existence of two ecdysteroid peaks is well established (Lafont et al., 1977b; Maroy and Tarnoy, 1978; Dean et al., 1980; Bollenbacher et al., 1981; Zimowska et al., 1985; Jones et al., 1986). In contrast to larval-larval and larval-pupal development, during pupal-adult development the major ecdysteroid titre peak occurs in the first half or middle of the pupal stadium (Dean et al., 1980). This titre peak is characterised by a slow rise from basal levels with the onset of pupal-adult apolysis. The rise continues over an extended period during pharate-adult development, followed by a slow return to basal levels before pupal-adult ecdysis. The major ecdysteroid titre peak in pupal stadia not only differs in temporal aspects from those which occur in larval stadia but is also much greater in magnitude and duration. These major peaks in the ecdysteroid titres are responsible for initiating and/or maintaining the events associated with moulting and the pharate development which occurs during the moult, i.e., between apolysis and ecdysis. The single major ecdysteroid titre peak detected during each larval and pupal stadium is considered a "moulting peak".

In addition to the major moulting peaks, there may be one or more additional "non-moulting peaks" in the ecdysteroid titre during certain stadia, particularly the stadium leading to the first metamorphic moult (Dean *et al.*, 1980). In contrast to the major moulting peaks, these non-moulting peaks are usually present during the intermoult period and are much more subtle in magnitude and duration than the larval moulting peaks. Some of the intermoult peaks have been found to correlate temporally with activities such as DNA synthesis, RNA synthesis, lipid synthesis and organelle formations but no firm conclusions can be drawn at the cellular level (Dean *et al.*, 1980). In *Manduca sexta* a subtle intermoult ecdysteroid titre peak has been detected about day 4 of the last larval stadium. This ecdysteroid peak and comparable intermoult peaks reported for other holometabolous insects are thought to be responsible for reprogramming the epidermis from a larval to a pupal commitment (Dean *et al.*, 1980; Riddiford, 1985).

1.2.2 Factors regulating the secretory activity of prothoracic glands

The factors which regulate the production and secretion of ecdysone by the prothoracic glands are important regulators of ecdysteroid titre. Since prothoracic glands have been shown to release ecdysone immediately following its synthesis, the factors which regulate gland activity do so at the level of ecdysone biosynthesis. The principal regulator of prothoracic gland activity is the cerebral neuropeptide, the PTTH. In addition to the PTTH, other factors also influence the rate of ecdysone biosynthesis either by indirectly affecting the synthesis and release of PTTH or by directly acting on the prothoracic glands. Such factors include environmental and neural stimuli, juvenile hormone, haemolymph proteins and ecdysteroids.

Prothoracicotropic hormone

The principal regulators of prothoracic gland activity is the PTTH from the brain (Bollenbacher and Bowen, 1983; Bollenbacher and Granger, 1985). Utilizing *in vitro* assay, it has been demonstrated that PTTH from the brain directly activates the prothoracic glands of *Manduca sexta* by eliciting an *increase* in the basal rate of ecdysone biosynthesis (Bollenbacher *et al.*, 1979). This activation was found to be both specific and dose-dependent. PTTH acts similarly under *in vivo* conditions and it is one of the primary physiological regulators of prothoracic gland activity.

By analogy with vertebrate steroidogenic peptides such as adrenocorticotropic hormones and leutinizing hormone, PTTH has been hypothesized to exert its effect through the action of a cyclic AMP (cAMP) as second messenger (Gilbert *et al.*, 1980). An involvement of cAMP in steroidogenesis by the insect prothoracic glands was indicated initially by the ability of phosphodiesterase inhibitors to stimulate ecdysone synthesis by the prothoracic glands of *Manduca sexta* (Vedeckis *et al.*, 1976) and to mimic neurohormone mediated changes, in membrane potential of the prothoracic glands of *Galleria mellonella* (Gersch and Birkenbeil, 1980).

Environmental factors

Environmental factors influence PTTH synthesis and release and it is known that a light sensitive circadian clock is involved in the release of PTTH (Truman and Riddiford, 1974; Gelman and Hayes, 1982; Denlinger, 1985). Possibility exists that some environmental factors may also act directly on the prothoracic glands. A study with *Samia cynthia ricini* larvae has suggested, based on transplantation experiments that in addition to light sensitive cephalic circadian clock for the release of PTTH, these organisms also possess a light-sensitive prothoracic gland clock for the release of ecdysone (Mizoguchi and Ishizaki, 1982).

Neural factors

Neural regulation of ecdysone biosynthesis is either indirectly via regulation of PTTH or directly on the prothoracic glands. Studies with R. prolixus and L. migratoria have implicated that PTTH synthesis/release may be under direct neural regulation (Steel, 1978; Girardie and De Reggi, 1978; Steel and Davey, 1985). Although proof for such neural regulation has not been demonstrated, it is known that the prothoracic glands are innervated by various neural processes, including neurosecretory axons from several ventral

ganglia (Benedeczky *et al.*, 1980; Granger and Bollenbacher, 1981). In *G. mellonella* larvae, the prothoracic glands are known to be innervated by the prothoracic, mesothoracic and suboesophageal ganglia (Singh and Sehnal, 1979) and all three of these ganglia have been reported to modify prothoracic gland secretory activity either when implanted into decapitated larvae with prothoracic glands *in situ* (Mala *et al.*, 1977) or when incubated *in vitro* with prothoracic glands (Gersch, 1978).

Inhibitory and stimulatory effects of JH

Both older and more recent results suggest that juvenile hormone (JH) can regulate prothoracic gland activity. The regulatory effect of JH on prothoracic glands is either stimulatory or inhibitory depending on the stage of development.

Inhibitory effects of JH

In the last instar larvae of Lepidoptera, JH treatments within the feeding period causes a delay in metamorphosis for a few days. This has been observed in *Manduca sexta* (Nijhout and Williams, 1974; Safranek *et al.*, 1980; Gruetzmacher *et al.*, 1984a,b; Rountree and Bollenbacher, 1986). *Mamestra brassicae* (Hiruma *et al.*, 1978; Hiruma, 1980), *Spodoptera littoralis* (Cymborowski and Stolarz, 1979; Cymborowski and Zimowska, 1984), Laspeyresia pomonella (Sieber and Benz, 1980) and Spodoptera mauritia (Santha and Nair, 1987).

In Manduca sexta, the inhibitory effects of JH or its analogues on the prothoracic glands before the commitment peak of ecdysteroids appears to be indirect. In this insect, JH appears to have an inhibitory effect on the release of PTTH by the brain, thus indirectly blocking ecdysone secretion by the prothoracic glands (Nijhout and Williams, 1974; Rountree and Bollenbacher, 1986). A similar effect has been observed in the larvae of Mamestra brassicae (Hiruma *et al.*, 1978). In several insect species the foremost being *Chilo suppressalis* and *Diatraea grandiosella*, JH appears to be a feedback inhibitor of PTTH release inducing and/or maintaining larval diapause (Yin and Chippendale, 1973; Takeda, 1978; Chippendale and Turunen, 1981). It has been suggested that an effect on PTTH release is a logical focus of control by JH, since release of this peptide is the initial event in the hormonal cascade leading to pupal commitment (Rountree and Bollenbacher, 1986).

However observations by other authors strongly argue against the concept that the inhibitory effect of JH on prothoracic glands is via PTTH. The findings on Spodoptera littoralis (Cymborowski and Stolarz, 1979), Galleria mellonella (Ciemior et al., 1979), Manduca sexta (Safranek et al., 1980; Rountree et al., 1987; Watson and Bollenbacher, 1988) and Spodoptera mauritia (Balamani and Nair, 1992) suggest that JH directly suppresses the function of larval prothoracic glands. Studies in *G. mellonella* revealed that JH regulates ecdysone synthesis by both these mechanisms (Sehnal *et al.*, 1981).

Stimulatory effects of JH

JH is thought to maintain prothoracic glands (Gilbert, 1962) and also to stimulate ecdysone secretion in many insects. In lepidopterans, the stimulatory effects of JH on prothoracic glands is restricted to the prepupal phase of last larval instar and to the pupal stage. The first evidence for the tropic effect of JH on pupal prothoracic glands was provided by Williams (1959). He showed that implantation of active corpora allata stimulated adult development in brainless pupae of *Hyalophora cecropia*. JH also has been shown to maintain and stimulate the pupal prothoracic glands of *M. sexta* (Dai and Gilbert, 1998).

During larval-pupal transformation in most lepidopterans, a brief burst of JH occurs in the prepupal phase of development (Varjas *et al.*, 1976; Yagi, 1976; Yagi and Kuranochi, 1976; Schooley *et al.*, 1976; Hsiao and Hsiao, 1977; Sieber and Benz, 1977; Riddiford and Truman, 1978; Bean *et al.*, 1982; Balamani and Nair, 1989, 1991). The major role attributed to this prepupal surge of JH is that this increase of JH along with the second release of PTTH is necessary for activating the prothoracic glands to their maximal rate of ecdysone secretion needed for moulting and metamorphosis (Hiruma *et al.*, 1978; Cymborowski and Stolarz, 1979; Hiruma, 1980; Sieber and Benz, 1980; **a.b** Safranek *et al.*, 1980; Gruetzmacher *et al.*, 1984; Cymborowski and Zimowska, 1984).

It has been suggested that JH activates prothoracic glands by eliciting the release of a prothoracic gland stimulatory factor from the fat body **a,b** (Gruetzmacher *et al.*, 1984). In *Manduca sexta* both the timing and magnitude of the commitment peak appear to be controlled by the interaction between three elements : PTTH, the stimulatory factor present in the haemolymph and the competence of the prothoracic glands to respond to these effectors. At the time of the first PTTH release, the titre of the factor is below the threshold for stimulation of glands (Watson et al., 1987a). It appears that PTTH acts on competent prothoracic glands in the presence of low level of stimulatory factor to elicit commitment peak in the haemolymph ecdysteroid titre. At the time of the second PTTH release, the titre of this factor has risen significantly (Watson et al., 1987a). Activation of competent prothoracic glands by the combination of PTTH and high level of stimulatory factor gives rise to the large peak in the ecdysteroid titre which elicits the pupal moult. In these interactions, JH regulates the release of PTTH, the haemolymph titre of the stimulatory factor and competence of prothoracic glands (Watson et al., 1987b, Baker *et al.*, 1987).

Ecdysteroids

Ecdysteroids themselves may contribute to the regulation of ecdysone biosynthesis by feedback loops to the brain or directly to the prothoracic glands. Early evidence from parabiosis experiments with pupal *Hyalophora cecropia* indicated that the prothoracic glands could be stimulated by their own hormone (Williams, 1952). This idea was supported by Siew and Gilbert (1971) who found that the injection of 20-hydroxyecdysone into intact nondeveloping saturniid pupae stimulated incorporation of ³H uridine into the prothoracic glands. Similar injections into pupae in which adult development had started elicited a suppression of ³H uridine incorporation. These results suggest that the response of prothoracic glands to 20-hydroxyecdysone changes from positive to negative with the initiation of development. Ecdysone and 20-hydroxyecdysone have been reported to stimulate the synthesis and release of PTTH in *Leucophaea maderae* (Marks *et al.*, 1972) and *Mamestra brassicae* (Agui and Hiruma, 1977a,b).

Indications for a feedback control loop controlling the haemolymph titre of ecdysteroids has been obtained in many insect species (Mala *et al.*, 1977; Mala and Sehnal, 1978; Steel *et al.*, 1982; Beydon and Lafont, 1983). Since ecdysteroids may directly inhibit ecdysone biosynthesis in the prothoracic glands. Injections of 20-hydroxyecdysone caused a decrease in RNA synthesis in saturniid pupal prothoracic glands (Siew and Gilbert, 1971). Since 20-hydroxyecdysone was also found to elicit such inhibition in decapitated pupae, it was suggested that this steroid acts directly on the prothoracic glands (Beydon and Lafont, 1983). In *Manduca sexta*, 20-hydroxyecdysone was found to inhibit ecdysone biosynthesis in fifth instar larval prothoracic glands *in vitro* (Scott, 1982). The finding that ecdysone synthesising cells as well as the presumed prothoracicotropes and their neurohaemal organs contain ecdysteroid receptors, suggest a multiple feedback mechanism, directly on ecdysone synthesis and indirectly on PTTH synthesis/release. The inability of prothoracic glands for conversion of ecdysone to 20-hydroxyecdysone may be due to the fact that prothoracic glands are a target tissue of the moulting hormone (Bidmon and Koolman, 1989).

In summary, the hormonal influences upon prothoracic gland secretion appear to be three-fold, initial stimulation by PTTH followed by modulation by both ecdysteroids and JH. The actions of the latter two hormones appear to change during development. Both these hormones are reported to influence the prothoracic glands by two distinct routes; a direct action of the hormones on prothoracic glands and an indirect action *via* influences on the secretion of PTTH. Depending on the temporal integration of effects of primary and secondary effectors of prothoracic glands these moieties can modulate gland activity and the haemolymph ecdysteroid titre.

Competency of prothoracic glands

The competency of prothoracic glands to respond to various regulatory factors appears to be another component in the complex systems regulating prothoracic gland activity. In *M. sexta, in vitro* studies revealed that prothoracic glands exhibited an absolute refractoriness to respond to tropic stimuli during the early part of last larval instar (days 0-2) and the glands subsequently acquired the competency to respond to effectors by about days 3-4 (Ciancio *et al.*, 1986). The regulation of competency may involve humoral bioregulatory substances like JH or perhaps PTTH itself. Alternatively, it could be an intrinsic property of the glands to respond to extrinsic effectors like photoperiod (Mizoguchi and Ishizaki, 1982).

1.3 JUVENILE HORMONES (JHs)

Site of synthesis

Juvenile hormones are produced and secreted by the corpora allata (CA) (Wigglesworth, 1934, 1936), endocrine glands of ectodermal origin associated with the brain and corpora cardiaca. The corpora allata received innervations by way of cerebral and central nervous system connections from neurosecretory cell bodies located primarily in the protocerebrum and tritocerebrum, as well as by the suboesophageal ganglion. In addition, the glands themselves contain numerous neurosecretory endings which represent the site of release of allato-regulatory peptides and amines, as well as PTTH.

Chemistry

Juvenile hormones have a terpenoid backbone and contain two hydrolytically unstable functional groups, a 10,11-epoxide and a terminal methyl ether conjugated with 2,3-double bonds (Hammock, 1985). Thus JHs are said to be methyl esters of epoxy-sesquiterpenoid acids. To date, six juvenile hormones have been identified from various insect orders. JH I and JH II the exclusively lepidopteran JHs, were the first to be identified (Röller *et al.*, 1967, Meyer *et al.*, 1968). JH III appears to occur in all orders and is the principal product of corpora allata in most insects. In the Lepidoptera, five JHs are produced, JH I, JH II, JH III, JHO and 4-methyl-JH I; the latter two are found in *Manduca* embryos and JHO in *Hyalaphora cecropia* males (Bergot *et al.*, 1980).

Synthesis and release of juvenile hormones

In vitro assays of CA activity have conclusively shown that the rate of JH release is proportional to the rate of JH synthesis over a wide range of glandular activity in *Schistocerca gregaria* (Tobe and Pratt, 1974b), *Periplaneta americana* (Pratt *et al.*, 1975a) and *Diploptera punctata* (Tobe and Stay, 1977). The species studied to date do not have any mechanism for the storage of biosynthesised JH-III, and intraglandular hormone levels are proportional to synthetic rates.

JH titre is believed to be the primary endocrine factor influencing the "quality" of developmental events during metamorphosis. The classic experiments with hemimetabolous (Wigglesworth, 1934. 1936) and holometabolous insects (Williams, 1961) provided qualitative information on the role of JH in metamorphosis. JH inhibits somatic morphogenesis early during postembryonic development, but in the final instar, its declining titre is essential for imaginal differentiation (Sbrenna et al., 1988). For the larval moult, JH must be present at the onset of the rising phase of the ecdysteroid titre (Truman, 1972; Fain and Riddiford, 1976). Appreciable quantities of JH are present at the beginning of the last larval instar as well as end of the instar. The elevated titre of JH at the beginning of the instar may control the subsequent ecdysteroid or PTTH surge responsible for initiation of ecdysis (Hiruma, 1986; Rountree and Bollenbacher, 1986). JH also is essential for the continued presence of the putative JH receptor in larval Manduca epidermis, particularly at times of low ecdysteroid titre. JH may also be involved in the regulation of expressions of the isoforms of the ecdysone receptor, depending on the presence of an ecdysteroid. In the absence of JH, the cells respond to ecdysteriods by switching to a metamorphic programme resulting in the production of a new type of cuticle. If JH levels remain high early in the final

instar, the insect fails to undergo successful larval-pupal or nymphal-adult transformation. Instead it becomes a giant larva or a supernumerary nymph. On the other hand, an induced depletion in the level of JH in early instars lead to irreversible precocious metamorphosis. It is generally assumed that the absence of JH is required for the metamorphosis to the adult in holometabolous insects (Riddiford, 1994). Thus the maintenance of a critical titre of JH at different developmental phases is absolutely essential for the orderly completion of insect metamorphosis, reproduction and behaviour.

1.3.1 Control of CA activity

There are a number of regulators of JH activity. The rate of JH production is an important component in the regulation of the JH titre and this in turn is regulated by neuropeptides that directly influence the biosynthetic activity of the corpora allata.

Haemolymph JH titre

Experiments utilizing either implantation of additional CA or treatment of animals with exogenous JH or analogues have clearly revealed that JH biosynthesis is suppressed in the presence of elevated JH titres (Schooneveld *et al.*, 1979; Tobe and Stay, 1980) and stimulated when JH titres are reduced (Szibbo and Tobe, 1981a). Khan *et al.* (1982) showed that suppression of JH synthesis is mediated by signals which reach the CA both through the haemolymph and through nervous tracts from the CNS (Tobe and Stay, 1980). These results indicate that feedback loops, acting through the CNS, are responsible in part for regulation of CA activity.

Role of the brain

The brain has been implicated as a source of neurosecretory factors which control CA activity. These factors may be released in neurohaemal organs such as the CC and act *via* the haemolymph, but may also be released inside the CA themselves (Hansen *et al.*, 1982). In addition, the brain may be the origin of neural signals controlling the release of neurosecretion at the periphery, inside or outside the CA. Neurotransmitters may also have direct effects on the activity of CA cells. All these signals may be inhibitory (allatostatic) or stimulatory (allatotropic). These are peptides which are capable of stimulating or inhibiting JH production *in vitro* (Tobe and Stay, 1985). Their potency in altering JH production *in vitro* argues strongly for a role in the control of CA activity.

JH titre is determined by a complex interaction among biosynthesis, degradation, and binding to JH specific binding proteins which may differ in hemi- and holometabolous larvae. JH-binding proteins may not only afford protection for JH in the presence of catabolic enzymes, but also may be agents responsible for reducing JH levels within tissues. JH also regulates ecdysteroid titre and appears to be responsible in part for the ecdysteroid surge that ultimately regulates the moulting process. The JH titre can be regulated by changes in the production of JH and this in turn is regulated by neuropeptides that directly influence biosynthetic activity of the corpora allata.

1.4 TESTES AS A SOURCE OF ECDYSTEROIDS

Ecdysteroids have been reported in adult males of Drosophila melanogaster (Hodgetts et al., 1977) and Calliphora vicina where they are localised primarily in the testes (Koolman et al., 1979). Subsequent investigations into testicular ecdysteroids in various lepidopteran species have largely been confined to immature stages, although the testes of adult Heliothis virescens have also been shown to contain ecdysteroids (Loeb et al., 1984). In the testes of the last larval instar of the European corn borer, Ostrinia nubilalis, 20-hydroxyecdysone and various high polarity ecdysteroids were detected by radioimmunoassay (RIA) while in pupae and pharate adults ecdysone was additionally detected (Gelman et al., 1988). The testes of the last instar larvae of Spodoptera littoralis was also reported to synthesize ecdysteroids (Jarvis et al., 1994).

Loeb et al. (1982) have shown that incubation in vitro of the testes of H.virescens from mid to late last larval instar and from pupae after three days of pupal development release large amounts of ecdysteroids into the medium. Testes from the O.nubilalis are also capable of synthesising immunodetectable ecdysteroids in vitro (Gelman et al., 1989). When pupal testes of diapausing Mamestra brassicae were incubated for 24 h in vitro, 2-deoxyecdysone, ecdysone and 20-hydroxyecdysone were found in the culture medium when analysed by HPLC - RIA (Shimizu et al., 1985). It has been predicted that the synthetic periods of the testes may affect, or be affected by internal changes in spermatocyst development, in addition to changes occurring outside the compartment containing spermatocysts and would also benefit from hormones released into the general circulation (Loeb et al., 1984). Considerable differences among lepidopteran species is observed regarding optimum condition for synthesis, times of synthesis and the nature of ecdysteroids produced. It was also suggested that exogenous ecdysteroids is needed to initiate (Loeb et al., 1988) or boost (Loeb et al., 1986) endogenous ecdysteroid production by the testes in vitro.

Testicular metabolism of ecdysteroids has been investigated in O. nubilalis, Lymantria dispar and H. virescens by incubating the testes with [³H] ecdysone in vitro. In pupae of L. dispar, the testes converted [³H] ecdysone into 20-hydroxyecdysone (Loeb et al., 1988) as they did in larvae, pupae and pharate adults of O. nubilalis (Gelman et al., 1985). In O. nubilalis, this conversion was localised in the testes sheath (Gelman et al., 1985). In H. virescens only the last larval instar was investigated and 20-hydroxyecdysone was again found to be the major product. However in this case, traces of 20,26-hydroxyecdysone and 3- epi, 20-hydroxyecdysone were also identified (Loeb and Woods, 1989). In the latter study, the conversion of [³H] cholesterol in the testes *in vitro* was also investigated, but no incorporation into ecdysteroids was obtained. The use of radiolabelled precursors in the last instar larvae of *Spodoptera littoralis* showed that of all the larval tissues investigated, the testes converted radiolabelled precursors into ecdysteroids most efficiently. The ability of the testes to convert [³H] 2,22,25- trideoxyecdysone into ecdysteroids has been shown to vary during the last larval instar (Jarvis *et al.*, 1994).

1.4.1 Ecdysteroid titres of the testes in relation to haemolymph titres

The total ecdysteroid levels in the haemolymph and testis exhibited a similar pattern of fluctuation during the fifth (final) instar and pupal stage of *H. virescens* (Loeb *et al.*, 1982), *O. nubilalis* (Gelman *et al.*, 1988) and the pupal stage of *M. sexta* (Friedlander and Reynolds, 1992). However, the testicular ecdysteroid levels differed from those in the surrounding haemolymph at certain times depending on the stage of development. The levels of all ecdysteroids were found to be lower in the testes than in the haemolymph (Gelman *et al.*, 1988). In *C. vicina* no ecdysteroids were detected in the haemolymph of the adult flies by Koolman *et al.*, (1979) even though large amounts of ecdysteroids in the testes of the adult were found to be lower

than in the gonads of late mid-last instar larvae and pupae, nevertheless, adult testes were found to be capable of statistically significant ecdysteroid synthesis *in vitro* (Loeb *et al.*, 1984). The existence of a blood-testis barrier has also been suggested (Kambysellis and Williams, 1972; Szollosi *et al.*, 1980). Further, Jarvis *et al.* (1994) reported that the ecdysteroids of both the testis and haemolymph of last instar larvae of *S. littoralis* were different both in the timing of their fluctuations and in their content.

1.4.2 Factors regulating synthesis of testicular ecdysteroids

Previous estimates suggest that neurosecretory stimuli which induce ecdysone synthesis by the prothoracic glands may be closely related to stimuli which induce ecdysteroid synthesis by the testes.

Hence, testicular ecdysteroid synthesis was suggested to be under the control of a brain peptide, testis ecdysiotropin (Loeb *et al.*, 1987). Regulation of the maturation of male genital tissues by inducing the synthesis of growth factors by the testis and fat body has also been suggested (Loeb and Hakim, 1991; Loeb, 1991b, 1994). However, no neurosecretory nerves has been found entering the testis of Lepidoptera (Szollosi, 1982). But a unique heart-testis complex has been reported in the last instar larvae of *H. virescens* (Meola and Loeb, 1995). This cardiotesticular muscle complex, attaches the heart to the testis, which in turn is innervated by neuroendocrine nerves. This has been proposed to be the route by which the brain peptide, testis ecdysiotropin

reaches the testis. Meola and Loeb (1995) also suggested that the cardiotesticular muscle and the products sequestered in the cardiotesticular sinus may be involved in the fusion of the paired testis and maturation of the male reproductive system, since they were present only during this stage of development.

The testis ecdysiotropic peptide from the brain of L. dispar has been purified and its structure determined to be NH₂-IIe-Ser-Asp-Phe-Asp-Glu-Tyr-Glu-Pro-Leu-Asn-Asp-Ala-Asp-Asn-Asn-Glu-Val-Leu-Asp-Phe-OH. The peptide is biphasic in activity with maximal activity in the pupal testis (Wagner *et al.*, 1997). It was suggested that the strongly bioactive portion of the molecule probably resides in residues 1-15. This molecule requires cations like Ca²⁺ for its biological activity (Loeb, 1991b). Atleast five natural testis ecdysiotropic analogues or LTE have been identified in *L. dispar* brains (Loeb *et al.*, 1997).

Role of testes sheath in the synthesis of ecdysteroids

Immunocytochemical techniques have implicated the testes sheath as the site of ecdysteroid production. Studies revealed sensitivity in and around cells of the inner layer of the testes sheath and the extensions of the inner layer that form the walls of each of the four testes follicles (Loeb, 1986). Transmission electron micrographs of sheaths of lepidopteran testes (Szollozi *et al.*, 1980, Szollozi, 1982) indicate that the tissue of the inner layer is
structurally different from that of the outer layer, the inner layer contain lipid and glycogen deposits, abundant mitochondria, and a peculiar network of swollen channels containing flocculent material. The channels if proved to be swollen smooth endoplasmic reticulum, the sheath of insect testes resemble other tissues capable of steroid synthesis.

Analysis of the contents of testes of H. virescens indicated that approximately half the ecdysteroids were in the sheath and half were in the luminal fluid with minor amounts in the spermatocysts (Loeb *et al.*,1984). Giebultowicz *et al.* (1987) showed that premeiotic spermatogonia can initiate meiosis and elongation *in vitro* if they are provided with calf serum and testes sheath factors. Their data suggest that testes sheath tissue releases a factor (s) stimulating meiosis. On the other hand, it is also suggested that the sheath may provide nourishment or other conditions for survival of cysts, thus allowing meiosis to occur as a step in germ cell autodifferentiation. The spermatogenesis promoting effect of testes sheath was found to be dose dependent and varied with the donor's age.

1.4.3. Insect Growth Regulators

Insect growth regulators (IGRs) are a group of compounds which interfere with normal growth, development and reproduction in insects by disrupting hormonally regulated physiological processes of insects. There are mainly three categories of IGRs: (a) compounds which directly influence hormones regulating development and reproduction (e.g., juvenile hormone analogues, anti-juvenile hormone agents, ecdysone mimics, anti-ecdysteroids, neurohormone analogues and their antagonists); (b) compounds which inhibit cuticle formation through an effect on chitin synthesis (e.g., Benzoy] phenyl ureas) and (c) compounds with miscellaneous mode of action (e.g., Azadirachtin). Several such compounds are known and their effects on metamorphosis and reproduction in a number of insect species have been extensively studied and reviewed (Retnakaran *et al.*, 1985; Staal, 1986; Darvas and Varjas, 1990; Nair, 1993). These compounds are considered to be useful in pest control programmes because they are target specific, nonpersistent, biodegradable and environmentally benign substances to which insects may not develop resistance.

IGRs based on insect hormones have great significance as pesticides of the future and also as excellent chemical probes to elucidate the role of hormones in the basic physiological processes of insects. Manipulation of the hormone levels in the haemolymph will cause derangement of hormone dependent processes of morphogenesis and reproduction. Based on this concept many hormone analogues and antihormones have already been developed.

(i) IGRs based on neurohormones

Since many of the neurohormones of insects have not been fully characterised, neuropeptide based IGRs for insect pest control is not well advanced. The diversity and complexity of neurohormones offer lot of possibilities for design and development of neurohormone analogues. The neurohormone biosynthesis, release, transport or receptor binding are all vulnerable steps for manipulation in insect pest management.

(ii) IGRs based on juvenile hormones

Williams (1967) suggested that juvenile hormone or its analogues could be used to control insect pests. Since then, numerous analogues with JH activity has been discovered and their biological and insecticidal properties have been well studied. Some of them are Epofenonane, Methoprene, Hydroprene, Kinoprene and Phenoxy carbamate. Changing the haemolymph JH titer by treatments with JHAs at critical periods during the life cycle of an insect will interfere with normal metamorphosis and reproduction.

The limited scope of JHAs as control agents led to the discovery of compounds with anti-JH activity. Examples of anti-JH agents are Precocenes, Fluoromevalonolactone (FMev), ETB, EMD, Compactin, Piperonyl butoxide, Allylic alcohols, Bisthiol carbamate etc. Anti-JH agents disrupt normal development of early larval instars and inhibit JH dependent reproductive **a,b,c** activities (Sam Mathai and Nair, 1984; Santha and Nair, 1986, 1988, 1991; Santha *et al.*, 1987; Nair, 1993).

(iii) IGRs based on ecdysteroids

Many compounds with ecdysone activity have been isolated from plants. Ponasterones, with moulting hormone activity was discovered by Nakanishi *et al.* (1966). Since then, many synthetic and natural steroids with moulting hormone activity have also been discovered. They are divided into four groups: Zooecdysoids, Phytoecdysoids, Synthetic ecdysoids and non-steroidal ecdysone agonists. An example of a zooecdysoid is the crustacean moulting hormone. The important phytoecdysoids are ponasterone, inokosterone, isoinokosterone etc. Synthetic ecdysteroids are steroids with moulting hormone activity. e.g., 22-isoecdysone.

Wing (1988) reported that a non-steroidal benzoyl hydrazine analogue, code named as RH 5849 possesses ecdysone mimetic properties. Treatment of insects with minute doses of RH 5849 interferes with feeding activity and induce a premature lethal moult (Wing, 1988; Sakunthala and Nair, 1995). RH 5849 was found to be effective against insects belonging to various orders. Yet another non-steroidal ecdysone mimic (RH 5992) discovered by Rohm and Haas Company, U.S.A. induces an incomplete moult in *Choristoneura fumiferana* (Palli *et al.*, 1995). This compound also induces spermatogenesis reinitiation in isolated abdomens of diapausing *Cydia pomonella* larvae (Friedlander and Brown, 1995) RH 5992 acts similar to 20-hydroxyecdysone by binding to the ecdysone receptor (Wing, 1988; Wing *et al.*, 1988). RH 5992 inhibited oviposition in *Spodoptera littoralis* (Smagghe and Degheele, 1997).

1.5. OBJECTIVES OF THE INVESTIGATION

Growth and development in insects, which are punctuated by periods of moulting are regulated by endogenous titres of ecdysteroids, JH and PTTH. These hormones are also involved in the regulation of reproduction of insects. Therefore, information on the occurrence, titre changes and factors regulating the hormone titres is essential to elucidate the role of hormones during insect development and reproduction. Although ecdysteroid titres during postembryonic development have been determined for a number of lepidopteran species, vast differences occur in the time sequence of developmental events in insects. Hence hormone titres obtained from one species at any given point of time may not accurately coincide with the same in another species. Hence the present investigation is undertaken to study the ecdysteroid titre changes in the haemolymph and testes during larval development and metamorphosis of Spodoptera mauritia Boisd. (Lepidoptera: Noctuidae). This lepidopteran insect is a sporadic pest of paddy in the state of Kerala, in India. This insect has been subjected to a number of experimental studies in our laboratory concerning the effect of insect growth regulators with hormonal and anti-hormonal activity on larval development, metamorphosis and reproduction (Nair, 1980, 1981, 1993; Sam Mathai and Nair, 1983, 1984a,b,c,d; Santha and Nair, 1986, 1987, 1988; Santha et al., 1987; Nair and Rajalekshmi, 1989; Pradeep and Nair, 1989; Sam Mathai et al., 1989; Balamani and Nair, 1989a,b; Santha and Nair, 1991; Balamani and Nair, 1991, 1992, 1994; Jagannadh and Nair, 1992; Sakunthala and Nair, 1995; Vengopalan et al., 1994; Benny and Nair, 1999). The interpretation of results obtained from these studies was often hampered by the lack of data concerning the timing of release of hormones during larval development and metamorphosis. Hence one of the major objectives of the present investigation is to analyse the timing of release of the principal morphogenetic hormones namely PTTH and JH and determination of haemolymph and testicular ecdysteroid titres. For this, the classical approach of ligation (neck or thorax) extensively utilised in endocrinological laboratories is employed. Further, the effects of exogenous treatment of a JH analogue (JHA), hydroprene and a nonsteroidal ecdysteroid agonist, RH 5992 on haemolymph and testicular ecdysteroid titres are also determined. The fluctuations in hormone titres are correlated with morphogenetic changes occurring during metamorphosis of normal insects and those of treated ones. Further, fluctuations in the testicular ecdysteroid content during testicular development and spermatogenesis are studied. The effects of treatments of hormone analogues/agonists on testicular ecdysteroid titres are also analysed. It is

hoped that the results of this investigation will lead to a better understanding of the endocrine regulation of larval development, metamorphosis and spermatogenesis and will form the basis for future investigations on various aspects of metamorphosis and reproductive maturation of *S. mauritia*.

Chapter 1 deals with a detailed review of insect hormones, testicular ecdysteroids and factors regulating their synthesis and release.

Chapter 2 provides basic information on the pest status and a detailed account of the rearing and maintenance of *Spodoptera mauritia* Boisd. (Lepidoptera: Noctuidae) under laboratory conditions. The general experimental techniques employed in this investigation are also included.

Chapter 3 examines the haemolymph ecdysteroid profile during larval-pupal and pupal-adult development of *S. mauritia*. It also incorporates the timing of PTTH release during larval-pupal development. Effects of hormones/hormone analogues on the ecdysteroid titres of the haemolymph are also analysed.

Chapter 4 examines the testicular ecdysteroid profile during larval-pupal and pupal-adult development in *S. mauritia*. Effects of hormones/hormone analogues on testicular ecdysteroid titres are also analysed.

CHAPTER 2

MATERIALS AND METHODS

5



2.1 PEST STATUS OF S. MAURITIA

Spodoptera mauritia Boisd. (Lepidoptera : Noctuidae) commonly known as rice swarming caterpillar, or army worm is a sporadic pest of Oryza sativa. The pest attack assumes serious proportions and cause considerable damage especially when there is a good start of monsoon followed by a prolonged dry spell. This insect is distributed all over India and usually occurs on paddy from July to September. Large swarms of larvae appear suddenly and destroy whole fields of paddy and then march on to the next. This tendency of the pest to migrate from field to field in large numbers gives it the name army worm. The larvae attack the nursery and early growing stages of paddy. The seedlings in the nursery are cut and completely eaten up as though grazed by cattle. Larval feeding progresses from the leaf margins leaving behind the leaf midribs thus completely destroying the young paddy seedlings. In a severe outbreak, serious damage results in heavy loss of crop. The loss in yield due to larval infestation ranges from ten to twenty percent. The pest status of Spodoptera mauritia is further complicated by their ability to migrate to alternate host plants (for eg. Ischaemum aristatum) during off-season periods at the end of which they make a full scale comeback on the nursery stages of paddy. Spodoptera species have been widely utilised in physiological, biochemical and endocrinological, research, holding the tenth position in the group of insect species often used for research.

2.2. REARING AND MAINTENANCE OF THE LARVAE OF S. MAURITIA

The female adult moths which were attracted to the light during night were collected using a sweeping insect net. In the laboratory, these moths were then transferred to glass chimneys closed at both ends with muslin cloth. Adult moths were fed on 10% solution of honey. Cotton swabs soaked in 10% solution of honey were kept in the glass chimneys for this purpose. The female laid eggs on the cloth or on the sides of the chimneys, from which the first instar larvae hatched after 3 days.

The larvae were reared in glass chimneys at the initial stages. The larvae were supplied fresh, tender leaves of the grass *Ischaemum aristatum* collected from paddy fields. The food material was changed every day. Uniform rearing conditions were provided to larvae and they were kept away from wind and intense light. The larvae were maintained at room temperature, RH 90 \pm 3% and 12:12 light : dark, photoperiod regime. Larvae were transferred to large plastic troughs as they grew in size. During summer days, the cloth covering the trough was wetted frequently. The pupae were kept seperately in beakers for adult emergence.

2.3 BIOLOGY OF S. MAURITIA

The S. mauritia larvae developed at a uniform rate and underwent six larval instars before pupating under the conditions provided in the laboratory.

(i) First instar larva

The newly hatched larvae congregated on the cloth covering of the chimney. The first instar larvae were characterised by the presence of a large, black head shield and light green coloured body. On each segment of the body setigerous, small, wart-like, dark, pigmented tubercles were present which were arranged in a cross wise row. The larvae moved in a characteristic leaping manner. They descended by means of silken threads to the tender grass leaves supplied for feeding. Newly hatched larvae did not feed immediately. The first instar larvae measured about 1 mm in length and 0.5 mm in width. These larvae moulted to the second instar after 2-3 days.

(ii) Second instar larva

In second instar larvae, three white longitudinal stripes appeared on the dorsal surface of the body extending from the prothorax to the last abdominal segment. The larval body was green in colour with two pairs of white, longitudinal, lateral stripes, one pair being more prominent. On each segment, setigerous, small, wart-like, dark tubercles were present. The second instar larvae also descended using silken threads. Newly moulted

PLATE I

Fig. 1.	Egg mass
Fig. 2.	Fourth instar larvae
U	A. Day 1 larva
	B. Day 3 larva
Fig. 3.	Fifth instar larvae
•	A. Day 1 larva
	B. Day 3 larva
Fig. 4.	Sixth instar larvae
-	A. Day 3 larva
	B. Day 4 larva (wandering stage)
	C. Prepupa
Fig. 5.	Pupa
Fig. 6.	Adults
	A. Female
	B. Male



second instar larvae measured about 2.5 mm in length and 0.5 mm in width. The duration of this instar was 2-3 days.

(iii) Third instar larva

The larvae no longer used silken threads to descend to the grass leaves. The third instar larvae possessed three white longitudinal stripes on the dorsal side and two pairs on the lateral side of the body extending from anterior to the posterior end. These larvae also had reddish black supraspiracular stripes extending from anterior to posterior end. The newly moulted third instar larvae measured 6 mm in length and 1 mm in width. After 2-3 days, larvae moulted to fourth instar.

(iv) Fourth instar larva

The three, dorsal longitudinal stripes became dull white in colour. Two lateral reddish black stripes were present, one on each side of the body. Body became greyish in colour. Black intermittent dots appeared on each segment dorsolaterally which broadened towards the later stages of the instar. The dorsum of the larva was paler than the supraspiracular area and was over laid with strands and flecks of brown. The newly moulted larva measured about 11.5 mm in length and 2 mm in width. The fourth instar larval period extended upto 2-3 days (Pl : I, Fig. 2). of gut purge, larvae start the wandering behaviour which lasts for 24 h. During this period the larvae measured 24 mm in length and 5 mm in width. These larvae wandered about at the bottom of the rearing trough. At the later stages of wandering, the body of larvae shrunk gradually and the movements reduced as the abdominal prolegs got retracted. The larvae thus got transformed into the prepupal stage. The prepupal stage was characterised by a highly wrinkled larva which underwent larval-pupal apolysis after 24 h. The prepupal stage measured 20 mm in length and 5 mm in width.

(vii) Pupal instar

Pupae were of the obtect type, dark brown in colour and measured 16 mm in length and 5 mm in width (Pl : I, Fig. 5) Adults emerged out of the female pupae after 7 days. Male pupae took 8 days for adult emergence.

(viii) Adults

Adults were medium sized moths and had a conspicuous spot on the forewings, which had wavy pattern on the fringe. Insects measured about 15 mm in length and had a wing span of 30-35 mm. Adult moths exhibited sexual dimorphism in their morphological characters. Males were dark greyish with white markings on forewings and were provided with large tufts of hairs on the forelegs. Females lacked both these white markings and tufts of hairs (Pl : I, Fig.6). Mating took place in the night within 24 h after emergence. Egg laying commenced 24 h after mating. Eggs were laid in masses of 100-500 each and were covered with buff coloured silken hairs (P1 : 1,Fig. 1). On the whole, embryonic period lasted 2-3 days, larval period 19-23 days and pupal period 7-8 days.

2.4 EXPERIMENTAL ANIMALS

The larvae/pupae/adults used for various experiments were collected from the laboratory colony reared and maintained as described above. Newly moulted larvae were isolated from the stock culture and reared in separate containers to facilitate the synchronous development of larvae and also for the precise determination of the age of the larvae used for experimental work. Larvae which showed head capsule slippage were kept isolated on the day prior to the moulting. The newly moulted larvae which had pale colour were separated on the day of the ecdysis itself. In the case of sixth instar larvae, to observe the determination and timing of gut purge, the experimental larvae were placed in separate containers, the bottom of which was lined with filter paper in order to absorb the fluid faeces. Gut purged larvae were separated using the criteria of loss of weight, colour change and the observation of fluid faeces in the filter paper. Sexing was possible in the late fifth instar stage when the undifferentiated testes became clearly visible through the transparent cuticle of male larvae. Insects were sexed in the fifth instar larval stage and reared separately. The experimental insects were reared and maintained as in the case of stock culture. The percentage of mortality during each instar was also observed.

Prepupae were separated from the stock culture to obtain newly moulted pupae. Newly ecdysed (day 0) pupae were easily recognised as they appeared pale cream coloured. Soon tanning of the pupal cuticle commenced changing the colour to light brown and finally dark brown. Adults (male) emerged after 8 days.

The age of larvae and pupae and adults were abbreviated to day n where day 0 indicates the day of ecdysis to this stage. Newly ecdysed larvae/pupae/adults were treated as day 0, larvae/pupae/adults 36 h old as day 1 and so on. Slight variations in the intermoult duration was frequently observed in the laboratory stock culture. So larvae showing synchronous development from the same egg mass were utilised for conducting experiments.

2.3 GENERAL EXPERIMENTAL TECHNIQUES

In order to analyse the role of hormones in insect postembryonic development especially during metamorphosis and spermatogenesis, the classical approach of neck/thorax ligation to eliminate the endogenous source of hormones and the hormone replacement therapy were employed. Sixth instar larvae of day 1 and day 4 were used for the experiments. Larvae of these age groups were selected because the head critical periods occur supposedly on day 1 and day 4 of sixth instar larval stage. Formation of spermatids and initial differentiation of sperm bundles also take place at this stage (Venugopalan *et al.*, 1994). In one set of experiments, sixth instar larvae were 'neck-ligated' to remove the endogenous source of cerebral neurosecretory factors and JH. In another set, the sixth instar larvae were 'thorax-ligated' to remove secretions of the cerebral neurosecretory cells and prothoracic glands.

In yet another series of experiments, neck-/thorax-ligated day 1/day 4 larvae and pupae were topically treated with a JHA, hydroprene or with a non-steroidal ecdysteroid agonist, RH 5992. The effects of these treatments on metamorphosis and spermatogenesis were carefully analysed.

Ligations

Sixth instar day 1 and day 4 larvae were mildly anaesthetised using diethyl ether and tightly ligated using a cotton thread. Ligations were carried out either around the neck or between the pro- and mesothorax. The former is designated as 'neck-ligated' larvae which lacked brain and corpora allata and the latter as 'thorax-ligated', which lacked not only brain with retrocerebral endocrine glands but also prothoracic glands.

Chemicals

The juvenile hormone analogue (JHA), hydroprene (ethyl 3,7,11trimethyl dodeca-2,4-dienoate) was a gift from Dr. G.B. Staal, Zoecon Corporation, California, USA. The non-steroidal ecdysteroid agonist RH 5992 (Tebufenozide, 1,2-dibenzoyl-1-tert-butyl hydrazide) was obtained as a gift from Rohm and Haas Company, Spring House, Pennsylvania, USA.

JHA or RH 5992 was dissolved in acetone to prepare a solution of $1 \mu g$ JHA/RH 5992 per μ l acetone or 5 μg JHA/RH 5992 per μ l acetone. For various treatment procedures, measured quantities of these compounds were used utilizing a Hamilton microsyringe.

Treatments

(a) Treatment of hydroprene on unligated larvae

Sixth instar day 0 larvae were treated topically with repetitive daily dose of 20 μ g JHA (day 0 - day 3) on the abdominal tergites using the microsyringe. Subsequent treatments were done at approximately the same time of the day. Haemolymph was extracted and the testes dissected from day 1 onwards. Control larvae were treated with the same volume of acetone.

(b) Treatment of RH 5992 on unligated larvae

Sixth instar day 0 and day 4 larvae were treated topically with a single dose of 2 μ g or 5 μ g RH 5992 on the abdominal tergites using the microsyringe. Control larvae were treated with the same volume of acetone.

(c) Treatment of hydroprene on ligated larvae

Neck-/thorax-ligated sixth instar day 1 larvae were treated topically with a daily dose of 5 μ g JHA (day 1 – day 3) on the abdominal tergites, using the microsyringe. The first dose was given thirty minutes after ligation and the second and third doses were given at the same time on subsequent days. Haemolymph was extracted and the testes dissected from the treated larvae on the next day of completion of hormone treatment. In a similar manner, neck-/thorax-ligated day 4 larvae were treated with a single dose of 5 μ g JHA and the haemolymph was extracted and the testes dissected on the subsequent days. Control larvae were treated with a similar volume of acetone.

(d) Treatment of RH 5992 on ligated larvae

Neck-/thorax-ligated sixth instar day 1 and day 4 larvae were treated topically with a single dose of 2 μ g or 5 μ g or 10 μ g RH 5992 on the abdominal tergites using the microsyringe. Haemolymph was extracted and the testes dissected from the treated larvae on the day of treatment and on all subsequent days of survival. Control larvae were treated with the same volume of acetone.

(e) Treatment of hydroprene on newly ecdysed pupae

Newly ecdysed pupae (day 0) of S. mauritia were treated topically with $0.5 \mu g$ and $1 \mu g$ JHA using the microsyringe. Control pupae were treated with the same volume of acetone. The haemolymph was extracted and the testes dissected from the treated pupae on all days of survival.

(f) Treatment of RH 5992 on newly ecdysed pupae

Newly ecdysed pupae (day 0) of S. mauritia were treated topically with 5 μ g or 10 μ g RH 5992 using the microsyringe. Control pupae were treated with the same volume of acetone. The haemolymph was extracted and the testes dissected from the treated pupae on day 2 and day 5 of the pupal stadium.

Preparation of haemolymph samples for RIA

Haemolymph was extracted in 300 μ l of 75% ice cold methanol in 1.5 ml Eppendorf tubes. The methanolic extracts were vortexed, refrigerated for at least 30 minutes and centrifuged at 14,000 rpm at 4°C for 5 min. Aliquots of the supernatant were evaporated in vacuo in 6 x 50 mm borosilicate tubes using Savant speedvac lyophiliser and stored in the freezer. Ecdysteroid titres were determined by radioimmunoassay following the methods of Borst and

O'Connor (1974), Bollenbacher et al. (1975) and Gelman et al. (1997). The antiecdysone, a gift from W.E. Bollenbacher was prepared from a hemisuccinate derivative of ecdysone (at the C-22 hydroxyl group) that had been coupled to thyroglobulin. Competitive studies showed that the antibody had a high affinity for ecdysone, 20-hydroxyecdysone and makisterone A (Gilbert et al., 1977; Kelly et al., 1981). Standards (50-4000 pg) were prepared from 20-hydroxyecdysone (Calbiochem. Corp., La Jolla, CA). The radioactively labelled antigen was tritiated ecdysone (63.5 Ci/mmol: New England Nuclear, Boston, M.A., USA). The antigen-antibody complex was precipitated with saturated ammonium sulphate yielding a final concentration of 50% ammonium sulphate. The precipitate was washed with 50% ammonium sulphate and then dissolved in 25 µl of distilled water. After the addition of 300 µl of Ecoscint A (National Diagnostics, Atlanta, GA, USA), each tube was vortexed and counted for [3H] in a Beckman (Columbia, MD, USA) LS 3801 Scintillation Counter. The concentration of ecdysteroids was determined from curves generated by the 20-hydroxyecdysone standards (50-4000 µg) that had been assayed with the samples, and was expressed as pg 20-hydroxyecdysone equivalents/µl haemolymph.

Preparation of testes samples for RIA

The testes were homogenised in 1.5 ml Eppendorf tubes containing 100 μ l ice cold 75% methanol using a plastic homogenizer. The homogenate and a

200 μ l wash of the homogenizer were vortexed, refrigerated for at least 30 min and centrifuged at 14,000 rpm at 4°C for 5 min. Aliquots of the supernatant were evaporated in vacuo in 6 x 50 mm borosilicate tubes using Savant speedvac lyophiliser and stored in the freezer. Ecdysteroid titres were determined by radioimmunoassay by the methods described earlier.

Statistical analysis of data

Ecdysteroid titres of the haemolymph and testes have been represented as line graph or as histogram. Each point or column in the graph/histogram represents the mean of at least three separate determinations. Student t-tests were employed to determine whether haemolymph/testicular ecdysteroid peaks were significant. The mean peak value was compared with the two mean baseline values on either side, not necessarily two adjacent points, but rather points that delineated the peak from the rest of the curve.

CHAPTER 3

DEVELOPMENTAL PROFILE OF HORMONES DURING METAMORPHOSIS AND THE EFFECTS OF HORMONE MIMICS

3.1 INTRODUCTION

During metamorphosis in lepidopteran insects, developmental changes occur in a sequential order as has been demonstrated in Bombyx mori by Kiguchi et al. (1985). These sequential events have been assumed to be under the control of ecdysteroids, JH and brain hormones (For review see Gilbert et al., 1996). Changes in haemolymph titre of ecdysteroids have been determined in the last larval stadium of various lepidopteran insects and a small ecdysteroid peak occurs immediately before gut purge (Dean et al., 1980; Smith, 1985). The small peak coincides with the occurrence of pupal commitment of epidermal cells in Manduca sexta (Bollenbacher et al., 1975; Riddiford, 1978) and is thus designated as the commitment peak. The low ecdysteroid titre peaks before the gut purge have been correlated with initiation of nucleolar activity, RNA synthesis and organelle formation in the fat body and epidermis (Dean et al., 1980) and development of wing discs in Bombyx (Kawasaki, 1995). A major peak precedes pupal development which triggers apolysis and subsequent events that lead to pupation (Riddiford, 1976). During pupal-adult development, a single large peak occurs in the first half of pupal stage which probably promotes apolysis and pharate adult development in all the lepidopterans examined. However, the prothoracic glands have been reported to undergo apoptosis (programmed cell death)

during the pupal stage (Dai and Gilbert, 1997). Hence the glandular source of ecdysteroids during pupal stage is still unclear.

Prothoracic glands of both larvae (Gibbs and Riddiford, 1977; Mala et al., 1977) and pupae are stimulated by PTTH from the brain (Agui et al., 1980). The head critical periods (HCP) in the last larval stadium of lepidopteran insects have been determined by neck ligation. HCP has been defined as the period during which a linearly increasing percentage of neckligated larvae exhibit dorsal vessel exposure during the same gate as intact larvae (Truman and Riddiford, 1974). However, later studies suggested that HCP signifies a change in the responsiveness of the prothoracic glands to PTTH (Sakurai et al., 1998).

In addition to environmental factors, the release of PTTH is regulated by endogenous factors such as JH. At the beginning of the last larval instar, JH may have a direct inhibitory effect on prothoracic gland function (Watson and Bollenbacher, 1988; Sakurai *et al.*, 1989), in addition to suppressing PTTH release (Nijhout and Williams, 1974; Rountree and Bollenbacher, 1986). In contrast to the inhibitory effect of JH early in the final instar, JH exerts a stimulatory effect following the commitment peak of ecdysteroids. At this time, JH increases haemolymph ecdysteroids and accelerates pupation (Cymborowski and Stolarz, 1979; Safranek *et al.*, 1980; Gruetzmacher *et al.*, 1984b). Juvenoids have also been reported to prevent apoptosis of pupal prothoracic glands and promote synthesis of ecdysteroids in the treated pupae (Dai and Gilbert, 1998).

The prothoracic glands have been shown to possess ecdysteroid receptors (Bidmon and Sliter, 1990). Hence ecdysteroids or ecdysteroid agonists (RH 5849, RH 5992) themselves may have a feedback effect on the prothoracic glands (Sakurai and Williams, 1989; Warren and Hetru, 1990). Further the nature of the ecdysteroid feedback effect (positive or negative) depends upon the secretory history of the prothoracic glands (Williams, 1952; Siew and Gilbert, 1971; Sakurai and Williams, 1989). The present study attempts to identify the timing of release of the principal morphogenetic hormones PTTH and JH and determine the haemolymph ecdysteroid profile during larval-pupal and pupal-adult development of Spodoptera mauritia Boisd. (Lepidoptera: Noctuidae). Further, this study examines the effects of exogenous treatments of a JH analogue (JHA), hydroprene and a non-steroidal ecdysteroid agonist, RH 5992 on haemolymph ecdysteroid titres. The present study also demonstrates that the fluctuations in the titre of ecdysteroids can be correlated with specific developmental events occurring during larval-pupal and pupal-adult metamorphosis.

3.2 MATERIALS AND METHODS

Animals

Larvae and pupae of S. mauritia were taken from the laboratory stock culture reared and maintained as described previously. The present studies were conducted on sixth instar larvae (final instar) and pupae of various ages.

Ligations

Ligations were done as described earlier (see page 45).

Chemicals

Chemicals were obtained from various sources as mentioned earlier (see page 46).

Treatments

Larvae and pupae were treated with different dosages of chemicals as described previously (see page 46).

Collection of haemolymph

Sixth instar larvae of different age groups were mildly anaesthetised with diethyl ether and 5-20 μ l of haemolymph was collected in a calibrated micropipette through an incision made on a proleg. Pupal haemolymph (2 μ l) was collected in a similar manner by means of punctures made near the head region. Haemolymph samples could not be taken from adults owing to the fat body dispersed in the haemolymph.

Preparation of haemolymph samples for RIA

Haemolymph samples were prepared (see page 48) and RIA done as described earlier.

3.3 RESULTS

3.3.1 Haemolymph ecdysteroid titres of larvae, pupae and pharate adults

Sixth (final) instar larvae of S. mauritia fed voraciously during the first three days. On day 4, the larvae stopped feeding, emptied their guts (gut purge) and transformed into wandering stage. Wandering larvae showed a reduction in size and transformed into prepupae by day 5. The prepupal stage is characterised by a highly wrinkled body. Prepupae underwent pupation within 20-24 h. On the whole, the sixth instar larval period extended up to 6 days. The pupal period lasted 7 days in females and 8 days in males. Dissection of pupae revealed that pharate adult development commenced on day 4.

The present study was undertaken to determine haemolymph ecdysteroid titres of male sixth instar larvae, pupae and pharate adults. Ecdysteroid titres in the haemolymph samples were determined by means of RIA and the results correlated with known developmental events occurring during larval-pupal-adult development and metamorphosis of *S. mauritia*. For a few stages sampled, haemolymph ecdysteroid titres of males were not significantly different from those of females. Results are expressed as pg 20-hydroxyecdysone equivalents/µl haemolymph.

(i) (a) Haemolymph ecdysteroid titres during sixth instar larval stadium

Haemolymph ecdysteroid titres for sixth instar larvae of *S. mauritia* are shown in Fig. 7. In the newly moulted (white head capsule) sixth instar larvae (day 0), the titres were 37 ± 21 pg/µl which decreased to 15 ± 2 pg/µl on day 1. Hormone titres gradually increased on day 2 (26 ± 5 pg/µl) to form a peak on day 3 (39 ± 20 pg/µl). Titres increased to more than 100 pg/µl haemolymph for the first time on day 4 (372 ± 13 pg/µl). It may be recalled that on day 4, larvae exhibit gut purge and show wandering behaviour. The increasing trend of hormone titres continued on day 5, the prepupal stage when it rose steeply to reach its maximum (2524 ± 453 pg/µl). Only the peak on day 5 was significant (P < 0.05).



2 i i 🕅



(b) Changes in haemolymph ecdysteroid titres of sixth instar larvae at 6 h intervals

Ecdysteroid titres of sixth instar larvae from 0 h to 132 h post ecdysis are shown in Fig. 8. In the newly moulted (white head capsule) sixth instar larvae (day 0), titres were just under 15 pg/µl namely, 11 ± 4 pg/µl, 9 ± 3 pg/µl, and $4 + 0 \text{ pg/}\mu\text{l}$ at 0 h, 6h and 12 h respectively. The small peak (Peak I) (21 ± 3 pg/µl) observed at 18 h (day 1) was significant at P < 0.05 (relative to ecdysteroid titres at 12 and 24 h). The titres then dropped to 6 ± 3 pg/µl at 24h, increased slightly at 30 h (7 \pm 3 pg/µl) and then showed a small peak (Peak II) at 36 h ($16 \pm 5 \text{ pg/}\mu$). There was a reduction in titres on day 2, at 42 h and 48 h when the titres were 10 ± 3 pg/µl and 4 ± 1 pg/µl respectively. A gradual rise occurred at 54 h $(10 \pm 2 \text{ pg/}\mu\text{l})$ leading to a significant peak (Peak III) at 60 h (47 ± 17 pg/µl) (P < 0.1, relative to titres at 54 and 78 h). After the 60 h stage, the titres gradually decreased on day 3 at 66 h (19 \pm 9 pg/µl), increased slightly at 72 h (25 ± 0 pg/µl) and then declined at 78 h and 84 h to $10 \pm 2 \text{ pg/}\mu\text{l}$ and $15 \pm 2 \text{ pg/}\mu\text{l}$. The hormone profile increased considerably on day 4 at 90 h and 96 h to 58 ± 15 pg/µl and 278 ± 56 pg/µl respectively and developed a shoulder at 102 h (156 \pm 25 pg/µl). The larvae purged their guts at 96 h. An increase in hormone titres was observed towards the end of day 4 at 108 h ($492 + 78 \text{ pg/}\mu$). This was the first instance in the sixth instar larval stage when hormone titres increased beyond 100 pg/µl haemolymph.



ł

SAR

GP – Gut purge ; HCP – Head critical period

Hormone titres rose steeply thereafter and reached a significant peak on day 5 at 114 h (1924 \pm 324 pg/µl) (P < 0.05, relative to titres at 84 and 120 h). This formed the premoult peak (Peak IV) in *S. mauritia* after which a decrease was seen at 120 h (243 \pm 48 pg/µl) and then a small but significant peak (Peak V) at 126 h (638 \pm 85 pg/µl) (P < 0.05, relative to titres at 120 and 132 h) which decreased at 132 h (336 \pm 34 pg/µl).

(ii) Haemolymph ecdysteroid titres during pupal stadium

Ecdysteroid titres of haemolymph from insects 0 - 7 days post-pupation are shown in Fig. 9. The haemolymph titres were low in the newly moulted pupae (day 0, 686 ± 63 pg/µl). Titres increased on day 1 (1973 ± 521 pg/µl) after which it rose rapidly and formed a significant peak on day 2 (14221 ± 1458 pg/µl, P < 0.05 relative to ecdysteroid titres on day 1 and P < 0.1 relative to ecdysteroid titres on day 5). Thereafter, hormone titres decreased gradually on day 3 (7831 ± 478 pg/µl), day 4 (5915 ± 1159 pg/µl), day 5 (894 ± 74 pg/µl), day 6 (574 ± 186 pg/µl) and day 7 (467 ± 185 pg/µl). Titres reached a minimum on day 7 which was almost the same as that on day 0.

3.3.2 Determination of head critical period (HCP) during larval-pupal transformation

This study was undertaken to assess the temporal patterns of PTTH secretion during larval-pupal transformation of *S. mauritia*. Sixth instar



ł



15

larvae were 'neck-ligated' at six hour intervals from the time of ecdysis to the sixth instar to remove the endogenous sources of cerebral neurosecretory factors and JH. The ligated larvae were then closely observed for signs of gut purge and pupal cuticle secretion.

The duration of development of normal, unligated sixth instar larvae of S. mauritia was 6 ± 0 days. Sixth instar larvae at the termination of the phagoperiod exhibited gut purge which occurred 95 ± 1 hours after ecdysis. These larvae underwent pupation 48 h after gut purge. These developmental phenomena, i.e. gut purge and pupation have been utilised to assess the temporal patterns of PTTH secretion during larval-pupal transformation. Fig. 10 shows the effect of neck ligation upon the timing of gut purge and pupation. Larvae ligated within 24 h showed no developmental response during their survival of seven days. Of the larvae ligated at 30 h and 36 h, 70% showed gut purge after 4.5 ± 0.5 days. The incidence of gut purged larvae was 100% when ligations were performed at 42 h, 48 h, 54 h, 60 h, 66h, 72 h, 78 h, 84 h, and 90 h. The time taken for gut purge was broadly distributed over a period of 5-7 days. The neck-ligated larvae survived for 11 + 1 days but never underwent pupation. Larvae neck-ligated at 96 h survived for 12 ± 1 days but never pupated. Post-feeding larvae neck-ligated at 102 h and 108 h after ecdysis underwent pupation in 7 ± 1 days. Occurrence of pupation was 100% when prepupae were neck-ligated. Prepupae neck-ligated at 120 h,
590

ł



Fig. 10 The timing of gut purge (open column) and pupation (black) in sixth instar larvae neck-ligated at (A) 24h (B) 30h (C) 36h (D) 42h (E) 48h (F) 54h (G) 60h (H) 66h (I) 72h (J) 78h (K) 84h (L) 90h (M) 96h (N) 102h (O) 108h (P) 114h (Q) 120h (n=30) 126 h, 132 h and 138 h underwent pupation in 6 ± 0 days. The tentative timings of the two PTTH releases (HCP) are illustrated in Fig. 8.

3.3.3 Effects of ligations of sixth instar larvae on metamorphosis and haemolymph ecdysteroid titres

This study was undertaken to investigate the role of brain factors and prothoracic gland secretions on metamorphosis and haemolymph ecdysteroid titres of *S. mauritia*. In one set of experiments, sixth instar larvae of day 1 and day 4 were 'neck-ligated' to remove the endogenous source of cerebral neurosecretory factors and JH. In another set of experiments, sixth instar larvae of day 1 and day 4 were 'thorax-ligated' to eliminate the endogenous sources of both head factors (cerebral neurosecretory factors and JH) and prothoracic gland secretions. Neck-/thorax-ligated larvae were closely observed for signs of gut purge/pupal cuticle secretion.

(i) Effects of ligations of sixth instar day 1 larvae on metamorphosis and haemolymph ecdysteroid titres

Earlier studies have suggested that the head critical period for gut purge occurred between 24 h and 36 h on day 1 followed by the commitment peak of ecdysteroids on day 3. Therefore in this experiment, sixth instar larvae were neck-/thorax-ligated on day 1 and the effects on metamorphosis and haemolymph ecdysteroid titres were studied. Seventy percent of larvae

Nature of ligation n	n	Days of survival	% of surviving larvae which showed:			
		$(Mean \pm SD)$	No change	Gut purge	Pupation	
Neck-ligated	25	9 <u>+</u> 1	30	70		
Thorax- ligated	25	9 <u>+</u> 1	30	70		
Unligated (Control)	15 each	6 <u>+</u> 0		100	100	

Table 1. Effects of ligations of sixth instar day 1 larvae onmetamorphosis

Nature of	n	Days of survival	% of surviving larvae which showed:		
ngation		(Mean <u>+</u> SD)	No change	Pupation	
Neck-ligated	25	4 <u>+</u> 1	100		
Thorax-ligated	25	4 <u>+</u> 1	100		
Unligated (Control)	15 each	2 <u>+</u> 0		100	

Table 2. Effects of ligations of sixth instar day 4 larvae onmetamorphosis

neck-/thorax-ligated on day 1 exhibited gut purge. Although the ligated larvae survived for 9 ± 1 days, none of them pupated while all the control larvae underwent gut purge and became pupae on day 6 (Table 1).

Ecdysteroid titres of experimental and control larvae from days 1-4 are shown in Fig. 11. Hormone titres in the neck-ligated larvae for day 1, day 2, day 3 and day 4 were 8 ± 2 pg/µl, 5 ± 1 pg/µl, 5 ± 1 pg/µl and 6 ± 1 pg/µl haemolymph respectively. Ecdysteroid titres for thorax-ligated larvae on day 1, day 2, day 3 and day 4 were 6 ± 1 pg/µl, 5 ± 0 pg/µl, 4 ± 2 pg/µl and 2 ± 1 pg/µl haemolymph respectively. Hormone titres of larvae kept as controls were 15 ± 2 pg/µl, 26 ± 5 pg/µl, 39 ± 20 pg/µl and 372 ± 13 pg/µl haemolymph on day 1, day 2, day 3 and day 4 respectively. Neck/thorax ligation of day 1 sixth instar larvae induced a significant reduction in haemolymph ecdysteroid titres (P < 0.1) on all days when compared to those of control larvae. Control larval titres exhibited an increasing trend from days 1-4.

(ii) Effects of ligations of sixth instar day 4 larvae on metamorphosis and haemolymph ecdysteroid titres

Earlier studies have clearly demonstrated that the head critical period for moulting in post-feeding larvae occurs during day 4 followed by the moulting peak of ecdysteroids on day 5. Hence it was thought worthwhile to investigate the effects of ligation of sixth instar day 4 larvae on larval-pupal



Fig.11 Effects of neck/thorax ligation of day 1 larvae on haemolymph ecdysteroid titres

metamorphosis and haemolymph ecdysteroid titres. Neck/thorax ligation of day 4 larvae resulted in a complete inhibition of pupation. The ligated larvae survived for 4 ± 1 days and died without pupating while control larvae underwent pupation on day 6 (Table 2).

Ecdysteroid titres of experimental and control larvae from days 4-8 are shown in Fig. 12. Hormone titres of neck-ligated larvae were 330 + 29 pg/µl on day 4, 385 ± 69 pg/µl on day 5, 570 ± 195 pg/µl on day 6, 989 ± 377 pg/µl on day 7 and 1785 ± 453 pg/µl on day 8. Ecdysteroid titres of thorax-ligated larvae were 309 ± 151 pg/µl, 315 ± 119 pg/µl, 124 + 29 pg/µl, 145 + 53 pg/µl and 148 ± 8 pg/µl on day 4, day 5, day 6, day 7 and day 8 respectively. Ecdysteroid titres of control unligated larvae on the other hand were 372 ± 13 pg/µl on day 4 and 2524 ± 453 pg/µl on day 5. Neck ligation of day 4 larvae does not prevent the normal increase in haemolymph ecdysteroid titres. Haemolymph ecdysteroid peak on day 8 in the neck-ligated larvae was not significantly different from the ecdysteroid peak on day 5 in the control larvae. Thorax ligation led to a significant reduction (P < 0.05) in haemolymph ecdysteroid titres when compared to that of control larvae on day 5. Conversely, titres in the control (unligated) larvae increased significantly (P < 0.05) to reach its peak on day 5 and pupation occurred on day 6.



Fig. 12 Effects of neck/thorax ligation of day 4 larvae on haemolymph ecdysteroid titres

		0/	Number of larvae transformed to:		
Dosage	Dosage n ⁷⁰ mortality super		Imperfect supernumerary larvae	Pupae	
20 µg	25	20	20		
5 µl Acetone (Control)	15			15	

Table 3.	Morphogenetic effects of repeated daily treatments of sixth
	instar larvae with JHA from day 0 onwards

3.3.4 Effects of treatments of JHA on metamorphosis and haemolymph ecdysteroid titres

Juvenile hormone (JH) has been known to be a necessary factor for the maintenance of larval state and also prevent metamorphosis in insects (Wigglesworth, 1934, Williams, 1961). The transformation of larvae into adults in exopterygote insects as well as the transformation of pupae into adults in endopterygote insects is affected by JH or its analogues. The interplay of ecdysteroids and juvenile hormones in larval insects serve to orchestrate the progression from one developmental state to the next, with ecdysteroids regulating the onset and timing of the moulting cycle and JH directing the quality of the moult (Riddiford, 1994). The present study analyses the effect of JHA treatment on morphogenesis and haemolymph ecdysteroid titres during larval-pupal and pupal-adult development.

(i) Effects of repetitive treatments of unligated sixth instar larvae with JHA on metamorphosis and ecdysteroid titres

Sixth instar day 0 larvae were topically treated with repeated daily dose of 20 μ g JHA from day 0 to day 3. Control larvae received an equal volume of acetone. The effects of these treatments on larval development, metamorphosis and haemolymph ecdysteroid titres were carefully studied.

Nature of	Decorre	_	% of surviving larvae which showed :			
ligation	Dosage	n	No change	Gut purge	Pupatio n	
Neck-ligated	5 µg	20	100			
Neck-ligated (Control)	5 μl Acetone	10	30	70		
Thorax-ligated	5 µg	20	100			
Thorax-ligated (Control)	5 μl Acetone	10	30	70		

Table 4. Morphogenetic effects of repeated daily treatments of JHA tosixth instar day 1 ligated larvae

Morphogenetic effects

The JHA treated larvae were similar to control larvae in appearance but they fed voraciously and were considerably larger in size. These larvae moulted into imperfect supernumerary larvae within 24 h of termination of hormone treatment, i.e. on day 4. The supernumerary larvae were more or less sluggish, with reduced feeding activity. Head and thoracic region showed pupal characters whereas the integument in general and the abdominal region were larval in appearance. Antennae showed signs of segmentation and were Thoracic legs and the mouth parts were mildly sclerotised sclerotised. especially the labio-hypopharyngeal complex. The paired wing discs on mesoand metathoracic segments were partially tanned. These supernumerary larvae survived for 6-7 days. Mortality rate was less after JHA treatments. Some of the larvae became prepupae but failed to pupate. Control larvae purged their guts and transformed into wandering stage by the end of day 4. These larvae transformed to prepupae at the end of day 5 which pupated after 24 h, on day 6 (Table 3).

Changes in ecdysteroid titres

Ecdysteroid titres of experimental and control larvae from day 1 - day 4 and that of day 0 supernumerary larvae are represented in Fig. 13. Haemolymph ecdysteroid titres of JHA treated larvae measured 14 ± 3 pg/µl, 36 ± 8 pg/µl, 28 ± 19 pg/µl and 67 ± 19 pg/µl for day 1, day 2, day 3 and day 4



Fig.13 Effects of repetitive treatments of larvae with 20 µg JHA from day 0 onwards on haemolymph ecdysteroid titres respectively. In the case of day 0 supernumerary larvae the haemolymph ecdysteroid titres measured 2 ± 1 pg/µl. Ecdysteroid titres of control (normal) larvae measured 15 ± 2 pg/µl (day 1), 26 ± 5 pg/µl (day 2), 39 ± 20 pg/µl (day 3), 372 ± 13 pg/µl (day 4) and 2524 ± 453 pg/µl (day 5). Ecdysteroid titres increased gradually to reach its peak on day 4 and decreased to very low titres in the supernumerary larvae (day 0) in the JHA treated larvae. Conversely, titres in the control larvae increased steadily from days 1-5 to reach its peak on day 5. The peak titre of JHA treated larvae was significantly reduced (P < 0.05) when compared to that of control larvae.

(ii) Effects of repetitive treatments of ligated sixth instar day 1 larvae with JHA on metamorphosis and ecdysteroid titres

Sixth instar larvae were neck- or thorax-ligated on day 1 and topically treated with a repetitive daily dose of 5 μ g JHA for three days. Ligated larvae kept as controls were treated with an equivalent quantity of acetone. The effects of these treatments on metamorphosis and haemolymph ecdysteroid titres were carefully studied.

Morphogenetic effects

The JHA treated larvae did not show the premetamorphic behaviour of gut purge while 70% of the control larvae exhibited gut purge after 4.5 ± 0.5 days. None of the JHA treated or control larvae pupated (Table 4).

Nature of	Dosage	n	% of surviving larvae which showed:		
ingation			No change	Pupation	
Neck-ligated	5 µg	20	20	80	
Neck-ligated (Control)	5 μl Acetone	10	100		
Thorax- ligated	5 µg	20	100		
Thorax- ligated (Control)	5 μl Acetone	15	100		

Table 5. Morphogenetic effects of treatments of JHA tosixth instar day 4 ligated larvae

Changes in ecdysteroid titres

Ecdysteroid titres of experimental and control larvae are represented in Fig. 14. Titres of JHA treated neck-ligated and thorax-ligated larvae on day 4 were 4 ± 2 pg/µl and 2 ± 1 pg/µl respectively. In the insects kept as control, the hormone titres on day 4 of the neck-ligated and thorax-ligated larvae were 6 ± 1 pg/µl and 2 ± 1 pg/µl respectively. Ecdysteroid titres of the JHA treated and control larvae were not significantly different.

(iii) Effects of treatments of ligated sixth instar day 4 larvae with JHA on metamorphosis and ecdysteroid titres

Sixth instar larvae were neck- or thorax-ligated on day 4 and topically treated with a single dose of 5 μ g JHA. Ligated larvae kept as control received an equivalent volume of acetone. The effects of these treatments on metamorphosis and haemolymph ecdysteroid titres were carefully studied.

Morphogenetic effects

JHA treatment of larvae neck-ligated on day 4 accelerated the moulting process and the larvae underwent pupal development. Eighty percent of ligated larvae treated with JHA transformed into headless pupae on day 7. Neck-ligated day 4 larvae kept as control showed a complete inhibition of pupation. The ligated larvae survived for 4 ± 1 days and died without pupating (Table 5).

669



Day 4

Fig.14 Effects of treatment of JHA to neck/thorax-ligated day 1 larvae on haemolymph ecdysteroid titres

1

Larvae thorax-ligated on day 4 failed to pupate even after treatment of JHA. The larvae survived for 4 ± 1 days. Thorax-ligated larvae kept as controls also showed a complete inhibition of pupation (Table 5).

Changes in ecdysteroid titres

Ecdysteroid titres of neck-ligated larvae treated with JHA and those of control larvae are represented in Fig. 15. Hormone titres of day 0 headless pupae were 1760 ± 453 pg/µl whereas that of control (neck-ligated) larvae were 989 ± 377 pg/µl on day 7. Titres of the headless pupae were higher than that of the control larvae although the increase was not significant.

Ecdysteroid titres of thorax-ligated larvae treated with JHA and those of control larvae are represented in Fig. 16. Hormone titres of the treated larvae on day 5, day 7 and day 8 were 240 ± 90 pg/µl, 84 ± 14 pg/µl and 89 ± 26 pg/µl respectively. Titres of the control larvae were 315 ± 119 pg/µl, 145 ± 53 pg/µl and 148 ± 8 pg/µl on day 5, day 7 and day 8 respectively. Ecdysteroid titres of the JHA treated and control larvae were not significantly different and both exhibited a declining trend.

(iv) Effects of treatments of pupae with JHA on metamorphosis and ecdysteroid titres

Newly ecdysed (day 0) pupae were treated with a single dose of 0.5 μ g or 1 μ g JHA. Control pupae received an equivalent quantity of acetone. The

Dosage	n	% mortality	No. of pupae transformed into normal adults
0.5 µg	25	18	
1 μg	25	20	
1 μl Acetone (Control)	20	10	18

· · · · · ·

Table 6. Morphogenetic effects of treatments of JHA today 0 pupae

.





Fig. 16 Effects of treatment of 5 μg JHA to thorax-ligated day 4 larvae on haemolymph ecdysteroid titres

 \mathcal{Q}

effects of these treatments on pupal development, metamorphosis and haemolymph ecdysteroid titres were carefully studied.

Morphogenetic effects

High mortality was observed in the treated pupae. Surviving pupae showed various abnormalities in pupal-adult development. Failure of emergence or the emergence of abnormal adults were observed. None of them emerged successfully as morphologically normal adults. When the pupal cases of such unemerged pupae were removed, they were found to contain pupaladult intermediates or adultoids. These adultoids showed normal imaginal differentiation in the head and thorax. However, the wings although developed were unstretched. Moths which emerged from the control pupae were normal and healthy (Table 6).

Changes in ecdysteroid titres

Ecdysteroid titres of experimental and control pupae from days 1 - 6 are shown in Fig. 17. Hormone titres in the pupae treated with 0.5 μ g JHA were $4819 \pm 221 \text{ pg/}\mu\text{l}$ on day 1, $8968 \pm 2928 \text{ pg/}\mu\text{l}$ on day 2, $14996 \pm 1878 \text{ pg/}\mu\text{l}$ on day 3, $8231 \pm 963 \text{ pg/}\mu\text{l}$ on day 4, $1922 \pm 129 \text{ pg/}\mu\text{l}$ on day 5 and $1833 \pm 268 \text{ pg/}\mu\text{l}$ on day 6. Titres in the pupae treated with 1 μ g JHA were $5107 \pm 2191 \text{ pg/}\mu\text{l}$ on day 1, $17679 \pm 1980 \text{ pg/}\mu\text{l}$ on day 2, $10010 \pm 1506 \text{ pg/}\mu\text{l}$ on day 3, $6831 \pm 748 \text{ pg/}\mu\text{l}$ on day 4, $4053 \pm 916 \text{ pg/}\mu\text{l}$ on day 5 and

62 P



.

Fig. 17 Effects of treatments of pupae with 0.5 and 1 /ug JHA on haemolymph ecdysteroid titres

 $3881 \pm 538 \text{ pg/}\mu \text{l}$ on day 6. Ecdysteroid titres of pupae kept as control were $1973 \pm 521 \text{ pg/}\mu \text{l}$ on day 1, $14221 \pm 1458 \text{ pg/}\mu \text{l}$ on day 2, $7831 \pm 478 \text{ pg/}\mu \text{l}$ on day 3, $5915 \pm 1159 \text{ pg/}\mu \text{l}$ on day 4, $894 \pm 74 \text{ pg/}\mu \text{l}$ on day 5 and $574 \pm 186 \text{ pg/}\mu \text{l}$ on day 6. Although the pattern of fluctuation was similar in the JHA treated and control pupae, titres of the 0.5 µg JHA treated pupae were significantly higher (P < 0.1) than those of control pupae on day 1, day 5 and day 6 while those of 1 µg JHA treated pupae were higher than those of controls on day 1, day 2, day 3, day 5 and day 6.

3.3.5 Effects of treatments of ecdysteroid agonist RH 5992 on metamorphosis and haemolymph ecdysteroid titres

Ecdysteroids are known to have positive/negative feedback effects on endogenous ecdysteroid biosynthesis (Sakurai and Williams, 1989). Hence it was thought worthwhile to investigate the effects of non-steroidal ecdysone agonist RH 5992 on metamorphosis and haemolymph ecdysteroid titres.

(i) Morphogenetic effects of treatments of unligated sixth instar larvae with RH 5992

Sixth instar larvae on day 0 and day 4 were treated topically with different doses of RH 5992 (2 μ g, 5 μ g). Control larvae received an equivalent volume of acetone. The effects of these treatments on metamorphosis and mortality were observed. The RH 5992 treated day 0 sixth instar larvae

Dosage	n	% of surviving larvae which showed:			% of larvae showing premature lethal
		No change	Gut purge	Pupation	larval moult
2 µg	10		100		100
5 µg	10		100		100
5 μl Acetone (Control)	5		100	100	

Table 7. Morphogenetic effects of treatments of RH 5992 tosixth instar day 0 larvae

Table 8. Morphogenetic effects of treatments of RH 5992 tosixth instar day 4 larvae

Dosage	n	% of surviving larvae showing abnormal and lethal pupation	% of larvae showing normal pupation
2 µg	10	100	
5 µg	10	100	
5 μl Acetone (Control)	5		100

showed symptoms of a prematurely induced larval moult. Within 24 h of treatment, the old head capsule had slipped down, revealing a fragile and often unsclerotised new head capsule underneath. Formation of a new larval cuticle was also observed. During this period, larval feeding and weight gain was significantly suppressed and the larvae underwent gut purge prematurely. In some cases loss of haemolymph and extrusion of the hindgut was also seen. Most treated larvae died in their old cuticle shortly afterwards. In some cases, the old cuticle was partially shed, with remnants remaining attached to the new cuticle. Likewise, rupture of the imperfectly formed new cuticle was sometimes observed. The RH 5992 treated larvae never showed pupation or adult emergence. Control larvae purged their guts and transformed into wandering stage by the end of day 4. These larvae underwent normal pupation on day 6 (Table 7).

The post feeding (day 4) sixth instar larvae treated with different doses of RH 5992 underwent abnormal and lethal pupation within 24-48 h. They were unable to ecdyse successfully or to build up a normal pupal cuticle. Control larvae underwent normal pupation on day 6 (Table 8).

(ii) Effects of treatments of sixth instar day 1 ligated larvae with RH 5992

Day 1 sixth instar larvae were neck- or thorax-ligated and treated topically with a single dose of 2 μ g or 5 μ g RH 5992. Ligated larvae kept as

Nature of	Dosage	n	% of surviving larvae which showed:			
ligation	Dosage	10	No change	Gut purge	Pupation	
Neck-ligated	2 µg	20		100		
Neck-ligated	5 µg	20		100		
Neck ligated (Control)	5 μl Acetone	15 each	30	70		
Thorax- ligated	2 µg	20		100		
Thorax- ligated	5 µg	20		100		
Thorax- ligated (Control)	5 μl Acetone	15 each	30	70		

Table 9. Morphogenetic effects of treatments of RH 5992 tosixth instar day 1 ligated larvae

control were treated with an equivalent quantity of acetone only. The effects of these treatments on metamorphosis and haemolymph ecdysteroid titres were carefully studied.

Morphogenetic effects

Ligated (day 1) larvae treated with various dosages of RH 5992 survived for 3 ± 1 days while larvae kept as control survived for 9 ± 1 days. However, both failed to pupate. All larvae treated with RH 5992 exhibited the premetamorphic behaviour of gut purge after 2.5 ± 0.5 days. Seventy percent of the control larvae also exhibited gut purge behaviour after 4.5 ± 0.5 days (Table 9). An imperfectly formed new cuticle was also observed beneath the partially shed, ruptured old cuticle.

(a) Changes in ecdysteroid titres of neck-ligated larvae treated with RH 5992

Ecdysteroid titres of neck-ligated larvae treated with RH 5992 and those of control larvae are represented in Fig. 18. Titres of neck-ligated larvae treated with 2 µg RH 5992 were 5 ± 1 pg/µl, 3 ± 1 pg/µl and 70 ± 21 pg/µl for day 1, day 2 and day 3 respectively. Those of 5 µg RH 5992 treated larvae were 8 ± 1 pg/µl, 8 ± 3 pg/µl and 42 ± 5 pg/µl for day 1, day 2 and day 3 respectively. Ecdysteroid titres of larvae kept as control were 8 ± 2 pg/µl, 5 ± 1 pg/µl and 5 ± 1 pg/µl for day 1, day 2 and day 3 respectively. Hormone





Fig. 18 Effects of treatments of RH 5992 to neck-ligated day larvae on haemolymph ecdysteroid titres

Nature of ligation	D		% of surviving larvae which transformed into:			
	Dosage	n	No change	Pupae	Larval-pupal intermediates	
Neck-ligated	2 μg	20	20		80	
Neck-ligated	5 µg	20	20		80	
Neck-ligated	10 µg	20	20		80	
Neck-ligated (Control)	5 μl Acetone	15 each	100			
Thorax- ligated	2 µg	20	20		80	
Thorax- ligated	5 µg	20	20		80	
Thorax- ligated (Control)	5 μl Acetone	15 each	100			

.

Table 10.Morphogenetic effects of treatments of RH 5992 to sixthinstar day 4 ligated larvae

titres that were reduced following neck ligation decreased further though not significantly in the 2 μ g RH 5992 treated larvae but remained unchanged in the 5 μ g RH 5992 treated larvae on day 1 and day 2. On day 3 an abrupt and significant increase (P < 0.05) in titres occurred both in the 2 μ g and 5 μ g RH 5992 treated larvae when compared to those of control larvae which remained low on all days.

(b) Changes in ecdysteroid titres of thorax-ligated larvae treated with RH 5992

Ecdysteroid titres of thorax-ligated larvae treated with RH 5992 and that of control larvae are represented in Fig. 19. Titres of thorax-ligated larvae treated with 2 µg RH 5992 were 11 ± 2 pg/µl, 7 ± 3 pg/µl and 3 ± 0 pg/µl for day 1, day 2 and day 3 respectively. Those of 5 µg RH 5992 treated larvae were 8 ± 2 pg/µl, 4 ± 0 pg/µl and 6 ± 2 pg/µl for day 1, day 2 and day 3 respectively. Ecdysteroid titres of larvae kept as control were 6 ± 1 pg/µl, 5 ± 0 pg/µl and 4 ± 2 pg/µl for days 1-3. RH 5992 treatment did not cause a significant change in the titres of the treated larvae when compared to those of the control larvae.



Fig. 19 Effects of treatments of RH 5992 to thorax-ligated day 1 larvae on haemolymph ecdysteroid titres

(iii) Effects of treatments of sixth instar day 4 ligated larvae with RH 5992

Day 4 sixth instar larvae were neck- or thorax-ligated and treated topically with a single dose of 2 μ g, 5 μ g or 10 μ g RH 5992. Ligated larvae kept as control were treated with an equivalent quantity of acetone only. The effects of these treatments on metamorphosis and haemolymph ecdysteroid titres were carefully studied.

Morphogenetic effects

Ligated (day 4) larvae treated with various dosages of RH 5992 transformed into larval-pupal intermediates on day 6. They survived for 3 ± 1 days. These intermediates had thoracic legs which were larval in appearance. Pupal cuticle was seen beneath the old larval one in the abdominal region. The treated larvae were unable to ecdyse successfully or to build up a normal pupal cuticle. Neck-/thorax-ligated day 4 larvae kept as control showed a complete inhibition of pupation, they survived for 4 ± 1 days and then died (Table 10).

(a) Changes in ecdysteroid titres of neck-ligated larvae treated with RH 5992

Ecdysteroid titres of neck-ligated larvae treated with RH 5992 and those of control larvae are represented in Fig. 20. Titres of neck-ligated



Fig. 20 Effects of treatments of RH 5992 to neck-ligated day 4 larvae on haemolymph ecdysteroid titres

larvae treated with 2 μ g RH 5992 were 224 ± 49 pg/ μ l, 413 ± 43 pg/ μ l, 269 ± 34 pg/µl and 222 ± 101 pg/µl on days 4 and 5 of sixth instar larval stage and day 0 and day 1 of larval-pupal intermediate stage respectively. Those of 5 µg RH 5992 treated larvae were 76 \pm 12 pg/µl, 161 \pm 23 pg/ μ l, 242 ± 39 pg/µl and 148 ± 13 pg/µl on days 4 and 5 of sixth instar larval stage and day 0 and day 1 of larval-pupal intermediate stage respectively. Titres of 10 µg RH 5992 treated larvae were 186 ± 17 pg/µl, 303 ± 10 pg/ul. 218 ± 46 pg/µl and 5 ± 3 pg/µl on day 4 and day 5 of sixth instar larval stage and day 0 and day 1 of larval-pupal intermediate stage respectively. The control larvae had the following titres for days 4 - 7: $330 \pm 29 \text{ pg/}\mu\text{l}$, $385 \pm 69 \text{ pg/}{\mu}$, 570 $\pm 195 \text{ pg/}{\mu}$ and 989 $\pm 377 \text{ pg/}{\mu}$. The hormone titres decreased significantly (P < 0.1) on most days following RH 5992 treatment when compared to control larvae which exhibited an increasing trend of ecdysteroids reaching its maximum on day 7.

(b) Changes in ecdysteroid titres of thorax-ligated larvae treated with RH 5992

Ecdysteroid titres of thorax-ligated larvae treated with RH 5992 and that of control larvae are represented in Fig. 21. Titres of thorax-ligated larvae treated with 2 µg RH 5992 were 328 ± 90 pg/µl, 90 ± 20 pg/µl, 165 ± 31 pg/µl and 113 ± 33 pg/µl on day 4 and day 5 of sixth instar larval stage and day 0 and day 1 of larval-pupal intermediate stage respectively.

Dosage	n	% Mortality	% of adult emergence	% of unemerged adultoids
5 µg	25	20	20	60
10 µg	25	20	20	60
5 µl Acetone (Control)	20	10	90	

.

,

Table 11. Effects of treatments of RH 5992 to day 0 pupae
aue



Fig. 21 Effects of treatments of RH 5992 to thorax-ligated day 4 larvae on haemolymph ecdysteroid titres

Those of 5 µg RH 5992 treated larvae were 386 ± 95 pg/µl, 84 ± 6 pg/µl and 16 ± 1 pg/µl on day 4 and day 5 of sixth instar larval stage and day 0 of larvalpupal intermediate stage respectively. The control larvae were 309 ± 151 pg/µl, 315 ± 119 pg/µl, 124 ± 29 pg/µl and 145 ± 53 pg/µl on day 4, day 5, day 6 and day 7 respectively. The hormone titres decreased considerably both in the RH 5992 treated and control larvae and there was no significant difference on most days between the treated and control larval titres.

(iv) Effects of treatment of pupae with RH 5992

Newly ecdysed (day 0) pupae were treated topically with a single dose of 5 μ g or 10 μ g RH 5992. Control pupae received an equivalent quantity of acetone. The effects of these treatments on pupal development, metamorphosis and haemolymph ecdysteroid titres were carefully studied.

Morphogenetic effects

All the treated pupae survived the pupal period like the control pupae. Mortality was observed towards the end of the pupal period. Most of the treated pupae died on or before the day of adult emergence but a few emerged as normal adults. The pupal period of the RH 5992 treated pupae was found slightly prolonged (9 \pm 1 days) whereas adults emerged from the control pupae on day 8 (Table 11).





Fig. 22 Effects of treatments of pupae with 5 , ug and 10 , ug RH 5992 on haemolymph ecdysteroid titres

27

Changes in ecdysteroid titres

Ecdysteroid titres of pupae treated with RH 5992 and those of control pupae are represented in Fig. 22. Titres of pupae treated with 5 µg RH 5992 were 3007 ± 1458 pg/µl and 2629 ± 1499 pg/µl on day 2 and day 5 respectively. Titres of pupae treated with 10 µg RH 5992 were 2087 ± 761 pg/µl and 1674 ± 390 pg/µl on day 1 and day 5 respectively. Ecdysteroid titres of control pupae on day 2 and day 5 were 14221 ± 1458 pg/µl and 894 ± 74 pg/µl respectively. Ecdysteroid titres in the treated pupae decreased significantly (P < 0.05) on day 2 when compared to the control pupae had titres significantly higher (P < 0.1) than those of the control pupae.

3.4 DISCUSSION

Developmental profile of ecdysteroids during larval development and metamorphosis

The primary source of ecdysteroids in most immature insects is the prothoracic glands (Chino *et al.*, 1974; King *et al.*, 1974). They produce a prohormone, ecdysone which is converted to the actual hormone, 20-hydroxyecdysone in the peripheral tissues (King, 1972). Both ecdysone, 20-hydroxyecdysone and their hormonally active metabolites are inactivated, biochemically and/or depleted by excretion (Koolman, 1978). Their content, i.e. ecdysteroid titre in the body obviously depends on the balance between the rate of ecdysone secretion and the rate of ecdysone metabolism. Information regarding the ecdysteroid profile of insects is vital to the understanding of endocrine regulation and interaction during postembryonic development. Because of vast differences in the time sequence of developmental events occurring in insects, ecdysteroid titres obtained from one species may not accurately represent the same event found in another species. Consequently, ecdysteroid titre determinations for individual species are essential if critical evaluation of titre changes and inference of its significance are required. Hence the ecdysteroid titres during the period between last larval instar and adult of *S. mauritia* has been determined. The relationships between the ecdysteroid titres and developmental events during larval development and metamorphosis are also analysed.

Haemolymph ecdysteroid profile during larval-pupal development

Changes in ecdysteroid titres during development of insects have been determined for a number of lepidopteran species (Burdette, 1962; Shaaya and Karlson, 1965a; Kaplanis, et al., 1966a; Hanaoka and Ohnishi, 1974; Bollenbacher et al., 1975, 1978; Calvez et al., 1976; Lagaeux et al., 1976; Dean et al., 1980; Fujishita et al., 1982; Loeb, 1982; Gelman and Woods, 1983; Kiguchi, 1983; Gelman and Brents, 1984; Yin and Chaw, 1984; Kiguchi et al., 1985; Okuda et al., 1985; Smith, 1985; Wolfgang and Riddiford, 1986; Kelly et al., 1986; Sakurai et al., 1998).

Haemolymph ecdysteroid profile of last larval stadium of S. mauritia is consistent with previous determinations in other lepidopteran insects. Ecdysteroid titres reach a small peak on day 3 and form a significant peak on day 5 during larval-pupal development. Histological and ultrastructural studies of prothoracic glands of S. mauritia also suggest two phases of secretory activity during last instar larval development (Balamani and Nair, 1994, 1997). Biphasic secretion of ecdysone during larval-pupal development has also been reported in Manduca sexta (Bollenbacher et al., 1975), Samia cynthia (Calvez et al., 1976; Fujishita et al., 1982), Galleria mellonella (Bollenbacher et al., 1978), Calpodes ethlius (Dean et al., 1980), Heliothis virescens (Loeb, 1982), Ephestia kuhniella (Kallenborn and Mosbacher, 1983), Ostrinia nubilalis (Gelman and Brents, 1984), Bombyx mori (Okuda et al., 1985), Trichoplusia ni (Jones et al., 1986) and Lymantria dispar (Kelly et al., 1986).

The first small peak was observed on day 3 of the sixth instar larvae. This peak is interpreted as a signal for the change in developmental commitment (reprogramming) causing the epidermal cells to synthesise a pupal cuticle when the high ecdysteroid titre peak preceding the pupal moult occurs (Riddiford, 1976; Bollenbacher *et al.*, 1981). This commitment peak causes the termination of feeding and the induction of both physiological (gut purge) and behavioural (wandering) syndromes which occur 24 h after the small surge (Truman and Riddiford, 1974; Fujishita *et al.*, 1982). Histological and ultrastructural observation of prothoracic glands of *S. mauritia* also reveal secretory activity on day 3 of last larval instar (Balamani and Nair, 1994, 1997).

The ecdysteroid titres increased exponentially on day 5 of the sixth instar to form a significant peak. Prothoracic glands of *S. mauritia* also showed high secretory activity on day 5 of the last larval instar (Balamani and Nair, 1994, 1997). A similar surge of ecdysteroid titre was observed in all the other lepidopteran insects listed earlier. This ecdysteroid peak was interpreted to be responsible for the initiation of apolysis and pharate pupal development (Riddiford, 1976). Daily sampling of haemolymph did not reveal smaller peaks which have been reported in other insects (Dean *et al.*, 1980; Agui and Hiruma, 1982; Gelman and Brents, 1984; Sakurai *et al.*, 1998). However six hour sampling of haemolymph revealed the presence of additional peaks.

Haemolymph ecdysteroid titres at six hour intervals during larval-pupal development

With 6 h sampling of haemolymph, five significant ecdysteroid peaks were detected during larval-pupal development. Only four peaks have been reported so far in C. ethlius (Dean et al., 1980), M. brassicae (Agui and Hiruma, 1982), O. nubilalis (Gelman and Brents, 1984), M. sexta (Wolfgang and Riddiford, 1986). In B. mori, a small increase in the haemolymph ecdysteroid titre appeared prior to gut purge afterwhich the titre increased with daily fluctuations (Sakurai et al., 1998). A comparison of the times of peak titres revealed that in each insect, peaks occurred at different times within the last instar. In S. mauritia, two small peaks occurred before the reprogramming peak as is observed in C. ethlius (Dean et al., 1980) and M. sexta (Wolfgang and Riddiford, 1986) and two after the reprogramming In M. brassicae (Agui and Hiruma, 1982) two small peaks occurred peak. after the reprogramming peak and in O. nubilalis (Gelman and Brents, 1984) one small peak occurred before and one after the reprogramming peak. In B. mori, there was a single ecdysteroid peak prior to gut purge after which daily ecdysteroid surges occurred (Sakurai et al., 1998). In S. mauritia, two small peaks Peak I and Peak II appeared at 18 h and 36 h respectively. Peak III which appeared at 60 h is probably associated with the change of epidermal commitment from larval to pupal (reprogramming peak) type. The reprogramming peak, typically occurs at 40-60% of the last instar, assuming that pharate pupal formation was initiated very soon after the premoult ecdysteroid peak (132 h in S. mauritia, 168 h in C. ethlius, 192 h in M. brassicae, 126 h in O. nubilalis). The major ecdysteroid peak, peak IV

which occurred at 114 h may be considered the "moulting peak" of

ecdysteroids. A much smaller peak, peak V appeared at 126 h whose function could not be ascertained. A peak at this stage has not been reported so far.

The two subtle peaks in S. mauritia, peak I and peak II may be associated with the initiation of nucleolar activity, RNA synthesis and organelle formation in the fat body and epidermis. It may also be correlated with fat body DNA synthesis, polyploidy and the initiation of low rate of lipid synthesis as reported during the intermoult peak in C. ethlius (Dean et al., The low haemolymph ecdysteroid concentration has recently been 1980). suggested to possess an important role in the development of Bombyx wing discs (Kawasaki, 1995). The reprogramming peak (Peak III) described earlier is believed to be responsible for the cessation of feeding and initiation of wandering. It also correlates with the initiation of epidermal DNA synthesis and mitosis, and with the progressive determination of pupal characteristics (Change in commitment, reprogramming). This ecdysteroid peak may also be involved in the massive intermoult synthesis in the epidermis and the fat body. Gut purge occurred on day 4 (96 h) in S. mauritia which was approximately 42 h after the appearance of the commitment peak. The premoult peak (Peak IV) was the largest peak and occurred at the pharate pupal stage promoting successful larval-pupal transformation (Riddiford, 1980). The peak appeared soon after gut purge (18 h) and the larvae pupated on day 6, 18 h after the appearance of the prepupal peak. The shoulder at 102 h and the relatively high standard error for 114 h could be explained by the lack of synchrony in pupal ecdysis. The function of peak V which appeared at 126 h could not be ascertained. Ecdysteroids have also been implicated in initiating testis fusion and torsion in *E. kuhniella* (Nowock, 1972) and in influencing the progress of spermatogenesis which will be discussed in a later section.

The daily fluctuation in ecdysteroid titres seen during postembryonic development is presumably brought about by the changing secretory activity of the prothoracic glands (Birkenbeil et al., 1979; Glitho et al., 1979; Sedlak et al., 1983). The cellular changes in prothoracic glands are followed by a major increase haemolymph ecdysteroids in T. molitor (Glitho et al., 1979), M. sexta (Sedlak et al., 1983) and S. littoralis (Zimowska et al., 1985). Fluctuations in the haemolymph ecdysteroid titre in last instar larvae of M. sexta (Bollenbacher et al., 1975; Ciancio et al., 1986), L. dispar (Kelly et al., 1986) and B. mori (Okuda et al., 1985) correlated temporally with fluctuations in the in vitro production of ecdysone by prothoracic glands from the last instar larvae of these species. Earlier studies on the histology and ultrastructure of prothoracic gland cells during sixth instar larval development of S. mauritia suggested a biphasic secretion of ecdysone (Balamani and Nair, 1994, 1997). The first phase of activity occurred in day 3 larvae and the second activity phase on day 5 during the late prepupal stage.

The findings corresponded well with the present determination in the appearance of the commitment and premoult peaks in the ecdysteroid titre determined during the sixth larval stadium. However additional peaks although much smaller, have also been determined which may be caused by low levels of activity of the prothoracic glands.

A possible cause for the sporadic prothoracic gland activity is the daily release of PTTH as reported in the last instar larvae of *B. mori* (Mizoguchi, 1995) and *R. prolixus* (Vafopoulou and Steel, 1996). It has been suggested that PTTH release after JH decline in the haemolymph may be the reason for increase of the basal secretory activity of the prothoracic glands (Rountree and Bollenbacher, 1986; Sakurai *et al.*, 1989) and the glands with higher basal secretory activity produce more ecdysone when stimulated by PTTH. Thus the daily fluctuation seen in the ecdysteroid titres may be due to daily PTTH releases in *S. mauritia*.

Haemolymph ecdysteroid profile during pupal-adult development

Ecdysteroid concentration of pupal haemolymph of S. mauritia showed a rapid increase between day 1 and day 2, peaking on day 2. A single large ecdysteroid peak has also been reported in pupae and pharate adults of *Pieris* (Lafont *et al.*, 1975), *B. mori* (Calvez *et al.*, 1976), *G. mellonella* (Bollenbacher *et al.*, 1978) and C. *ethlius* (Dean *et al.*, 1980). The peak comprises the first third to fifth of the pupal stadium. The position of the peak may be correlated with the time needed for the very extensive remodelling necessary for adult morphogenesis. The amount of ecdysteroids found in pupal and pharate adult stages of S. mauritia were several times (5.5x) greater than the amount detected in the post-feeding larval stage. The higher amount of ecdysteroid synthesis in the pupal and pharate adult stages reflect a greater need of ecdysteroid titres during this stage of development. Ecdysteroid titres of S. mauritia decreased gradually from days 3 - 7. Daily sampling of haemolymph did not detect smaller peaks associated with cuticle deposition in pharate adults and other developmental events. However, the overall pattern is similar to that reported for other insects (Dean *et al.*, 1980, Gelman and Brents, 1984).

It is well known that in insects prothoracic glands are the primary source of ecdysteroids. However, pupal-adult development occurs after the degeneration of prothoracic glands in *S. mauritia* (Balamani and Nair, 1994, 1997). It has been reported that prothoracic gland degeneration occurs due to programmed cell death during normal pupal-adult development of *M. sexta* (Dai and Gilbert, 1997). The ecdysteroid titres of *S. mauritia* decreased gradually during pupal-adult metamorphosis from day 3 to day 7 after the peak on day 2. This may be because prothoracic gland degeneration during metamorphosis is a gradual rather than abrupt process (Dai and Gilbert, 1997). The temporal response of individual gland cells to the programmed cell death signal may be different and the chronology of apoptosis may be tissueor gland-specific even in the same animal (Dai and Gilbert, 1998).

One cannot rule out the possibility that ecdysteroids synthesised during last instar larval development are stored in an inactive and undetectable form by conjugate formation (Delbecque *et al.*, 1988). This might be especially true in an embryo or pupa which is a closed system living on its reserves. At appropriate times during embryonic/postembryonic development, these conjugates may be hydrolysed to yield biologically active ecdysteroids which probably promotes pupal-adult transformation (Hoffmann *et al.*, 1980). It is possible that in *S. mauritia* in the absence of the prothoracic glands, pupaladult metamorphosis is affected by a similar mechanism.

Endocrine significance of critical periods during larval-pupal development

The central role of PTTH in insect growth and development is well documented (for review see Gilbert *et al.*, 1996). PTTH stimulates the synthesis of ecdysteroids or their immediate precursors by the prothoracic glands. Knowledge of the times during development when PTTH is released is crucial to understanding the endocrine regulation of development. 'Neck ligation' has been employed here to tentatively define "critical periods" of PTTH release during larval-pupal development. These critical periods were determined by percent of neck-ligated larvae pupating or showing other prodromal signs of pupation. The validity of the results of the timing of PTTH release was confirmed by comparing with the ecdysteroid titres of the insect determined during sixth instar larval stadium.

The present studies demonstrated that in *S. mauritia*, neck ligation of sixth instar larvae at 30 h, 36 h, 42 h, 48 h, 54 h, 60 h, 66 h, 72 h, 78 h, 84 h and 90 h after ecdysis had no effect on gut purge. Gut purge although slightly delayed, occurred in the usual manner in the ligated larvae. But larvae neckligated at 6 h, 12 h, 18 h and 24 h did not exhibit gut purge behaviour. This may be because the first release of PTTH occurred during the scotophase between 24 h and 36 h after ecdysis to the sixth larval stadium. This PTTH secretion probably activated the prothoracic glands to release the commitment peak of ecdysteroids (Peak III) on day 2 (60 h) described earlier. Gut purge was induced 36 h after the commitment peak in *S. mauritia*.

Post-feeding larvae neck-ligated at 96 h did not exhibit pupal cuticle secretion. On the other hand, larvae neck-ligated at 102 h, 108 h, 114 h, 120 h, 126 h and 132 h after ecdysis showed normal larval-pupal transformation. These findings suggest that the second release of PTTH necessary for the increase in ecdysteroids needed for pupation took place 96 h after ecdysis but prior to 102 h. Concomitantly, ecdysteroid titres rose at 114 h (Peak IV) on day 5. This is identified as the moulting peak of ecdysteroids. Thus two head critical periods for PTTH release have been identified, one for gut purge and the other for pupal moult.

In S. mauritia the time interval between the possible head critical periods for PTTH release and the ecdysteroid titres is 1-2 days in the case of the first release of ecdysteroids whereas it is nearly 12 h in the case of the second peak of ecdysteroids at the prepupal stage. Similar results are observed in Samia cynthia ricini (Fujishita and Ishizaki, 1982). In M. sexta (Truman and Riddiford, 1974) and O. nubilalis (Gelman and Hayes, 1982) an interval of 10 h separate the two critical periods. A wide separation in the magnitude of days for the critical period of ecdysone secretion from that of PTTH release is observed in other lepidopterans.

Direct measurement of haemolymph PTTH concentration in *Bombyx* showed that PTTH secretion occurs three times on different days before gut purge and two in the prepupal period (Shirai *et al.*, 1993). Similarly in *R. prolixus* an *in vitro* assay revealed that PTTH is released into the haemolymph three times (Vafopoulou and Steel, 1996). Moreover, a direct measurement of haemolymph PTTH concentration in *Bombyx* larval haemolymph by time resolved fluoroimmunoassay showed that PTTH is released everyday (Mizoguchi, 1995). Sakurai *et al.* (1998) suggested that HCP in *B. mori* signifies a change in the responsiveness of the prothoracic glands to PTTH so that daily PTTH releases may result in daily ecdysteroid

surges. Hence, HCP was regarded as the turning point of all developmental events after which metamorphic events occurred in a progressive, sequential pattern. Thus, additional releases of PTTH may be causing the additional ecdysteroid peaks in *S. mauritia*. However, in the course of this study it was found that the two HCPs particularly was essential for the activation of prothoracic glands and PTTH if released at other times might be important to synchronise the developmental rhythm by stimulating the prothoracic glands.

Effect of ligation on ecdysteroid titres

Role of brain factors and prothoracic gland secretions in the endocrine regulation of metamorphosis and haemolymph ecdysteroid titres were further analysed by neck/thorax ligation of day 1 and day 4 sixth instar larvae. Although the commitment peak failed to appear in the neck-/thorax-ligated day 1 larvae and the haemolymph ecdysteroid titres of the ligated larvae showed lower values, 70% of the ligated larvae underwent gut purge after a slight delay. In the last instar larvae of lepidopterans, it is the commitment peak of ecdysteroids which induces gut purge (Truman and Riddiford, 1974; Fujishita *et al.*, 1982). In the light of earlier studies on the HCP in *S. mauritia*, the first pulse of PTTH may have already been released in the ligated larvae. Prevention of further stimulation of prothoracic glands by PTTH in the neck-ligated larvae and the absence of prothoracic glands in the thorax-ligated larvae would have caused the decrease in ecdysteroid titres following ligation. The promotion of gut purge in the ligated larvae suggest that the HCP may alter the tissue responsiveness to ecdysteroids (Sakurai *et al.*, 1998) so that even low titres of ecdysteroids can bring about gut purge. Hence it may be assumed that the increased titre of ecdysteroids (so-called commitment peak) found in day 3 larvae serve as pacemakers which synchronise developmental events during metamorphosis. It may also be possible that JH released before the ligation has already disappeared and hence the tissue response to ecdysteroids is not modified by endogenous JH titres. The ligated larvae however do not pupate, therefore these low titres may not be sufficient to induce pupation.

Neck/thorax ligation of day 4 larvae prevented pupation while the control (normal) larvae pupated on day 6. However neck ligation did not prevent the increase in ecdysteroid titres in the haemolymph. Titres of the ligated larvae on day 8 were comparable with the premoult peak on day 5 of normal larvae. The failure of the ligated larvae to undergo pupation even at these high titres suggest that increased titre of ecdysteroids alone will not induce pupation. The absence of JH in the ligated larvae may also prevent pupation. JH is known to enhance the metamorphic response of tissues to ecdysteroids and/or alter the sensitivity of these tissues to ecdysteroids (Denlinger, 1979). Moreover, determination of JH titres during last instar larval development of several lepidopteran species revealed the occurrence of a

premoult peak of JH in the prepupal stage (Jones et al., 1990; Edwards et al., 1995). There is also suggestive evidence to show that prepupal increase of JH has a morphogenetic role in promoting development of normal pupal morphology (Safranek et al., 1980; Jones and Hammock, 1985; Balamani and Nair, 1989). Earlier studies evaluating the HCP of S. mauritia have suggested that the second HCP occurs 96 h after ecdysis (day 4) to the sixth instar. Thus PTTH may already be released in the ligated larvae. It may be recalled that haemolymph ecdysteroid titre in the day 4 larvae is significantly high. Hence, the increase in titres after neck ligation may be due to a positive feedback effect of the initially secreted ecdysteroids (Sakurai and Williams, 1989) on the prothoracic glands after initial activation by PTTH. Ecdysteroid titres in the thorax-ligated larvae however decreased considerably following ligation. This may be due to the absence of the prothoracic glands in these ligated larvae which is known to be the major source of haemolymph ecdysteroids (Chino et al., 1974; King et al., 1974).

Role of JH in the endocrine control of metamorphosis

The interplay of ecdysteroids and juvenile hormones in larval insects serve to orchestrate the progression from one developmental stage to the next, with ecdysteroids regulating the onset and timing of the moulting cycle and JH regulating the quality of the moult (Riddiford 1985, 1994; Sehnal, 1985). During the last larval stadium of holometabolous insects such as Lepidoptera, a reduction of the haemolymph JH levels is a necessary step in the initiation of metamorphosis (Riddiford, 1980; Sehnal, 1985). However, increase in JH levels associated with the wandering stage of last instar larvae has been shown to be critical for the success of larval-pupal ecdysis (Shooley *et al.*, 1976; Hsiao and Hsiao, 1977; Bean *et al.* 1982; Jones and Hammock, 1985; Balamani and Nair, 1991; Edwards *et al.*, 1995). There is very little published information on JH levels in lepidopterous pupae. The accepted view is that JH is largely absent in the pupa in order that development of adult structures may proceed. However, increased levels of JH has been reported in pharate adults and adults (Edwards *et al.*, 1995).

Inhibitory effects of JH on ecdysteroid titres

Repetitive treatments of high dosage of JHA to early sixth instar larvae of S. mauritia interfere with larval development and metamorphosis. The treated larvae moulted into imperfect supernumerary larvae within 24 h of termination of hormone treatment. These were larval in appearance except for a few pupal characteristics especially the mouth parts, antennae and thoracic legs which were sclerotised. The paired wing discs were also mildly sclerotised. These findings confirm and supplement earlier observations in S. mauritia (Santha and Nair, 1987; Balamani and Nair, 1991; Venugopalan, 1995). Similar results were observed with treatments of JHA to last instar Galleria larvae (Slama and Mala, 1984) and S. litura (Hatakoshi et al., 1986).

The commitment peak of ecdysteroids did not appear in the JHA treated S. mauritia larvae. The large premoult peak preceding larval-pupal moult was also missing. A much smaller peak appeared one day earlier (day 4) to the premoult peak (day 5) of the control larvae and the treated larvae moulted into supernumerary larvae on the next day. The induction of the formation of supernumerary larvae may be caused due to suppression of normal haemolymph ecdysteroid titres by the JHA. JHA also caused ecdysteroid titres to appear 12 - 24 h earlier than normal. It seems that prothoracic glands of treated larvae become active earlier than those of controls. JH apparently accelerates the release of PTTH, which in turn stimulates the prothoracic glands (Sehnal et al., 1981). It has also been reported that juvenile hormone or its analogue, administered externally exerts its morphogenetic effects along with other environmental stimuli by either inhibiting or accelerating moulting processes through interference with the prothoracic glands and brain (Slama, 1971; Gilbert et al., 1980; Sehnal, 1983). JH converts the metamorphic cycle into a larval one by modifying the growth and metabolism of prothoracic glands (Slama and Mala, 1984). In holometabolous insects, the commitment for a larval-pupal moult is induced by 20-hydroxyecdysone in the absence of JH (Riddiford, 1996). Therefore, persistence of JH or JHA during that time may have resulted in the formation of supernumerary larvae.

Repetitive treatments of *S. mauritia* with JHA also seems to affect the growth, differentiation and developmental competence of imaginal structures resulting in the formation of imperfect superlarvae. This may be because of alterations in the haemolymph titre of ecdysteroids leading to unco-ordinated development of different tissues since different tissues may have different hormonal requirements for further development (Riddiford, 1981).

In neck-ligated sixth instar larvae of *S. mauritia*, ecdysteroid titres that were reduced following neck ligation were further reduced after JHA treatment. However, titres in the thorax-ligated larvae were similar for treated and control larvae. From previous studies, head critical period for gut purge is tentatively proposed to occur during the period from 24 h to 36 h after ecdysis in *S. mauritia*. It follows that PTTH is already released in the ligated larvae. Hence application of JHA is not expected to have any effect on gut purge. However, JHA treated, ligated larvae did not exhibit gut purge whereas 70% of the control larvae exhibited the premetamorphic behaviour of gut purge.

The ecdysteroid data support the concept that JH can directly inhibit the secretory activity of prothoracic glands (Cymborowski and Stolarz, 1979; Safranek *et al.*, 1980). In *M. sexta* JH affects the phophorylation of the 34 KDa proteins which regulate a rate-limiting step in the ecdysone biosynthetic pathway (Rountree *et al.*, 1987). Hydroprene suppressed the

steroidogenic competence of prothoracic glands in neck-ligated larvae during the phagoperiod suggesting that the hormone acted directly on prothoracic glands in *M. sexta* (Watson and Bollenbacher, 1988). In *B. mori* JH inhibits PTTH and also ecdysone release from prothoracic glands (Sakurai, 1984). Further studies in *B. mori* have shown that JH acts during the early stages of the instar to suppress both the secretory activity of prothoracic glands and also the acquisition of competence to respond to PTTH (Sakurai et al., 1989). From neck ligation experiments it has been established that JH is able to suppress ecdysteroid production. This strengthens the observation that JH inhibits ecdysteroid production independent of the brain (Safranek et al., 1980; Sehnal et al., 1981; Sakurai, 1983). However, ecdysteroid titres were not significantly different in the neck-ligated JHA treated larvae compared to the control larvae and were similar in the thorax-ligated, JHA treated and control larvae. Hence prevention of gut purge by decreased ecdysteroid titres alone in the JHA treated larvae does not seem plausible. Neck/thorax ligation of day 1 larvae have shown that even low levels of ecdysteroids could cause gut purge. After the HCP, sensitivity of tissues to ecdysteroids increase (Sakurai et al., 1998) hence this change. But JHA may have an inhibitory effect at the receptor level preventing gut purge although ecdysteroid titres are comparable in both the JHA treated and control larvae. In this case, JHA may also be acting at the level of other target tissues responsible for gut purge.

It is generally accepted that JH is present at relatively high levels, throughout all but the final larval stadium. The presence of JH in larval insects serves to maintain the larval state, and prevent metamorphosis. JH shows high titre early in the last larval instar which then declines to undetectable levels by the end of the feeding period (Bollenbacher *et al.*, 1981; Baker *et al.*, 1987; Jones *et al.*, 1990; Edwards *et al.*, 1995). In the last larval instar, the absence of JH by the end of the phagoperiod enables the process of metamorphosis to ensue. The subsequent release of ecdysteroids cause gut purge and wandering behaviour.

Stimulatory effects of JH on ecdysteroid titres

Increased levels of JH has been associated with the prepupal stage of several lepidopteran species (Baker *et al.*, 1987; Jones *et al.*, 1990; Edwards *et al.*, 1995). JH titres reportedly increase approximately 24 h prior to pupal ecdysis (Edwards *et al.*, 1995). It has also been reported that after gut purge, JH or analogues actually seem to accelerate prepupal development by elevating ecdysteroid titres (Cymborowski and Stolarz, 1979; Hiruma, 1980; Safranek *et al.*, 1980).

During the course of this study, it was found that neck-ligated day 4 sixth instar larvae treated with JHA underwent larval-pupal ecdysis while thorax-ligated larvae treated with JHA and control larvae (neck- and thoraxligated) remained unchanged. The JHA treated neck-ligated day 4 larvae transformed into headless pupae. Earlier studies have shown that PTTH may already be released at this time. Ecdysteroid titres of the headless pupae (day 0) were higher than control larvae on day 7 but slightly lower than that of the neck-ligated larvae on day 8. Ecdysteroid titres of the JHA treated, thorax-ligated and control larvae were low. This suggested that the presence of prothoracic glands and hence a critical titre of ecdysteroids is essential for JH to bring about pupation. Induction of pupation in the JHA treated neckligated larvae is consistent with the observations of Masner et al. (1975), Mitsui and Riddiford (1978) and other authors that JH can lower the hormonal requirements of epidermis for cuticle secretion. This may be the reason for the induction of pupation in the JHA treated, neck-ligated larvae and the failure of pupation in the control larvae at similar titres. Increased levels of JH has also been reported immediately prior to the moult in antipenultimate and penultimate instar larvae and also in pharate adults (Edwards et al., 1995) of the tomato moth Lacanobia oleracea. Hence it may be assumed that these pre-ecdysial increases may be intimately associated

with the promotion of morphogenesis in addition to the stimulation of ecdysteroid production by the prothoracic glands. During larval-pupal transformation in most Lepidoptera, a brief burst of JH occurs in the prepupal phase of development (Bean *et al.*, 1982; Baker *et al.*, 1987; Jones *et al.*, 1990; Edwards *et al.*, 1995). Several studies have demonstrated that this increase of JH along with the second release of PTTH activates the prothoracic glands to their maximal rate of ecdysone secretion needed for moulting and metamorphosis (Cymborowski and Stolarz, 1979; Hiruma, 1980; Sieber and Benz, 1980; Balamani and Nair, 1991). Further, experiments have indicated that JH activation of prothoracic glands may be mediated by a prothoracic gland stimulatory factor which is induced in the fat body by JH (Gruetzmacher et al., 1984b). There is also suggestive evidence to show that prepupal increase of JH has a morphogenetic role in promoting the development of normal pupal morphology (Safranek et al., 1980; Jones and Hammock, 1985; Balamani and Nair, 1991). Treatments of last instar larvae of S. mauritia during prepupal phase with anti-JH agents induced the production of larval-pupal intermediates (Santha and Nair, 1986; Nair and Rajalekshmi, 1989). It has also been suggested that the presence of a critical titre of JH in the prepupal phase is needed for ecdysteroids to exert their effects on imaginal structures (Balamani and Nair, 1991). JH may also enhance the response of ecdysteroids and/or alter the sensitivity of tissues to ecdysteroids (Denlinger, 1979).

In Lepidoptera, prothoracic glands degenerate during or after eclosion to the adult form (Gilbert, 1962) by undergoing programmed cell death or apoptosis (Dai and Gilbert, 1997). However, treatment of JHA to newly ecdysed pupae of *S. mauritia* increased the levels of ecdysteroid titres when compared to controls. The increase in titres appeared to be a dose dependent phenomenon as titres increased with higher dosage of JHA. It has been reported that in paurometabolous insects, the degeneration of prothoracic glands is prevented by application of JHA suggesting that either the presence of JHA acts to maintain prothoracic glands (Smith and Nijhout, 1983) or the absence of JH programmes the glands for degeneration (Lanzrein, 1975). JH may also have a tropic effect on the prothoracic glands in addition to preventing its degeneration (Sakurai and Gilbert, 1990). The premoult peak of ecdysteroids appeared on day 3 in the 0.5 μ g JHA treated pupae while the peak appeared on day 2 in the 1 µg JHA treated and control pupae and titres decreased gradually thereafter. Increased levels of ecdysteroids on most days in the JHA treated pupae may be due to increased rates of synthesis elicited by JHA treatment in the prothoracic glands. In M. sexta, administration of JH prevented apoptosis of the prothoracic glands even 11 days after JH injection into young pupae. The prothoracic glands remained intact and their ability to synthesise ecdysteroids was maintained at a fairly active level as ascertained by RIA after in vitro incubation (Dai and Gilbert, 1998). Thus in addition to its maintenance function, JH acts as a tropic factor by stimulating ecdysteroidogenesis in the prothoracic glands of various insects (Sakurai and The present experiment confirms this tropic function in Gilbert. 1990). Spodoptera. JH may stimulate PTTH secretion from the brain retrocerebral complex as well, since the ecdysteroid biosynthetic capacity of JH-injected M. sexta pupae having an intact brain was significantly higher than that of

JH-injected brainless pupae (Dai and Gilbert, 1998). While JH plays a role in maintaining the glands, prothoracic gland degeneration itself is stimulated by ecdysone in the absence of JH (Smith and Nijhout, 1983).

The presence of ecdysteroid receptor (ER) complex in the prothoracic glands is well accepted (Bidmon and Sliter, 1990; Talbot et al., 1993) and that the sensitivity of the prothoracic glands to ecdysteroids is dependent upon the ER concentration. It was hence suggested that JH could maintain the glands by interfering with the increase in ER concentration elicited by circulating Thus it may be possible that in the absence of JH, the ecdysteroids. prothoracic glands initiate and modulate adult development by synthesising and secreting large quantities of ecdysteroids and in that way also elicit their self destruction, i.e. it is a suicidal event for the glands (Dai and Gilbert, 1997). The composite data suggest that JH can both maintain and stimulate the prothoracic glands. Therefore during normal pupal-adult metamorphosis, the absence of JH is a prerequisite for both the initiation and completion of prothoracic gland degeneration. Also, the generally accepted view is that JH is largely absent in the pupa in order that development of adult structures may proceed.

JHA treatment was found to interfere with transformation of pupae into adults. Slama *et al.* (1974) reported that application of different doses of the juvenoids during the critical period evokes the formation of intermediate forms between the previous and subsequent developmental stages. The failure of emergence of JHA treated insects may be due to the inhibition of synthesis or release of eclosion hormone (Truman, 1971). Since eclosion hormone is released by a drop in the ecdysteroid level after a peak level, prior to ecdysis (Riddiford, 1985). In this respect, high ecdysteroid titres caused by the prothoracicotropic action of JHA, may cause an inhibition of eclosion hormone release in *S. mauritia*. The development of pupal-adult intermediates may be due to differential sensitivity of different tissues to ecdysteroids.

Effects of ecdysteroid agonist RH 5992 on metamorphosis of sixth instar larval stadium

In the course of this study, it was found that RH 5992 acts primarily by inducing a premature, lethal moult. This mode of action has been confirmed by several authors by applying RH 5847, a similar ecdysteroid agonist to larvae of several Lepidoptera: Spodoptera eridania (Aller and Ramsay, 1988) *M. sexta* (Wing et al., 1988), *B. mori* (Kiuchi, 1990) *Plodia interpunctella* (Silhacek et al., 1990), *S. littoralis* (Smagghe and Degheele, 1992), *S. frugiperda* (Monthean and Potter, 1992), *Plutella xylostella* and *P. brassicae* (Darvas et al., 1992), *S. exempta* and *S. exigua* (Smagghe and Degheele, 1993) and *S. exempta, S. littoralis, S. exigua, M. brassicae, G. mellonella* (Smagghe and Degheele, 1994) and *S. mauritia* (Sakunthala and Nair, 1995). In some 595.78 MOND coleopteran species, RH 5849 caused a variety of effects on moulting and behaviour (Aller and Ramsay, 1988; Darvas *et al.*, 1992; Smagghe and Degheele, 1993, 1994). RH 5992 also induced precocious, lethal moulting in Lepidoptera: *S. exempta, M. brassicae, S. littoralis, S. exigua* and *G. mellonella* (Smagghe and Degheele, 1994b) and *Choristoneura fumiferana* (Retnakaran *et al.*, 1997). However, RH 5992 had no effect on larval instars of Coleoptera, Heteroptera and Orthoptera (Smagghe and Degheele, 1994b).

Treatment of RH 5992 to newly moulted sixth instar S. mauritia larvae induced an incomplete larval moult resulting in the formation of a new untanned head capsule in 48 h. Since the treatment was before the commitment peak of ecdysteroids, the newly secreted cuticle was larval. Acceleration of gut purge behaviour also occurred in these larvae. Gut purge has been demonstrated to be an ecdysteroid dependent phenomenon (Truman and Riddiford, 1974; Fujishita et al., 1982). Hence the treated RH 5992 may have stimulated this behaviour. When post-feeding (day 4) larvae were treated with RH 5992, the newly secreted cuticle was pupal. This may be because the treatment was done after the commitment peak of ecdysteroids when the epidermal cells are already committed to the pupal programme. Ultrastructural studies on treated larvae also showed that ecdysteroid agonists caused a forced, untimely synthesis of a new, however incomplete cuticle (Smagghe et al., 1994). Treatment of RH 5992 to post-feeding larvae

101

NB 2952

resulted in the formation of larval-pupal intermediates. The larval-pupal intermediates have an unsclerotised pupal cuticle in the abdominal region and thoracic tergum beneath the old cuticle whereas the thoracic legs were larval in appearance. Since thoracic legs develop from imaginal discs, they may have different hormonal requirements from the epidermal cells. Injection of ecdysterone to post-feeding ligated larvae of *S. mauritia* also induced the formation of larval-pupal intermediates (Balamani and Nair, 1991).

It was observed that feeding and weight gain ceased simultaneously in treated larvae after the induction of premature moulting. This was also the case in normal *S. mauritia* larvae where cessation of feeding and fresh weight gain occurred prior to pupation. Similar effects on growth and development have been reported in other insects, either during normal development, or when treated with natural ecdysteroids (Kiguchi and Agui, 1981; Sehnal *et al.*, 1981; Kubo *et al.*, 1983). For non-steroidal ecdysteroid agonists, the observed effects agree with those of Wing *et al.* (1988) who reported that RH 5849 caused precocious moulting and cessation of feeding in *M. sexta* larvae. Likewise, Silhacek *et al.* (1990), Darvas *et al.* (1992) and Smagghe and Degheele (1994a,b) reported suppression of weight gain in various insect species treated with RH 5849 and RH 5992.

Positive feedback effect of RH 5992

Treatment of RH 5992 induced gut purge and precocious lethal moulting in both neck-ligated and thorax-ligated day 1 larvae. Induction of gut purge and moulting in thorax-ligated larvae which lacked a biosynthetic source of ecdysteroids strengthens the assumption that RH 5992 and other ecdysteroid agonists behave as a bonafide ecdysone in Lepidoptera. Moreover, Smagghe and Degheele (1992) have reported that ecdysteroid agonists (RH 5849) stimulate epidermal cells to undergo apolysis and synthesise a new cuticle via binding of RH 5849 to ER of epidermal cells. They also demonstrated by using electrophoretic methods that malformation of the new larval cuticle was due to the disappearance of several cuticular and haemolymph proteins in the treated larvae. Also, RH 5849 and RH 5992 showed receptor binding activity, ecdysteroid responsive gene expression activity and induction of morphological changes in Drosophila KC cells corresponding to their in vivo potencies (Mikitani, 1996). Acceleration of gut purge behaviour may also be due to a similar mechanism.

Ecdysteroid titres of treated neck-ligated larvae decreased marginally on day 1 and day 2 but an abrupt increase was observed on day 3 when compared to control titres which showed a decreasing trend. The reduction in titres on day 1 and day 2 in RH 5992 treated and control larvae may be due to the absence of cephalic endocrine structures secreting PTTH which is known to be the major neurohormone regulating ecdysteroid production in insects (Sakurai and Gilbert, 1990). The abrupt increase on day 3 may be due to positive feedback effect of RH 5992 on the prothoracic glands. The prothoracic glands of the ligated larvae are relatively inactive and it has been postulated that the nature of the feedback (positive or negative) is likely to depend on the secretory history of the glands (Williams, 1952; Siew and Gilbert, 1971; Sakurai and Williams, 1989). Since prothoracic glands have also been known to possess ecdysteroid receptors (Bidmon and Sliter, 1990; Talbot et al., 1993), it is possible that receptor induction allows the prothoracic glands to respond to ecdysteroids leading to a rise in ecdysteroid production by prothoracic glands and hence the haemolymph ecdysteroid titres. A major contributing factor to the efficacy of RH 5992 is its superior absorption and metabolic stability in Lepidoptera or their tissues (Retnakaran et al., 1995). This effect did not manifest itself on day 1 and day 2 because larvae exposed to RH 5992 after ligation still contains a high JH titre and JH can directly inhibit the secretory activity of prothoracic glands (Cymborowski and Stolarz 1979; Safranek et al., 1980). Synthesis of a larval cuticle following RH 5992 treatment during early sixth larval stadium also indicates that JH titre is high during this time.

Thorax-ligated larvae treated with RH 5992 did not show a significant difference in ecdysteroid titres when compared to control larval titres. The ligated larvae lacked cephalic and thoracic endocrine structures and hence the reduction in titres in both the treated and control larvae.

Negative feedback effect of RH 5992

Treatment of RH 5992 induced abnormal and lethal pupation both in neck-ligated and thorax-ligated day 4 larvae 48 h after treatment. Larvalpupal intermediates with unsclerotised pupal cuticle in the abdominal region and thoracic tergum, larval thoracic legs and fused testis were formed. A similar phenomenon has been reported by Friedlander and Brown (1995) causing reinitiation of pupal development in isolated abdomens of diapausing *Cydia pomonella* larvae treated with RH 5992. Synthesis of pupal cuticle and testicular fusion, which are pupal characteristics were also observed in the treated isolated abdomens. Treatment of RH 5849 and RH 5992 inducing larval-pupal intermediates has also been reported in other Lepidoptera: *S. exempta, S. exigua, S. littoralis, M. brassicae* and *G. mellonella* (Smagghe and Deheele, 1994a, b).

Pupal cuticle synthesis in the treated, thorax-ligated larvae which lacked prothoracic glands further strengthened the ER-activation concept. The newly synthesised cuticle was pupal because the epidermal cells were already committed to pupal phase as has been described previously. Testicular fusion was also observed in all the treated larvae while the control larvae failed to exhibit any of the above mentioned changes in morphogenesis and died as prepupae.

Treatment of RH 5992 to neck-ligated larvae led to a decrease in haemolymph ecdysteroid titres on all days following treatment both in the sixth instar larvae and the larval-pupal intermediates. Conversely, titres showed an increasing trend in the control larvae. The decline in titres may be due to a negative feedback effect of RH 5992 on prothoracic glands in the neck-ligated larvae. The prothoracic glands of S. mauritia of day 3 has been reported to be moderately active (Balamani and Nair, 1994, 1997). Haemolymph ecdysteroid titre determination of S. mauritia reported earlier has also suggested active secretory activity of prothoracic glands during day 3 - day 4. That the nature of the feedback effect depends on the secretory history of the prothoracic glands has already been discussed. Hence, it follows that the reduction in ecdysteroid titres may be due to a negative feedback effect. Thus a change in sensitivity of the prothoracic glands cells to RH 5992 is observed, the feedback effect being positive in the day 1 ligated larvae and negative in day 4 ligated larvae.

Treatment of RH 5992 to thorax-ligated day 4 larvae led to a decreasing trend of titres. Control larvae also exhibited a similar trend of declining ecdysteroid titres. This may be due to the absence of prothoracic glands in both treated and control larvae. Moreover, other ecdysteroid secreting tissues in the abdomen may be refractory to the action of RH 5992 at this stage due to negative feedback effect.

In S. mauritia treatment of RH 5992 to newly ecdysed pupae caused high mortality in the treated pupae towards the end of the pupal period. The pupal period of the treated pupae was also found to be prolonged. Ecdysteroid titres were found to be significantly reduced in both the dosages treated when compared to the control pupae. They also did not exhibit the moulting peak on day 2. On day 5, titres were lower than that on day 2 in the treated pupae but significantly higher than those of control pupae. RH 5992 may have a negative feedback effect on the prothoracic glands (as observed in late sixth instar larvae) preventing the premoult ecdysteroid peak on day 2 and in the process preventing apoptosis.

Beydon and Lafont (1983) have reported a similar phenomenon in $P.\ brassicae$ pupae where 20-hydroxyecdysone acts on ecdysteroid production as a negative-feedback regulator. 20-hydroxyecdysone also suppressed ecdysone production by the prothoracic glands *in vitro* (Scott, 1982). Prothoracic glands possess ecdysteroid receptors (Bidmon and Sliter, 1990; Talbot *et al.*, 1993) and the nature of the feedback effect depends on the secretory history of the glands (Williams, 1952; Siew and Gilbert, 1971; Sakurai and Williams, 1989). Ecdysteroid titres of the normal pupae of *S. mauritia* show that prothoracic glands of day 0 pupae are moderately

active, hence the negative feedback effect. RH 5992 however induces pupaladult metamorphosis *via* ER activation. Hence pupal-adult metamorphosis could be promoted by the treated RH 5992 which serves as the "moulting peak" for the development of imaginal structures in the absence of endogenous premoult ecdysteroid peak. The high levels of RH 5992 also prevents ecdysis by inhibiting the release of eclosion hormone since eclosion hormone is released by a drop in the ecdysteroid level after a peak titre, prior to ecdysis (Riddiford, 1985).
3.5 SUMMARY

- 1. The haemolymph ecdysteroid profile of S. mauritia was determined during the sixth (final) larval instar and pupal stage by means of RIA. Effects of hormones and hormone mimics on haemolymph ecdysteroid titres were studied by means of ligation and treatments of a JHA (hydroprene) and a non-steroidal ecdysteroid agonist (RH 5992). Control insects were treated with acetone.
- 2. Haemolymph ecdysteroid profile of sixth instar larvae revealed two peaks, the commitment peak on day 3 and the moulting peak on day 5 when daily sampling was done. Six hourly sampling of sixth instar larval haemolymph revealed five peaks. In addition to the commitment and moulting peaks of ecdysteroids, two small peaks appeared before the commitment peak and one small peak after the moulting peak. The commitment peak induces gut purge and other prodromes of pupation while the moulting peak promotes pupation. Daily sampling of pupal haemolymph revealed a single large peak on day 2 which promotes pupal-adult metamorphosis.
- 3. The timings of PTTH release during larval-pupal development were studied by observing the percentage of neck-ligated larvae pupating or showing other prodromal signs of pupation. Sixth instar larvae were neck-ligated at 6 h intervals from the time of ecdysis and the effect on two ecdysteroid-dependent developmental processes i.e., gut purge and pupal cuticle secretion were observed. This study suggested that the first release of PTTH occurs during the scotophase between 24 h and 36 h after ecdysis. The second release of PTTH probably occurs after 96 h of ecdysis but prior to 102 h. The first release of PTTH probably stimulates the secretion of the commitment peak and the second release of PTTH, the moulting peak of ecdysteroids.
- 4. Influence of brain factors and prothoracic gland secretions on haemolymph ecdysteroid titres were studied by neck/thorax ligation of day 1 and day 4 sixth instar larvae. Haemolymph ecdysteroid titres were reduced significantly following ligation on day 1 but 70% of the ligated larvae purged their gut after a slight delay even at these reduced ecdysteroid titres. Following neck ligation on day 4, titres increased gradually to reach a peak on day 8 which was not significantly different from the ecdysteroid titre peak on day 5 (prepupae) of the control larvae. But thorax ligation led to a significant decrease in titres of the haemolymph. However, both neck- and thoraxligated larvae failed to undergo pupation.

- 5. The influence of JH on haemolymph ecdysteroid titres was investigated by topical application of JHA to unligated day 0 larvae and neck-/thorax-ligated day 1 and day 4 sixth instar larvae and intact pupae of S. mauritia. The effects of these treatments on larval-pupal and pupal-adult transformations were observed.
- 6. Repetitive treatments of 20 µg JHA to intact sixth instar larvae induced an imperfect supernumerary larval moult on day 5. The commitment peak failed to appear and a much smaller premoult peak appeared on day 4 in the treated larvae. The commitment peak appeared on day 3 and a larger premoult peak appeared on day 5 in the control larvae. JHA treatment probably promoted PTTH release 24 h earlier than in controls and also suppressed prothoracic gland secretion of ecdysteroids.
- 7. Repetitive treatments of 5 μ g JHA to neck-/thorax-ligated day 1 larvae prevented gut purge in the treated larvae while 70% of the control larvae purged their gut. Haemolymph ecdysteroid titres of JHA treated, neck-ligated larvae were slightly lower than control while titres were equal in the thorax-ligated and control larvae. Ecdysteroid titres of JHA treated and control larvae were not significantly different, yet the prevention of gut purge in the treated larvae suggested that JH in addition to suppressing the prothoracic glands also alters tissue sensitivity to ecdysteroids.
- 8. Single treatment of JHA to neck-/thorax-ligated day 4 larvae promoted pupation in the neck-ligated larvae but failed to induce pupation in the thorax-ligated larvae. Control larvae failed to undergo pupation in both the neck- and thorax-ligated larvae. Ecdysteroid titres of the headless pupae were higher than those of the control larvae on the same day. The presence of JH may be essential for normal pupal morphogenesis. JH may also stimulate higher rates of ecdysteroid synthesis by the prothoracic glands at this stage. Thorax-ligated, JHA treated larvae exhibited ecdysteroid titres lower than that of control larvae. The failure of pupation in the JHA treated (thorax-ligated) larvae may be due to the absence of the prothoracic glands.
- 9. JHA treatment to pupae induced formation of adultoids which failed to emerge. Haemolymph ecdysteroid titres increased considerably following JHA treatment. However, the pattern of fluctuation was similar in the treated and control pupae. JHA prevents apoptosis (programmed cell death) of the pupal prothoracic glands and also has a tropic effect on the glands, stimulating ecdysteroidogenesis. The increased titres maintained during the latter half of pupal development

probably prevented release of eclosion hormone resulting in the failure of emergence of adults in the treated pupae.

- 10. The effects of ecdysone agonist RH 5992 on larval development and metamorphosis and haemolymph ecdysteroid titres was investigated by topical application of RH 5992 on unligated larvae and neck-/thoraxligated day 1 and day 4 sixth instar larvae and intact pupae of *S. mauritia.*
- 11. Treatment of RH 5992 to intact sixth instar day 0 larvae induced a premature, lethal larval moult and acceleration of gut purge in the treated larvae. The newly synthesised cuticle was larval since the treatment was before the commitment peak of ecdysteroids. Treatment of RH 5992 to intact day 4 larvae induced the formation of larval-pupal intermediates. The control larvae underwent normal pupation on day 6.
- 12. Treatment of RH 5992 to neck-/thorax-ligated sixth instar day 1 larvae with different dosages of RH 5992 caused a precocious lethal larval moult in the ligated larvae. All the treated larvae exhibited gut purge precociously while only 70% of the control larvae purged their gut that too after some delay. Ecdysteroid titres in the RH 5992 treated neckligated larvae decreased on days 1 and 2 but increased abruptly on day 3 when compared to control titres which showed a decreasing trend on all days. Titres of the RH 5992 treated and control thorax-ligated larvae were similar. Induction of a moult at low titres indicated that RH 5992 promotes a moult at the receptor level without an increase in haemolymph ecdysteroid titres. Increase in titres on day 3 may be due to a positive feedback effect of RH 5992 on prothoracic glands. In the absence of prothoracic glands in the thorax-ligated larvae, such a feedback effect may not have occurred.
- 13. Treatment of different dosages of RH 5992 to neck-/thorax-ligated sixth instar day 4 larvae induced an abnormal and lethal pupation. The larvae transformed into larval-pupal intermediates. Control larvae failed to exhibit any change and died as larvae. Ecdysteroid titres of the treated, neck-ligated larvae decreased on all days while control titres increased to reach a peak on day 8. The reduction in titres in the treated larvae may be due to negative feedback effect of RH 5992 on prothoracic glands. Ecdysteroid titres of the thorax-ligated larvae also decreased following treatment as in the case of control larvae. Induction of pupation in the treated larvae even at low titres of ecdysteroids further demonstrates that RH 5992 acts at the receptor level.

RH 5992 treatment to pupae induced formation of adultoids. The pupal 14. period got extended following treatment and mortality occurred towards the end of the pupal period. The moulting peak failed to appear in the treated pupae while control pupae exhibited a single large premoult peak on day 2. On day 5, titres of the treated pupae decreased slightly but was higher than that of the control pupae. Reduction in titres on day 2 in the treated pupae may be due to negative feedback effect of RH 5992 which inturn may prevent apoptosis. Hence, the basal secretory activity of the prothoracic glands was maintained even on day 5 in the treated pupae. This may cause the higher titre in the treated pupae on day 5. The relatively high ecdysteroid titres and retention of RH 5992 in the tissues during the latter half of pupal stage may have prevented release of eclosion hormone and hence the mortality towards the end of pupal period.

CHAPTER 4

1 (**)**

DEVELOPMENTAL PROFILE OF TESTICULAR ECDYSTEROIDS AND THE EFFECTS OF HORMONE MIMICS

4.1 INTRODUCTION

Ecdysteroids have been reported in the testes of many lepidopteran species (Loeb et al., 1982, 1984, 1988; Shimizu et al., 1985; Gelman et al., 1989; Jarvis et al., 1994). Earlier studies revealed that ecdysteroid secretion in testes occurred at discrete periods during the last larval instar and pupal stadium of various Lepidoptera. Adult testes have also been reported to contain ecdysteroids. Synthesis of ecdysteroids is suggested to be induced by a brain peptide, testis ecdysiotropin (Loeb et al., 1986, 1987). Testes from larvae (final instar) and pupae released large amounts of ecdysteroids in vitro (Loeb et al., 1982, 1984; Shimizu et al., 1985). Testicular metabolism of ecdysteroids has been investigated by incubating the testes with [3H] ecdysone which was found to be converted mostly to 20-hydroxyecdysone and traces of other ecdysones (Gelman et al., 1985; Loeb et al., 1988; Loeb and Woods, 1989). Comparison of ecdysteroids in the testes and haemolymph of H. virescens (Loeb et al., 1982), O. nubilalis (Gelman et al., 1988a) and S. littoralis (Jarvis et al., 1994) revealed that the pattern of fluctuation of ecdysteroids is similar for total levels but differs in the timing of their fluctuations and in their content.

A vast amount of literature has accumulated implicating hormones in the regulation of spermatogenesis (reviewed by Dumser, 1980; Koeppe *et al.*, 1985). JH has been implicated both as an inhibitor (Fukuda, 1944; Nowock, 1973; Leviatan and Friedlander, 1979; Deb and Chakravorty, 1981) and stimulator (Nishitsutsuji-Uwo, 1961; Yagi, 1975; Shinyaeva, 1981) of testicular development. The stimulatory effects of ecdysteroids on spermatogenesis has also been well documented in several insect species both in *in vitro* and *in vivo* experiments (Nishitsutsuji-Uwo, 1961; Yagi *et al.*, 1969; Dumser and Davey, 1975; Dumser, 1980; Gelman and Borkovec, 1986).

The present work investigates the fluctuations in testicular ecdysteroids of S. mauritia during the immature (larva, pupa) and adult stages of development. Role of head factors and prothoracic gland secretions in influencing testicular ecdysteroid content has been analysed by ligation techniques as described earlier. Effects of treatments of a JHA (hydroprene) and an ecdysteroid agonist (RH 5992) on testicular ecdysteroid titres have also been investigated. The endogenous testicular ecdysteroid levels during the sixth (final) instar larval and pupal stages are compared with those in the haemolymph. This information, it was thought would facilitate correlation of timing of testicular ecdysteroid peaks with developmental stage in S. mauritia and thus uncover some clues on the physiological importance of testicular ecdysteroid content. Testicular ecdysteroid titres have also been correlated with known stages of spermatogenesis in S. mauritia (Venugopalan et al., 1994) to ascertain the role of testicular ecdysteroids in regulating testicular development and spermatogenesis.

4.2 MATERIALS AND METHODS

Animals

Larvae, pupae and adults of *S. mauritia* were taken from the laboratory stock culture reared and maintained as described previously. The present studies were conducted on sixth (final) instar larvae, pupae and adults of various ages.

Ligations

Ligations were done as described earlier (see page 45).

Chemicals

Chemicals were obtained from various sources as mentioned earlier (see page 46).

Treatments

Larvae and pupae were treated with different dosages of chemicals as described previously (see page 46).

Dissection of testes lobes

Sixth instar larvae, pupae and adults of *S. mauritia* were anaesthetised in specimen tubes containing a wad of cotton soaked in diethyl ether at the bottom. The anaesthetised insects were taken out after 5 minutes and pinned dorsal side up in a wax lined petridish. The dissections were carried out in insect Ringer solution (Ephrussi and Beadle, 1936). After dorsal incision, the cuticle was cut open along the mid-dorsal line upto the hind edge of the head capsule, and pinned back exposing the internal anatomy. The testis was taken out using fine forceps, cleaned of extraneous tissue and rinsed in insect Ringer solution to remove adherent haemolymph.

Preparation of tissue samples for RIA

Tissue samples were prepared by the procedure described earlier (see page 49) and RIA done as described earlier.

4.3 RESULTS

4.3.1 Testicular ecdysteroid titres during larval-pupal-adult development

The larvae of *S. mauritia* pass through six instars before they pupate. Sixth instar larvae purge their gut and exhibit wandering behaviour on the fourth day of the sixth (final) instar. Pupation occurs on day 6 and the adult males emerge after eight days.

Testes of S. mauritia are a pair of kidney shaped organs attached to the body wall on either side of the mid-dorsal line in the fifth abdominal segment. The testicular volume increases gradually during the fifth (penultimate) instar and reaches a maximum on day 3 sixth instar larvae, when it measures 4.3 ± 0.9 mm³. During the pharate pupal stage, the paired testes fuse in the mid-dorsal line to form a single median, spherical structure. The fused testis shows a remarkable increase in volume from early pupal stage and reaches its maximum size in day 2 pupae when it measures 10.5 ± 0.86 mm³. The testis volume decreases from day 2 onwards and is least in the adult stage. The compartmentalisation of the testicular lumen into four follicles is also lost in the late pupal stages.

The present study was undertaken to determine testicular ecdysteroid titres of larvae, pupae and adults of *S. mauritia*. Testicular ecdysteroids were determined by means of RIA and compared with known stages of spermatogenesis. Results are expressed as pg 20-hydroxyecdysone equivalents/testes pair.

(i) Testicular ecdysteroid titres during sixth instar larval stadium

Testicular ecdysteroid titres for sixth instar larvae of S. mauritia are \therefore shown in Fig. 23. In the newly moulted (white head capsule) sixth instar larvae (day 0), the titres were 145 ± 31 pg which increased to 151 ± 29 pg on day 1. On day 2 hormone titres got reduced to 120 ± 7 pg. Ecdysteroid titres started to rise steadily on day 3 (225 ± 59 pg). The increasing trend continued on day 4 (460 ± 37 pg). The paired testes fused on day 5, the prepupal stage. An increase in ecdysteroid titres were also observed on day 5 (594 ± 153 pg). The testicular ecdysteroid titres increased greatly towards the late sixth

117P



Nature of	n	Days of survival	% of sur	Nature of testes		
Ilgation		(Mean <u>+</u> SD)	No change	Gut purge	Pupation	lobes
Neck- ligated	25	9 <u>+</u> 1	30	70		Separate
Thorax- ligated	25	9 <u>+</u> 1	30	70		Separate
Unligated (Control)	15 each	6 <u>+</u> 0		100	100	Fused

Table12. Effects of ligations of sixth instar day 1 larvae onmetamorphosis and testicular fusion

instar larval stage and formed a significant peak (P < 0.05) on day 5, following testis fusion.

(ii) Testicular ecdysteroid titres during pupal stadium

Ecdysteroid titres of the testes during the post-pupal stage are represented in Fig. 24. In the newly moulted pupae (day 0), the hormone titres were low (851 ± 226 pg). Ecdysteroid titres increased considerably to reach its highest value on day 1 (2642 ± 192 pg). Titres declined steadily on day 2 (2106 ± 335 pg), day 3 (1913 ± 446 pg), day 4 (1452 ± 534 pg), day 5 (773 ± 139 pg), day 6 (320 ± 68 pg) and day 7 (269 ± 43 pg). The testicular ecdysteroid titres increased significantly (P < 0.05, relative to titres on day 0 and day 6) in the early pupal stage and then declined gradually, reaching a minimum on day 7.

(iii) Testicular ecdysteroid titres during adult stadium

Testicular ecdysteroid titres of day 0 adults were 17 ± 12 pg. Hormone titres reduced considerably in the adult.

4.3.2 Effects of ligations of sixth instar larvae on testicular ecdysteroid titres

This study was undertaken to investigate the role of brain factors and prothoracic gland secretions on testicular ecdysteroid titres of *S. mauritia*. In one set of experiments, sixth instar day 1 or day 4 larvae were 'neck-ligated' to

.



Nature of	n	Days of survival	% of survivi which sh	Nature of	
IIgation		(Mean <u>+</u> SD)	No change	Pupation	testes lobes
Neck- ligated	25	4 <u>+</u> 1	100		Separate
Thorax- ligated	25	4 <u>+</u> 1	100		Separate
Unligated (Control)	15 each	2 <u>+</u> 0		100	Fused

Table 13. Effects of ligations of sixth instar day 4 larvae onmetamorphosis and testicular fusion

.

remove the endogenous source of cerebral neurosecretory factors and JH. In another set of experiments, sixth instar day 1 or day 4 larvae were 'thoraxligated' to eliminate the endogenous sources of both head factors (cerebral neurosecretory factors and JH) and prothoracic gland secretions.

(i) Effects of ligations of sixth instar day 1 larvae on testicular ecdysteroid titres

The present study was undertaken to determine the effect of neck/thorax ligation of day 1 larvae on testicular ecdysteroid titres. Seventy percent of the ligated larvae exhibited gut purge, however none of them underwent pupation although the larvae survived for 9 ± 1 days. On the other hand, all the control larvae exhibited gut purge and became pupae on day 6. The testes of the ligated larvae did not undergo fusion while the testes of the control larvae underwent fusion on day 5 (Table 12).

Ecdysteroid titres of experimental and control larvae are shown in Fig. 25. Hormone titres of the neck-ligated larvae were 130 ± 47 pg, 148 ± 20 pg, 53 ± 17 pg and 45 ± 4 pg for day 1, day 2, day 3 and day 4 respectively. Those of thorax-ligated larvae were 96 ± 26 pg, 33 ± 19 pg, 32 ± 16 pg and 0 pg for day 1, day 2, day 3 and day 4 respectively. Hormone titres of larvae kept as control were 151 ± 29 pg, 120 ± 7 pg, 225 ± 59 pg and 460 ± 37 pg for day 1, day 2, day 3 and day 4 respectively. Ligation of sixth instar day 1 larvae caused a significant reduction (P < 0.05) in testicular

1198



Fig. 25 Effects of neck/thorax ligation of day 1 larvae on testes ecdsteroid titres

ecdysteroid titres, on day 3 and day 4 in the neck-ligated larvae and on day 2, day 3 and day 4 in the thorax-ligated larvae. In the control larvae, hormone titres exhibited an increasing trend on all days.

(ii) Effects of ligations of sixth instar day 4 larvae on testicular ecdysteroid titres

The present study was undertaken to determine the effects of neck/thorax ligation of day 4 larvae on testes ecdysteroid titres. The ligated larvae survived for 4 ± 1 days and died without pupating. The testes of the ligated larvae remained separate and did not undergo fusion. On the other hand, the control larvae pupated on day 6 and their testes underwent fusion on day 5 (Table 13).

Ecdysteroid titres of experimental and control larvae are shown in Fig. 26. Hormone titres of the neck-ligated larvae were 434 ± 62 pg, 468 ± 55 pg, 511 ± 114 pg, 562 ± 169 pg and 712 ± 84 pg on day 4, day 5, day 6, day 7 and day 8 respectively. Those of thorax-ligated larvae were 399 ± 43 pg, 346 ± 86 pg, 185 ± 20 pg, 161 ± 20 pg and 209 ± 19 pg on day 4, day 5, day 6, day 7 and day 8 respectively. Hormone titres of larvae kept as control showed testicular ecdysteroid content of 460 ± 37 pg and 594 ± 153 pg on day 4 and day 5 respectively. Ecdysteroid titres in the neck-ligated larvae increased gradually to form a peak on day 8. Testicular ecdysteroid peak on day 8 was not significantly different from the peak titre on day 5 in the control larvae.





Fig. 26 Effects of neck/thorax ligation of day 4 larvae on testes ecdysteroid titres

Table 14.	Effects of repeated daily treatments of JHA to sixth ins	star
larvae fr	com day 0 onwards on metamorphosis and testicular fus	ion

		97	Number of transforme			
Dosage	n	% mortality	Imperfect supernumerar y larvae	Pupae	testes lobes	
20 µg	25	20	20		Separate	
5 µl Acetone (Control)	15			15	Fused	

On the other hand thorax ligation led to a significant reduction (P < 0.05) in testicular ecdysteroid titres on day 8 when compared to control larval titres on day 5. Control larval titres on the other hand, increased on day 4 and day 5.

4.3.3 Effects of treatments of JHA on testicular ecdysteroid titres

Previous studies on *S. mauritia* suggested that JHA has an inhibitory or stimulatory effect on metamorphosis and haemolymph ecdysteroid titres depending on the stage of development of the insect. The present investigation was undertaken in order to find out whether JHA has a similar effect on the testicular ecdysteroid titres of *S. mauritia* during larval-pupal and pupal-adult development.

(i) Effects of repetitive treatments of unligated sixth instar larvae with JHA

Sixth instar larvae were topically treated with repeated daily dose of 20 μ g JHA from day 0 to day 3. Control larvae received an equivalent volume of acetone. The effects of these treatments on testicular ecdysteroid titres were carefully studied. The treated larvae moulted into imperfect supernumerary larvae within 24 h of termination of hormone treatment, i.e. on day 4. The supernumerary larvae survived for 6-7 days. However, the control larvae underwent pupation on day 6. The testes remained separate in the JHA treated larvae whereas testes became united on day 5 in the control

Nature of ligation	Dosage	n	% of sur	Nature of		
			No change	Gut purge	Pupation	testes lobes
Neck- ligated	5 µg	20	100			Separate
Neck- ligated (Control)	5 μl Acetone	10	30	70		Separate
Thorax- ligated	5 µg	20	100			Separate
Thorax- ligated (Control)	5 μl Acetone	10	30	70		Separate

Table 15. Effects of repeated daily treatment of JHA to sixth instar day1 ligated larvae on metamorphosis and testicular fusion

_

larvae. The testes volumes also were much lower in the treated larvae when compared to that of the control larvae (Table 14).

Ecdysteroid titres of experimental and control larvae are represented in Fig. 27. Hormone titres of JHA treated larvae were 84 ± 31 pg, 150 ± 26 pg, 155 ± 33 pg, 269 ± 77 pg and 207 ± 66 pg for day 1, day 2, day 3, day 4 sixth instar larval stage and day 0 supernumerary larval stage respectively. Testicular ecdysteroid titres of larvae kept as control were 151 ± 29 pg, 120 ± 7 pg, 225 ± 59 pg, 460 ± 37 pg and 594 ± 153 pg on day 1, day 2, day 3, day 4 and day 5 respectively. Ecdysteroid titres increased gradually to reach its peak on day 4 and decreased in the supernumerary larvae (day 0). Titres in the control larvae increased steadily to reach its peak on day 5. The peak titre of JHA treated larvae was significantly lower (P < 0.1) than that of the control larvae.

(ii) Effects of repetitive treatments of ligated sixth instar day 1 larvae with JHA

Sixth instar larvae were neck- or thorax-ligated on day 1 and topically treated with a repetitive daily dose of 5 μ g JHA for three days. Ligated larvae kept as control were treated with an equivalent volume of acetone. The effects of these treatments on testicular ecdysteroid titres were carefully studied. The JHA treated larvae did not show the premetamorphic behaviour of gut purge while 70% of the control larvae exhibited gut purge after 4.5 ± 0.5 days.

800 -700 · Control 600 · JHA treated pg Ecdysteroid / Testes pair 500 400 300 -200 100 0 з Day 2 0 Supernumerary 1 4 larva

1220

Fig. 27 Effects of repetitive treatments of larvae with 20 μg JHA from day 0 onwards on testes ecdysteroid titres

Nature of	Dosage	n	% of surviv which s	Nature of testes		
			No change	Pupation	lobes	
Neck- ligated	5 µg	20	20	80	Fused	
Neck- ligated (Control)	5 μl Acetone	10	100		Separate	
Thorax- ligated	5 µg	20	100		Separate	
Thorax- 5 µl ligated Acetone (Control)		15	100		Separate	

Table 16. Effects of treatments of JHA to sixth instar day 4 ligatedlarvae on metamorphosis and testicular fusion

None of the JHA treated or control larvae pupated. Testes remained separate in the JHA treated and control larvae (Table 15). Testes volumes decreased in the JHA treated larvae when compared to the control larvae.

Ecdysteroid titres of the experimental and control larvae are represented in Fig. 28. The ecdysteroid titres of the JHA treated, neck-ligated and thorax-ligated larvae were 17 ± 2 pg and 0 pg respectively on day 4. In the control larvae, the hormone titres of neck-ligated and thorax-ligated larvae were 45 ± 4 pg and 0 pg respectively for the same day. Titres of the JHA treated, neck-ligated larvae were significantly lower (P < 0.05) than those of the control larvae whereas titres of both JHA treated, thorax-ligated larvae and control larvae were reduced to zero.

(iii) Effects of treatments of ligated sixth instar day 4 larvae with JHA

Sixth instar day 4 larvae were neck- or thorax-ligated and topically treated with a single dose of 5 μ g JHA. Ligated larvae kept as control were treated with an equivalent volume of acetone. The effects of these treatments on testicular ecdysteroid titres were carefully studied.

Eighty percent of the neck-ligated day 4 larvae treated with JHA transformed into headless pupae on day 7 while neck-ligated day 4 larvae kept as control showed a complete inhibition of pupation. The ligated larvae survived for 4 ± 1 days and died without pupating. Testes of the headless

NBP



day 1 larvae on testes ecdysteroid titres

pupae were fused while those of the control larvae remained separate. Larvae thorax-ligated on day 4 failed to pupate even after treatment of JHA. These larvae survived for 4 ± 1 days. Control larvae also showed a complete inhibition of pupation. The testes of both the JHA treated and control larvae remained separate (Table 16).

(a) Changes in ecdysteroid titres of neck-ligated larvae treated with JHA

Ecdysteroid titres of experimental and control larvae are represented in Fig. 29. Testicular ecdysteroid titres of day 0 headless pupae were 1031 ± 160 pg while those of control (neck-ligated on day 4) larvae were 562 ± 169 pg. Titres of headless pupae were significantly higher (P < 0.1) than that of control larvae (day 7).

(b) Changes in ecdysteroid titres of thorax-ligated larvae treated with JHA

Ecdysteroid titres of experimental and control larvae are represented in Fig. 30. Testicular ecdysteroid titres of treated larvae were 106 ± 8 pg, 411 ± 131 pg and 181 ± 4 pg on day 5, day 7 and day 8 respectively. Titres of control larvae were 346 ± 86 pg, 161 ± 20 pg and 209 ± 19 pg on day 5, day 7 and day 8 respectively. Hormone titres decreased both in the JHA treated and



larvae on testes ecdysteroid titres





Fig. 30 Effects of treatment of 5 µg JHA to thorax-ligated day 4 larvae on testes ecdysteroid titres

36

control larvae. There was no significant difference between ecdysteroid titres of the treated and control larvae on day 8.

(iv) Effects of treatments of pupae with JHA on testicular ecdysteroid titres

Newly ecdysed day 0 pupae were treated with a single dose of $0.5 \mu g$ or 1 μg JHA. Control pupae received an equivalent volume of acetone. The effects of these treatments on testicular ecdysteroid titres were carefully studied. High mortality was observed in the treated pupae. Surviving pupae showed various abnormalities in pupal-adult development and none of them succesfully emerged as morphologically normal adults. When the pupal case of such unemerged pupae were removed they were found to be pupal-adult intermediates or adultoids. Conversely, moths which emerged from the control pupae were normal and healthy.

Ecdysteroid titres of experimental and control pupae are shown in Fig. 31. Hormone titres of pupae treated with 0.5 µg JHA were 1112 ± 257 pg, 2547 ± 656 pg, 2937 ± 451 pg, 1963 ± 137 pg, 1396 ± 305 pg and 780 ± 194 pg on day 1, day 2, day 3, day 4, day 5 and day 6 respectively. Titres of pupae treated with 1 µg JHA were 5852 ± 1424 pg, 4124 ± 1066 pg, 4370 ± 933 pg, 1917 ± 556 pg, 2607 ± 149 pg and 1285 ± 197 pg on day 1, day 2, day 3, day 4, day 5 and day 6 respectively. Ecdysteroid titres of pupae kept as controls were 2642 ± 192 pg, 2106 ± 335 pg, 1913 ± 446 pg, 1452 ± 534 pg, 773 ± 139 pg


Nature of	Deserve		% of sur	Nature of		
ligation	Dosage	n	No change	Gut purge	Pupation	testes lobes
Neck-ligated	2 μg	20		100		Separate
Neck-ligated	5 μ g	20		100		Separate
Neck-ligated (Control)	5 μl Acetone	15 each	30	70		Separate
Thorax-ligated	$2~\mu extbf{g}$	20		100		Separate
Thorax-ligated	5 µg	20		100		Separate
Thorax-ligated (Control)	5 μl Acetone	15 each	30	70		Separate

Table 17. Effects of treatments of RH 5992 to sixth instar day 1 ligatedlarvae on metamorphosis and testicular fusion

and 320 ± 68 pg day on 1, day 2, day 3, day 4, day 5, and day 6 respectively. Ecdysteroid titres were significantly higher (P < 0.1) in the 0.5 µg JHA treated pupae on day 5 and day 6 and almost on all days in the 1 µg JHA treated pupae when compared to control titres. Testicular ecdysteroid titres increased greatly in the early pupal stage and then declined both in the JHA treated and control pupae.

4.3.4 Effects of treatments of ecdysteroid agonist RH 5992 on testicular ecdysteroid titres

Previous studies on *S. mauritia* suggested that RH 5992 induces a precocious moult in the treated larvae. However, depending on the stage of development, RH 5992 had a positive or negative feedback effect on the haemolymph ecdysteroid titres of the treated animals. The present investigation was undertaken in order to find out whether RH 5992 has a similar effect on the testicular ecdysteroid titres of *S. mauritia* during larvalpupal and pupal-adult development.

(i) Effects of treatments of ligated sixth instar day 1 larvae with RH 5992

Sixth instar day 1 larvae were neck- or thorax-ligated and treated topically with a single dose of 2 μ g or 5 μ g RH 5992. Ligated larvae kept as controls were treated with an equivalent volume of acetone. The effects of these treatments on testicular ecdysteroids were carefully studied. Ligated larvae treated with RH 5992 survived for 3 ± 1 days while larvae kept as control survived for 9 ± 1 days. RH 5992 treatment induced gut purge in all the treated larvae whilel only 70% of the control larvae underwent gut purge. The testis remained separate both in the treated and control larvae (Table 17). The testes volume increased slightly following RH 5992 treatment when compared to control larvae.

(a) Changes in ecdysteroid titres of neck-ligated larvae treated with RH 5992

Ecdysteroid titres of neck-ligated larvae treated with RH 5992 and those of control larvae are represented in Fig. 32. Titres of neck-ligated larvae treated with 2 µg RH 5992 were 74 ± 11 pg, 60 ± 32 pg and 58 ± 27 pg on day 1, day 2 and day 3 respectively. Those of 5 µg RH 5992 treated larvae were 56 ± 5 pg, 83 ± 23 pg and 174 ± 40 pg on day 1, day 2 and day 3 respectively. Ecdysteroid titres of larvae kept as controls on day 1, day 2 and day 3 were 130 ± 47 pg, 148 ± 20 pg and 53 ± 17 pg respectively. Testicular ecdysteroid litres were significantly lower (P < 0.1) in the RH 5992 treated larvae on day 1 and day 2 when compared to control larvae but an abrupt but significant increase (P < 0.05) in titres occurred in the 5 µg RH 5992 treated larvae on day 3. nial P



Fig. 32 Effects of treatments of RH 5992 to neck-ligated day 1 larvae on testes ecdysteroid titres
Nature of ligation	Dosage	n	% of surviving larvae which transformed into:			Nature
			No change	Pupae	Larval-pupal intermediate s	of testes lobes
Neck-ligated	2 µg	20	20		80	Fused
Neck-ligated	5 µg	20	20		80	Fused
Neck ligated	10 µg	20	20		80	Fused
Neck-ligated (Control)	5 μl Acetone	15 each	100			Separate
Thorax-ligated	$2~\mu m g$	20	20		80	Fused
Thorax-ligated	δμg	20	20		80	Fused
Thorax-ligated (Control)	5 μl Acetone	15 each	100			Separate

Table 18. Effects of treatments of RH 5992 to sixth instar day 4 ligatedlarvae on metamorphosis and testicular fusion

(b) Changes in ecdysteroid titres of thorax-ligated larvae treated with RH 5992

Ecdysteroid titres of thorax-ligated larvae treated with RH 5992 and those of control larvae are represented in Fig. 33. Titres of thorax-ligated larvae treated with 2 µg RH 5992 were 61 ± 15 pg, 19 ± 8 pg and 23 ± 23 pg on day 1, day 2 and day 3 respectively. Those of 5 µg RH 5992 treated larvae were 75 ± 19 pg, 55 ± 20 pg and 28 ± 9 pg on day 1, day 2 and day 3 respectively. Ecdysteroid titres of larvae kept as control (on day 1, day 2 and day 3) were 96 ± 26 pg, 33 ± 19 pg and 32 ± 16 pg respectively. Hormone titres of both 2 µg and 5 µg RH 5992 treated larvae were lower than those of control larvae for most days. However, the difference in titres of RH 5992 treated and control larvae were not significant.

(ii) Effects of treatments of ligated sixth instar day 4 larvae with RH 5992

Sixth instar day 4 larvae were neck-/thorax-ligated and treated with a single dose of 2 μ g or 5 μ g or 10 μ g RH 5992. Ligated larvae kept as control were treated with an equivalent quantity of acetone. The effects of these treatments on testicular ecdysteroid titres were carefully studied. Ligated larvae treated with different doses of RH 5992 transformed into larval-pupal intermediates on day 6. They survived for 3 ± 1 days. Neck-/thorax-ligated day 4 larvae kept as control showed a complete inhibition of pupation, they

VEOP



day 1 larvae on testes ecdysteroid titres

30

survived for 4 ± 1 days and then died. Testes of RH 5992 treated larvae underwent fusion while those of the control larvae remained separate (Table 18).

(a) Changes in ecdysteroid titres of neck-ligated larvae treated with RH 5992

Ecdysteroid titres of neck-ligated larvae treated with RH 5992 and those of control larvae are represented in Fig. 34. Titres of neck-ligated larvae treated with 2 μ g RH 5992 were 350 \pm 43 pg and 942 \pm 252 pg on day 4 and day 5 of sixth instar larval stage and 378 + 43 pg and 437 + 83 pg on day 0 and day 1 of larval-pupal intermediate stage respectively. Those of 5 µg RH 5992 treated larvae were 355 ± 45 pg and 288 ± 50 pg on day 4 and day 5 of sixth instar larval stage and 325 ± 14 pg and 179 ± 34 pg on day 0 and day 1 of larval-pupal intermediate stage respectively. Titres of 10 µg RH 5992 treated larvae were 442 ± 39 pg and 453 ± 62 pg on day 4 and day 5 of sixth instar larval stage and 198 ± 57 pg and 154 ± 12 pg on day 0 and day 1 of larval-pupal intermediate stage respectively. Ecdysteroid titres of the neckligated control larvae were 434 ± 62 pg, 468 ± 55 pg, 511 ± 114 pg and 562 + 169 pg on day 4, day 5, day 6, and day 7 respectively. Testes ecdysteroid titres declined (not all are significant at P < 0.1) in the RH 5992 treated larvae on all days but for a few exceptions. Control larvae exhibited an increasing trend in the titre of ecdysteroids reaching its maximum on day 7.

129A



Fig. 34 Effects of treatments of RH 5992 to neck-ligated day 4 larvae on testes ecdysteroid titres

40

(b) Changes in ecdysteroid titres of thorax-ligated larvae treated with RH 5992

Ecdysteroid titres of thorax-ligated larvae treated with RH 5992 and those of control larvae are represented in Fig. 35. Titres of thorax-ligated larvae treated with 2 µg RH 5992 were 554 ± 86 pg and 171 ± 47 pg on day 4 and day 5 of sixth instar larval stage and 124 ± 41 pg on day 0 of larval-pupal intermediate stage respectively. Those of 5 µg RH 5992 treated larvae were 430 ± 54 pg, 188 ± 23 pg on day 4 and day 5 of sixth instar larval stage and 140 ± 16 pg on day 0 of larval-pupal intermediate stage respectively. Ecdysteroid titres of control larvae were 399 ± 43 pg, 346 ± 86 pg and 185 ± 20 pg on day 4, day 5 and day 6 respectively. Testicular ecdysteroid titres decreased in the RH 5992 treated and control larvae, however titres were lower but not significantly in the RH 5992 treated larvae.

(iii) Effects of treatments of pupae with RH 5992

Newly ecdysed day 0 pupae were treated with a single dose of 5 μ g or 10 μ g RH 5992. Control pupae received an equivalent volume of acetone. The effects of these treatments on testicular ecdysteroid titres were carefully studied.

Most of the treated pupae survived for 9 ± 1 days but failed to emerge. Removal of the pupal cases of such unemerged pupae revealed the presence of

BOP



Fig. 35 Effects of treatments of RH 5992 to thorax-ligated day 4 larvae on testes ecdysteroid titres

adultoids. High mortality was observed towards the end of the pupal period. Most of the pupae died on or before the day of adult emergence. The pupal period of the RH 5992 treated pupae was found to be slightly prolonged $(9 \pm 1 \text{ days})$. Conversely, moths which emerged from the control pupae were normal and healthy.

Ecdysteroid titres of experimental and control pupae on day 2 and day 5 are represented in Fig. 36. Hormone titres of the pupae treated with 5 µg RH 5992 were 2090 \pm 335 pg and 780 \pm 70 pg on day 2 and day 5 respectively. Titres of 10 µg RH 5992 treated pupae were 1373 \pm 604 pg and 774 \pm 42 pg on day 2 and day 5 respectively. Ecdysteroid titres of the control pupae were 2106 \pm 335 pg and 773 \pm 139 pg on day 2 and day 5 respectively. Ecdysteroid titres of treated and control pupae did not show significant difference on day 2 and day 5.

4.3.5 Comparison of testicular and haemolymph ecdysteroid titres during larval-pupal and pupal-adult development

(i) Testicular ecdysteroid profile Vs Haemolymph ecdysteroid profile during sixth instar larval stadium

Testicular and haemolymph ecdysteroid titres for sixth instar larvae of S. mauritia are shown in Fig. 37. In the newly moulted (white head capsule) larvae (day 0) the testicular ecdysteroid titres were 145 ± 31 pg and increased



Fig. 36 Effects of treatments of pupae with 5 Jug and 10 Jug RH 5992 on Testes ecdysteroid titres



19:00

GP – Gut purge ; W – Wandering ; Pp – Prepupa

to 151 ± 29 pg on day 1. On day 2, hormone titres decreased $(120 \pm 7 \text{ pg})$ and increased on day 3 $(225 \pm 59 \text{ pg})$. The increasing trend continued on day 4 $(460 \pm 37 \text{ pg})$. Fusion of the paired testes occurred on day 5, the prepupal stage. Testicular ecdysteroid titres increased greatly on day 5 to form the highest titre $(594 \pm 153 \text{ pg})$ during the sixth larval stadium. In the newly moulted (white head capsule) sixth instar larvae (day 0), titres were 37 ± 21 pg. Hormone titres decreased on day 1 $(15 \pm 2 \text{ pg})$ and increased slightly on day 2 $(26 \pm 5 \text{ pg})$ and day 3 $(39 \pm 20 \text{ pg})$. Haemolymph titres increased rapidly on day 4 $(372 \pm 13 \text{ pg})$ and formed the highest peak on day 5 $(2524 \pm 453 \text{ pg})$. Ecdysteroid titres of the testes and haemolymph exhibited a minor increase on day 3 and a significant peak on day 5.

(ii) Testicular ecdysteroid profile Vs Haemolymph ecdysteroid profile during pupal stadium

Testicular and haemolymph ecdysteroid titres of pupae stadium of S. mauritia are shown in Fig. 38. In the newly moulted pupae (day 0), hormone titres were low (851 ± 226 pg). Titres increased rapidly to reach its maximum on day 1 (2642 ± 192 pg). Testes ecdysteroids then declined steadily on day 2 (2106 ± 335 pg), day 3 (1913 ± 446 pg), day 4 (1452 ± 534 pg), day 5 (773 ± 139 pg), day 6 (320 ± 68 pg) and day 7 (269 ± 43 pg). In the newly moulted pupae, titres were low (686 ± 63 pg), which increased gradually to 1973 + 521 pg on day 1. Hormone titres increased abruptly at this stage to



*

Fig. 38 Changes in testes and haemolymph ecdysteroid titres during the pupal instar development

reach its maximum $(14221 \pm 1458 \text{ pg})$. Titres then declined on day 3 $(7831 \pm 478 \text{ pg})$, day 4 $(5915 \pm 1159 \text{ pg})$, day 5 $(894 \pm 74 \text{ pg})$, day 6 $(574 \pm 186 \text{ pg})$ and day 7 $(467 \pm 185 \text{ pg})$. Testicular and haemolymph ecdysteroid titres increased significantly during the early pupal stage. Testicular titres formed a peak on day 1 while haemolymph titres formed a peak on day 2. Titres declined gradually thereafter in both the haemolymph and testes to reach their minimum on day 7.

4.4 DISCUSSION

In S. mauritia, the timetable of testicular development and spermatogenesis (Venugopalan et al., 1994) is synchronised with the somatic development of the animal as has been reported for other insects. In insects, spermatogenesis can be sub-divided into three consecutive phases. During the first phase, the germ cells undergo mitotic proliferation and form clusters containing a species specific, defenitive number of spermatogonia. The second phase begins with the transformation of each spermatogonium into a primary spermatocyte which in turn produces four spermatids through two successive of spermatogenesis meiotic divisions. During the last phase (= spermiogenesis), the spermatids differentiate into mature spermatozoa (Dumser, 1980). Histological studies of spermatogenesis have revealed that spermatogenesis of S. mauritia conform to the general pattern reported for other lepidopterans (Venugopalan et al., 1994). Testes of S. mauritia grow in size gradually throughout the sixth (final) larval instar and reach its maximum volume in day 2 pupae and decrease gradually thereafter. Testes of the fifth (penultimate) instar larvae is mainly composed of spermatogonial cells which undergo repeated mitotic divisions towards the later stages of the instar. In the early sixth instar larvae, the spermatogonia aggregate into cysts (spermatocysts) in which mitotic divisions occur resulting in the formation of spermatocytes. These spermatocytes undergo meiotic division to form spermatids which appear on day 2 of the sixth instar larvae. The differentiation of eupyrene and apyrene sperm bundles is initiated in the prepupal phase (day 4 and day 5) of the sixth instar. Also, the paired testes undergo fusion along the mid-dorsal line to form a single median structure in the pharate pupal (day 5) phase. The fused testis undergoes torsion in the young pupa. Fully differentiated eupyrene and apyrene sperm bundles also appear for the first time during the early pupal period. Maximum number of eupyrene and apyrene sperm bundles is found in day 6 pupae. The spermatogonial and spermatocyte cysts disappear during the late pupal period and are no longer visible in the testis of the adults. In the adult testis, sperm bundles both eupyrene and apyrene are found distributed throughout the lumen of the testis. The number of sperm bundles decrease towards adult eclosion (Venugopalan et al., 1994).

Ecdysteroid synthesis by the testis

Ecdysteroids have been reported in adults of D. melanogaster (Hodgetts et al., 1977) and C. vicina where they are localised primarily in the testes (Koolman et al., 1979). Subsequent investigations on testicular ecdysteroids in various lepidopteran species have largely been confined to the immature stages, although the testis of adult H. virescens have also been shown to contain ecdysteroids (Loeb et al., 1984). In the testes of last instar larvae of O. nubilalis, 20-hydroxyecdysone and various high polarity ecdysteroids were detected, while in pupae and pharate adults, ecdysone was additionally **a,b** detected (Gelman *et al.*, 1988). Ecdysteroids have also been reported in the testes of L. separata (Shimizu et al., 1989) and B. mori (Fugo et al., 1995). Testes of H. virescens and L. dispar secrete ecdysteroids in vitro during midand late-last instar larval stage and mid to late developmental period during the pupal stage (Loeb et al., 1982, 1984, 1988). Testes of O. nubilalis was also found to synthesise ecdysteroids in vitro especially during late last larval instar (Gelman et al., 1989). However, exogenous edcysteroids is needed to initiate (Loeb et al., 1988) or boost (Loeb et al., 1986) endogenous ecdysteroid production in the testes in vitro. Testes of M. brassicae and S. littoralis have also been found to spontaneously secrete ecdysteroids in vitro. When pupal testes of diapausing M. brassicae were incubated in vitro, 2-deoxyecdysone, ecdysone and 20-hydroxyecdysone were found in the culture medium when

analysed by HPLC-RIA (Shimizu *et al.*, 1985). HPLC-RIA analysis of larval testes ecdysteroids of *S. littoralis* also yielded similar results (Jarvis *et al.*, 1994). Immunocytochemical techniques and dissections have implicated the testis sheath as the site of synthesis of ecdysteroids in the testis (Loeb, 1986). The results of the present study also shows that ecdysteroids are clearly present in the testes of *S. mauritia*. Testicular ecdysteroid titres also exhibit significant fluctuations during larval-pupal-adult development.

Testicular ecdysteroid profile during larval-pupal development

The testes of S. mauritia grow in size throughout the last larval instar and attain maximum volume in day 3 larvae. Histological studies on spermatogenesis of S. mauritia have shown that spermatogonial cells undergo multiplication and proliferation from day 0 - day 3 of sixth larval instar. Further, the first phase of spermiogenesis, the formation of spermatids, occur during day 2 and day 3 of sixth larval instar (Venugopalan *et al.*, 1994). The testicular ecdysteroid titres of the sixth instar larvae of S. mauritia exhibited fluctuations concomitant with the developmental phase of the testicular cyst cells. Titres which were at a base level on day 0 and day 1 decreased on day 2 and formed a peak on day 3. This titre peak may be promoting meiotic division and hence the formation of spermatids. The proliferation of gonial cells may be occurring at low levels of ecdysteroids observed on day 0 and day 1. The reduction in titres on day 2 may be a cue for the gonial cells to undergo meiotic division for the formation of spermatids. The increase in titres on day 3 may further promote spermatid formation. Benny and Nair (unpublished data) reported that the DNA and RNA content of the testes of S. mauritia increased from day 0 to day 3 and the protein content also increased gradually during the sixth larval instar.

In S. mauritia, mid-last instar production of ecdysteroids in the testes was observed one day after the change from mitotic to meiotic division in the testicular gonial cells. Testicular ecdysteroid titres of H. virescens also increased during mid-last instar larval period which coincided with the change from mitotic to meiotic division (Loeb and Birnbaum, 1981), increase in testes size and eupyrene cyst maturation (Loeb et al., 1984). The Heliothis testes contained and released a spectrum of at least 12 different ecdysteroids (Loeb et al., 1982). Testicular ecdysteroids have also been reported to stimulate meiosis and elongation of eupyrene spermatocysts in vivo in O. nubilalis (Gelman and Borkovec, 1986). Ecdysteroids have been reported to stimulate mitotic and/or meiotic division during early stages of spermatogenesis (Yagi et al., 1969; Kambysellis and Williams, 1971b, 1972; Takeda, 1972; Dumser and Davey, 1975). The present determination suggests that synthetic periods of the testes may affect or be affected by internal changes of spermatocyst development. Testes ecdysteroids may regulate maturation of male genital

tissues by induction of growth factors secreted by the testis itself and the fat bodies (Loeb and Hakim, 1991; Loeb, 1991b, 1994).

The differentiation of eupyrene and apyrene sperm bundles is initiated in the prepupal phase (day 4 and day 5) of the sixth larval instar of S. mauritia (Venugopalan et al., 1994). The fusion of the paired testes occurs in the pharate pupal stage (day 5). The present study reveals that testicular ecdysteroid titres increased on day 4 and reaches a peak on day 5. This increase in titres on day 4 and day 5 may be stimulating the differentiation of eupyrene and apyrene sperm bundles. Testicular fusion may also be under the influence of the testes ecdysteroid peak on day 5. Testicular ecdysteroids may also be influencing RNA content of the testes of S. mauritia which declined on day 4 but increased dramatically on day 5. Testicular sheath peptides were also reported to increase on day 5 (Benny and Nair, unpublished results). Ecdysteroid content of the Heliothis testes also exhibited a peak in late-last instar larvae, one day after the start of spermatocyst elongation (Loeb et al., 1984). The role of ecdysteroids in stimulating the maturation and differentiation of male germ cells has been suspected since the early work of Schmidt and Williams (1953) in S. cynthia. Also, the role of ecdysteroids in promoting spermatogenesis has been well documented in vivo (Takeuchi, 1969; Dumser and Davey, 1975) and in vitro (Yagi et al., 1969; Kambysellis and Williams, 1971b, 1972; Takeda, 1972;

Fukushima and Yagi, 1975) observations. Ecdysteroids have also been reported to stimulate apyrene meiosis and elongation in *O. nubilalis* (Gelman *et al.*, 1988). That ecdysteroids are involved in testicular fusion has also been reported by a few authors (Nowock, 1972, 1973; Loeb *et al.*, 1984; Benny and Nair, 1999). A peak in testicular ecdysteroids at the time of fusion of the paired testes has also been reported in *O. nubilalis* (Gelman *et al.*, 1988) and *A*. *B. mori* (Fugo *et al.*, 1995). Deficiency of ecdysteroids in the testes caused by heat treatment of larvae induced male sterility during the wandering and pharate pupal stage of *B. mori* (Fugo *et al.*, 1995).

Testicular ecdysteroid profile during pupal-adult development

In S. mauritia, fully formed eupyrene and apyrene sperm bundles appeared for the first time during the early pupal period. A major increase in sperm bundles and torsion of the fused testis also occurred in the pupa (Venugopalan *et al.*, 1994). These changes in testicular development and spermatogenesis coincided with the single large peak of testicular ecdysteroids observed on day 1 of pupal development in S. mauritia. Ecdysteroid titres decreased gradually during the rest of the pupal stadium. The testis volume also reaches a maximum in day 2 pupa which then decreases gradually. Histological studies of spermatogenesis of S. mauritia revealed that maximum number of eupyrene and apyrene sperm bundles occurs in day 6 pupa (Venugopalan *et al.*, 1994). However, testicular ecdysteroid titres were on a downward trend during this time probably signalling the impending moult. Hence, increase in sperm bundles may be promoted by the decreasing titre of ecdysteroids thus preparing for adult eclosion. Thus reproductive maturity may be achieved during pupal stadium in response to the decreasing titre of ecdysteroids. Testicular DNA content also increases considerably during the latter half of the pupal stadium while RNA content of the testis increases during the first half of the pupal period of S. mauritia (Benny, 2001) when the testicular ecdysteroid titres were fairly high. Testicular proteins also increases dramatically during the pupal stage reaching a maximum in day 6 pupa (Benny, 2001). Increased titres of testicular ecdysteroids during the first half of pupal stadium has also been reported in *H. virescens* (Loeb et al., 1984) and O. nubilalis. In Ostrinia, this increase in the titre corresponds to the time when apyrene spermatocysts are elongating (Gelman *et al.*, 1988). Testicular ecdysteroids also play an important role in the induction of development of male genital tract of lepidopteran pupae (Nowock, 1972, 1973; Leclerg -Smekens, 1974). Ecdysteroids also interact with haemolymph macromolecular factor to induce spermatocyte differentiation (Kambysellis and Williams, 1971b, 1972) and influence the development of apyrene sperms (Hoffmann and Behrens, 1982; Loeb et al., 1982; Gelman et al., 1989). Ecdysteroids are also believed to be necessary for resumption of spermatogenesis after diapause (Dumser, 1980).

Testicular ecdysteroid titres during adult development

The adults of S. mauritia have a very short life span, hence spermatogenesis is completed in the pupal stage itself. In the adult testis, both eupyrene and apyrene sperm bundles were distributed throughout the lumen of the testis. However, the early germinal cysts were completely absent (Venugopalan et al., 1994). Testicular ecdysteroid titres of adults (day 0) although significant were considerably lower than those of mid-last instar larvae and pupae of S. mauritia. Testes of adult H. virescens were also capable of synthesising ecdysteroids in vitro (Loeb et al., 1984). Large amounts of ecdysteroids were found in the testes of adult male blowfly C. vicina (Koolman et al., 1979). It has been reported that sexual receptivity of adult male mosquitoes is affected by ecdysteroid titres (Beach, 1980). Thorson and Riemann (1982) have shown that photoperiodically controlled release of eupyrene, but not apyrene sperms from lepidopteran testes into the vas deferens is inhibited by injection of 20-hydroxyecdysone. A decline in haemolymph ecdysteroid titres is also essential for the initiation of sperm release from the testes of L. dispar (Giebultowicz et al., 1990). Hence it is possible that ecdysteroid release by the adult testes of insects mediate bursts of sperm release and sexual behaviour.

Data obtained in the present study suggests that testicular ecdysteroid titres in S. mauritia are correlated with testicular development and spermatogenesis. The testicular ecdysteroid profile of S. mauritia compares well with testicular titres determined for other lepidopteran species. Hence, it may be concluded that testicular development and spermatogenesis in S. mauritia and possibly other insect species are influenced greatly if not solely by testicular ecdysteroid titres.

Comparison between testicular and haemolymph ecdysteroid profile during larval-pupal-adult development

In S. mauritia, fluctuations of testicular and haemolymph ecdysteroid titres appear more or less at the same time during the different stages of development. Testicular ecdysteroids exhibited a minor peak during mid-last instar larval period (day 3) which coincided with the formation of spermatids in the testis (Venugopalan et al., 1994). The commitment peak of ecdysteroids in the haemolymph also appeared during this time. This peak was observed at 60 h when 6 h sampling of haemolymph was done and on day 3 when sampling was done on a daily basis. Testicular ecdysteroid titres increased further on day 4 and formed a significant peak on day 5. The differentiation of eupyrene and apyrene sperm bundles was initiated during this phase (day 4 and day 5) of the sixth instar (Venugopalan et al, 1994). Haemolymph ecdysteroid titres also exhibited a similar trend, titres increased on day 4 and formed the moulting peak on day 5. Fusion of the paired testes lobes also occurred during the pharate pupal stage (day 5).

During the pupal stadium, testicular ecdysteroids exhibited a single large peak on day 1 after which titres decreased gradually during the subsequent days. Fully formed eupyrene and apyrene sperm bundles appeared for the first time and a major increase in their number also occurred at this time. In S. mauritia, maximum number of eupyrene and apyrene sperm bundles occurred in day 6 pupae (Venugopalan et al., 1994). Torsion of the fused testis occurred in the young pupa. Testicular volume reaches a maximum on day 2 of the pupal stage and decreases gradually thereafter. Haemolymph ecdysteroid titres of S. mauritia also exhibited an increase in the first half of the pupal stadium and decreased in the latter half. The decrease in haemolymph titres has been demonstrated to be caused by apoptosis of the prothoracic glands during pupal-adult development (Dai and Gilbert, 1997). Whether an apoptotic event occurs in the steroid secreting cells of the testes during pupal-adult development has not been reported in any insect. In the absence of an apoptotic event during pupal-adult testicular development, the pattern of ecdysteroid fluctuations in the testes may be a mere reflection of the haemolymph ecdysteroid titres. Thus testicular ecdysteroids may have been sequestered from the haemolymph ecdysteroid pool. It is also possible that the testes may be a target tissue for the haemolymph ecdysteroids. This facilitates synchronous somatic and testicular development.

Testicular ecdysteroid titres could not be compared with those of haemolymph in the adult stadium as enough blood could not be drawn from the adults to determine haemolymph ecdysteroid titres. However, no ecdysteroids were detected in the haemolymph of adult male blowfly C. vicina by Koolman *et al.* (1979) although large amounts of such compounds were found in the testes. Testicular ecdysteroid titres of day 0 adults of S. mauritia were much lower than mid-instar larvae and pupae. Spermatogenesis in S. mauritia is completed in the pupal stage itself. Hence it is possible that in S. mauritia testicular ecdysteroids in the adult may have been sequestered from the haemolymph during earlier stages of development.

Thus in S. mauritia the total ecdysteroid content of the testes reflect that of the haemolymph during larval and pupal development. Similar observations have also been made in H. virescens (Loeb, 1982), O. nubilalis (Gelman et al., 1988) and pupae of M. sexta (Friedlander and Reynolds, 1992). However titre levels in the haemolymph were much greater than those of the testes in all the insects examined. Ecdysteroid composition of the testes and haemolymph at identical times of development also differed greatly (Gelman et al., 1988; Jarvis et al., 1994). This in itself does not rule out the possibility of sequestration of haemolymph ecdysteroids by the testes. It is always possible that differential uptake of ecdysone and 20-hydroxyecdysone from the haemolymph could occur in the testes thus explaining the observed differences. Thus the present data suggests that spermatogenesis and testicular development of *S. mauritia* and possibly other insects are controlled by both haemolymph and testicular ecdysteroids. It could not be conclusively proved that sequestration of haemolymph ecdysteroids occur in the testes, but there is definitely an undeniable reflection of haemolymph ecdysteroid content in the testicular ecdysteroid levels.

Effects of ligations on testicular ecdysteroid profile

Testicular ecdysteroid titres were reduced considerably following neck ligation on day 1 and titres in the thorax-ligated larvae were reduced to nil four days after ligation. On the other hand, unligated (normal) control larvae exhibited an increasing trend of ecdysteroids reaching a maximum on day 4. The reduction in titres following ligation may be due to the lack of testes ecdysiotropins from the brain which (may) stimulate synthesis and release of $_{,1994b}$ ecdysteroids by the testes of immature insects (Loeb *et al.*, 1987; Loeb, 1994). Although both neck- and thorax-ligated larvae lacked the brain, the suspected source of testis ecdysiotropin, testes of the neck-ligated larvae contained significant amounts of ecdysteroids while titres in the thorax-ligated larvae were reduced to nil. Therefore, the total absence of ecdysteroids in the testes of thorax-ligated larvae may be due to the absence of prothoracic glands. Previous studies demonstrated that haemolymph ecdysteroid titres of *S. mauritia* also decreased following neck and thorax ligation, more so in thorax-ligated larvae. Thus the presence of prothoracic glands and/or haemolymph ecdysteroid titres may be essential for the synthesis of testicular ecdysteroids.

Spermatogenesis in *S. mauritia* was completely blocked in the ligated larvae and the testes contained an undifferentiated mass of tissue (Venugopalan, 1995). As mentioned earlier, the commitment peak of ecdysteroids in the haemolymph and consequently the testicular ecdysteroid peak on day 3 may be promoting maturation of male germ cells. The absence of this peak would have caused the disintegration of the cells.

Earlier studies on S. mauritia have demonstrated that neck/thorax ligation of day 4 larvae prevented pupation. Testes lobes also do not undergo fusion in these larvae. However neck-ligation on day 4 does not prevent the increase in testicular ecdysteroid titres. Same was the case with haemolymph ecdysteroid titres of S. mauritia following neck ligation on day 4. But thorax ligation led to decreased titres both in the testes and haemolymph ecdysteroid titres. Reduction in haemolymph ecdysteroid titres following thorax ligation was due to the absence of prothoracic glands. Thorax-ligated larvae contained the testes, yet titres reduced in the ligated larvae. Hence the testes ecdysteroid titres may be dependent on haemolymph titre or the presence of prothoracic glands as has been previously suggested.

Spermatogenesis was arrested following neck/thorax ligation on day 4 in S. mauritia, the whole testes appeared to be an unorganised mass and spermatocysts were not clearly distinguishable (Venugopalan, 1995). In the neck-ligated larvae, although ecdysteroid titres increased following ligation, the rise in titre is not rapid as in the control (unligated) larvae. The titres in the neck-ligated larvae increased gradually and reached a peak 4 ± 1 days after ligation while titres in the control larvae peaked on day 5. The ecdysteroid titre peak of neck-ligated larvae although delayed was comparable to the peak titre of control larvae. The differentiation of sperm bundles during day 4 - day 5 may be triggered by the ecdysteroid peak on day 5, the absence of which may cause arrest of spermatogenesis in the neck-ligated larvae. The absence of JH in the ligated larvae may also arrest spermatogenesis since JH is reported to promote spermatogenesis during late last larval stage (Yagi, 1975; Nishitsutsuji-Uwo, 1975; Nishitsutsuji-Uwo, 1961; Shinyaeva, 1981; Venugopalan, 1995). The role of ecdysteroids in promoting testicular development has already been addressed in an earlier section. The reduced ecdysteroid titres coupled with the absence of JH in the thorax-ligated larvae may be the cause of disintegration of the testicular tissue. This may also be the reason for the testes lobes not undergoing fusion in the control larvae. The absence of JH which is secreted from the corpora allata in the ligated larvae may further prevent testicular fusion and spermatogenesis. This aspect will be discussed in a later section. Meola and Loeb (1995) suggested

that testis ecdysiotropins may also be involved in the fusion of the paired testis and maturation of the male reproductive system. Hence, the absence of testis ecdysiotropins may also be the reason for the failure of testicular fusion in the ligated larvae.

Effects of JHA on testicular ecdysteroid profile

Inhibitory effects of JHA on testicular ecdysteroid titres

Previous studies have demonstrated that repetitive treatments of unligated sixth instar larvae of S. mauritia with JHA affect the growth, differentiation and developmental competence of imaginal structures resulting in the production of imperfect superlarvae. The testes of these larvae remained separate. Testicular ecdysteroid titres of the treated larvae did not exhibit a peak on day 3 unlike the control (normal) larvae. From earlier studies, it is assumed that this peak is responsible for the change from mitotic to meiotic division of the gonial cells of the testes leading to the formation of spermatids. Venugopalan (1995) reported that JHA treatment interfered with normal testicular development and spermatogenesis during last instar larval development in S. mauritia. The volume of the testes of the treated larvae also exhibited a progressive decline during the post-treatment days. From the present study, it follows that the absence of the testicular ecdysteroid peak on day 3 may be the cause for the arrest of development of gonial cells and hence the lack of spermatid cysts and sperm bundles in the supernumerary larvae.

The major ecdysteroid peak which appeared at the pharate pupal stage (day 5) in the control larvae was also missing in the JHA treated larvae. Instead, a much smaller peak appeared on day 4, one day prior to the supernumerary This may also contribute to the developmental arrest of moult. spermatogenesis and prevention of fusion of testes lobes of the treated larvae. Earlier studies have revealed that the commitment peak and the large premoult peak of haemolymph ecdysteroids failed to appear following JHA treatment to last stage larvae. Haemolymph ecdysteroid titres of the imperfect supernumerary larvae also exhibited a small peak on day 4, prior to the supernumerary moult. Thus the ecdysteroid titres of the testes and haemolymph showed a similar pattern of fluctuation following JHA treatment. It further demonstrates that JHA may be inhibiting spermatogenesis and fusion of testes lobes by inhibiting ecdysteroid production by the prothoracic glands and/or the testes. Similar profile of ecdysteroids in the testes and haemolymph suggests that the testes sequesters haemolymph ecdysteroids. The role of JH and its analogues in inhibiting the secretory activity of the prothoracic glands has already been addressed in the previous chapter. Hence, it is also possible that steroid secreting cells of the testes are inhibited in a similar way by JHA.

Previous studies have shown that neck-/thorax-ligated day 1 larvae treated with JHA remained unchanged. Their testes also did not undergo fusion. Testicular ecdysteroid titres of the neck-ligated, JHA treated larvae were much lower than that of the control larvae. Titres in the thorax-ligated larvae treated with JHA were reduced to nil. Control larval titres were also reduced to nil. Haemolymph ecdysteroid titres of neck-ligated, JHA treated *S. mauritia* larvae also showed low titres. Titres in the thorax-ligated, treated larvae were further reduced. The lower testicular ecdysteroid titres in the JHA treated, neck-ligated larvae suggests an inhibitory action of JHA on biosynthesis of testicular ecdysteroids. This may be due to direct inhibition of testicular synthesis of ecdysteroids by JHA or just a reflection of the reduced haemolymph titres following JHA treatment. In the thorax-ligated, control larvae, absence of prothoracic glands may have caused total absence of ecdysteroids in the testes. Therefore, JHA treatment did not alter that status.

Histological studies of spermatogenesis of S. mauritia revealed a complete inhibition of spermatogenesis following ligation coupled with JHA treatment to day 1 larvae. The volume of the testes of the treated larvae were lower than that of the control larvae. The testes of the treated larvae was a mass of disintegrating cells. The testes of the control larvae also contained an undifferentiated mass of tissue (Venugopalan, 1995). Testicular DNA, RNA and proteins of *S. mauritia* were also considerably reduced following JHA treatment to ligated day 1 larvae (Benny, 2001). From the present studies, it may be assumed that the disintegration of germ cells in the ligated and JHA treated larvae may be due to the reduced ecdysteroid titres in the haemolymph and the testes. Inhibitory effects of JHA on macromolecular synthesis of the testes may also be due to the reduced testicular ecdysteroid titres which in turn may have been affected by the low ecdysteroid titres in the haemolymph of the JHA treated larvae.

Stimulatory effects of JHA on testicular ecdysteroid titres

Previous studies have shown that neck-ligated day 4 larvae of S. mauritia treated with JHA transformed into headless pupae while control (larvae neck-ligated on day 4, untreated), failed to pupate. However, treatment of JHA to thorax-ligated day 4 larvae did not bring about pupation in the treated larvae. Control larvae (larvae thorax-ligated on day 4, untreated) also failed to pupate. Testes lobes of the headless pupae were fused while those of the thorax-ligated, treated larvae and those of controls failed to undergo fusion.

Testicular ecdysteroid titres of the headless pupae were much higher than those of the control larvae (day 7). However, it may be recalled that testicular ecdysteroid titres of neck-ligated, post feeding larvae on day 8 were of the same range as that of the normal prepupal testicular ecdysteroid titres, but the testes did not undergo fusion. This shows that JH has an important role in testicular fusion. Histological studies of the testes of the headless pupa demonstrated that spermatogenesis had progressed and also revealed the presence of apyrene sperm bundles. On the other hand, in the testes of the control larvae, spermatocysts were not clearly distinguishable and the whole testes was an unorganised mass (Venugopalan, 1995). The DNA, RNA and protein content of the testes of the headless pupae also showed very high values compared to controls (Benny, 2001). Results of the present studies suggest that JH or its analogue modifies the response of the target tissues to ecdysteroids and hence may act at the receptor level since delayed increase in ecdysteroid titres in the absence of JH in the control larvae did not cause reinitiation of spermatogenesis but JH treatment did.

The present results suggest that the prepupal increase in JH has a critical role to play not only in pupation but also in promoting spermatogenesis. The role of testis ecdysiotropins in the fusion of testes lobes (Meola and Loeb, 1995) has already been discussed. Testis ecdysiotropins may already be released in the day 4 larvae and it may act in conjunction with JH and ecdysteroids. Earlier studies have shown that haemolymph ecdysteroid titres of headless pupae were higher than those of control (day 7). Thus the testes and haemolymph ecdysteroid titres were similarly affected by JHA treatment.

Testicular ecdysteroid titres of the thorax-ligated JHA treated larvae did not differ much from the control larvae. Titres were found to decrease in both the treated and control larvae. Administration of JHA to these larvae did not bring about pupation and testicular fusion in these larvae. This may be due to the decreased titres of ecdysteroids in the testes following ligation. This shows that successful larval-pupal transformation and spermatogenesis requires a critical titre of ecdysteroids and JH in the haemolymph and testes. The absence of either may affect metamorphosis and spermatogenesis.

Earlier studies on S. mauritia have shown that treatment of JHA to pupae caused high mortality in the treated pupae. The surviving pupae showed various abnormalities in pupal-adult development. Testicular ecdysteroid titres increased considerably in the JHA treated pupae. Increased dosage of JHA elicited higher titres of ecdysteroids. Hence titres were greater in the 1 μ g JHA treated pupae than in the 0.5 μ g JHA treated pupae. Hormone titres increased significantly in the first half of the pupal stage and declined gradually in the latter half both in the treated and control pupae. However ecdysteroid peaks appeared two days later (day 3) in the JHA treated pupae while the peak appeared on day 1 in the control pupae. Earlier studies have shown that haemolymph ecdysteroid titres of pupae also showed considerable increase following JHA treatment and here too the increase in titres was directly proportional to the dosage of JHA. Displacement of peak titres also occurred in the haemolymph where the peak appeared one day later (day 2) in the 0.5 µg JHA treated pupae. Haemolymph titre peaks of control

and 1 μ g JHA treated pupae however occurred on day 1. Thus the haemolymph ecdysteroid titres also increased during the first half of the pupal stadium and declined in the latter half both in the JHA treated and control pupae.

Pupal-adult development normally occur in the absence of JH. Also, a decline in the JH titre during the final larval instar is necessary for the initiation of metamorphosis (Wigglesworth, 1955; Gilbert, 1962). It has already been discussed that the prothoracic glands undergo apoptosis during pupal-adult development (Dai and Gilbert, 1997). JHA treatment prevents apoptosis (Dai and Gilbert, 1998) and causes an increase in ecdysteroid titres in the haemolymph. Whether apoptosis occurs in the steroid secreting cells of the testes is not known. However, the reduction in titres during the latter half of (normal) pupal development and increased synthetic rates following JHA treatment point to such a possibility. It may also be possible that these fluctuations in the testicular titres are caused by fluctuations in the haemolymph titres.

JHA treatment also caused developmental arrest of spermatogenesis and necrosis of already formed sperm bundles of *S. mauritia* pupae (Venugopalan, 1995). From the present study it may be assumed that lysis of the fully formed sperm bundles may be caused by the unusually high ecdysteroid titres following JHA treatment. It may be recalled that a significant increase in sperm bundles occurred during the latter half (day 6) of pupal development when the ecdysteroid titres were on a downward trend. But JHA treatment maintained abnormally high ecdysteroid titres even during the latter half of pupal development which may be responsible for the arrest of spermatogenesis and necrosis of already formed sperm bundles. Further, JH is normally not present during pupal development and its occurrence may directly cause disturbances in testicular development.

Effects of ecdysteroid agonist RH 5992 on testicular ecdysteroid profile

Effects of RH 5992 on ligated sixth instar larvae

Non-steroidal ecdysteroid agonists (RH 5849, RH 5992) possess strong ecdysonergic activities, inducing a premature, lethal moult in the treated larvae of many lepidopterans and other insect species (Wing *et al.*, 1988; Monthean and Potter, 1992; Smagghe and Degheele, 1994; Sakunthala and Nair, 1995). The present study also demonstrated that RH 5992 induced a precocious moult in larvae of *S. mauritia*. Treatment of different dosages of RH 5992 to neck-/thorax-ligated day 1 larvae induced a decrease in testicular ecdysteroid titres on day 1 and day 2. An abrupt increase in titres occurred on day 3 in the neck-ligated larvae treated with 5 μ g RH 5992 while those in others continued to decrease. Titres decreased in the control larvae also. However the decrease in titres was more pronounced in the thorax-ligated larvae. Histological studies revealed that the treatment of ecdysteroid agonist neck-/thorax-ligated dav 1 larvae accelerated to spermatogenesis (Venugopalan, 1995). The presence of well differentiated spermatogonial, spermatocytal and spermatidal cysts and eupyrene sperm bundles were observed in the testes of the treated larvae while spermatogenesis was completely arrested in the control larvae. The testes lobes were separate in the treated and control larvae but volumes of testes of the treated larvae were higher than those of the control larvae. Testicular ecdysteroid titres of the RH 5992 treated larvae were lower than those of the control larvae on most days, yet the acceleration of spermatogenesis suggested that RH 5992 acted at the level of ecdysteroid receptors (ER). The prothoracic glands have been reported to possess ER (Bidmon and Sliter, 1990; Talbot et al., 1993). It is possible that the testes sheath may also possess ER through which RH 5992 and haemolymph ecdysteroids regulate testicular ecdysteroid titres and spermatogenesis. Thus, the testes may act as a target tissue for haemolymph ecdysteroid titres or ecdysteroid mimics. It has been reported that nonsteroidal ecdysteroid agonists cause precocious moulting in treated larvae by binding to the ER of epidermal cells (Smagghe and Degheele, 1992) without causing an increase in haemolymph ecdysteroid titres. A similar mode of action may be operating in the testes of the treated larvae which causes the acceleration of spermatogenesis without an increase in testicular ecdysteroid titres. It is also possible that the decrease in titres of testicular ecdysteroids
may be a reflection of the decline in the haemolymph ecdysteroid titres in the RH 5992 treated larvae. The spontaneous increase in testicular ecdysteroid titres in some of the treated larvae may also be a reflection of the haemolymph ecdysteroid titres or due to a positive feedback effect of RH 5992 on the steroid secreting cells of the testis sheath. Ecdysteroids and ecdysteroid agonists have been reported to have such a feedback effect on prothoracic glands of insects (Williams, 1952; Siew and Gilbert, 1971; Sakurai and Williams, 1989).

Previous studies showed that treatments of different dosages of RH 5992 to neck-/thorax-ligated day 4 larvae induced the development of larval-pupal intermediates. The testes of these larvae underwent fusion while control larvae remained unchanged and the testes lobes failed to fuse. Testicular ecdysteroid titres were found to decrease following treatment of RH 5992 in the neck-ligated and thorax-ligated larvae but for a few exceptions. However, titres in the control larvae increased gradually in the case of the neck-ligated ones but decreased further in the thorax-ligated larvae. Here too, histological studies revealed that treatment of ecdysteroid agonist to neck-/thorax-ligated day 4 larvae accelerated spermatogenesis (Venugopalan, 1995). Eupyrene sperm bundles were observed in the treated larvae but apyrene sperm bundles were not clearly distinguishable and the whole testes was an unorganised mass. Similar results were obtained with treatment of RH 5992 to C. pomonella which induced spermatogenesis reinitiation in isolated abdomens of diapausing codling moth larvae (Friedlander and Brown, 1995). Testicular ecdysteroid titres of neck-ligated, control larvae (day 7) were almost the same range as the normal prepupal (day 5) testicular titres. However, spermatogenesis was arrested and the testes remained separate in the control larvae. On the other hand, RH 5992 treatment accelerated spermatogenesis and testis fusion both in the neck- and thorax-ligated day 4 larvae even at low testicular ecdysteroid titres. This suggested RH 5992 that stimulates testicular development and spermatogenesis at the level of the ER of the testis as has been previously discussed. Induction of larval-pupal intermediate formation by RH 5992 treatment also occurred at low haemolymph ecdysteroid titres. It has been suggested that RH 5992 caused spermatogenesis renewal in isolated abdomens of diapausing C. pomonella larvae by attaching to and saturating, a specific number of receptors (Friedlander and Brown, 1995). Hence, spermatogenesis renewal may not have occurred in the control (neck-ligated) larvae because the "ecdysteroid peak" for saturation of the ER of the testes was delayed and not available during the prepupal period (day 5). In the absence of the required peak titre of ecdysteroids, the germinal cysts would have degenerated. Hence, it may be postulated that a critical titre of ecdysteroids during a specific period of time in the testes and/or haemolymph

is essential for normal progression of spermatogenesis without which the germinal cysts would undergo degeneration.

Effects of RH 5992 on pupae

Treatment of pupae with different dosages of RH 5992 induced the development of adultoids in S. mauritia as has been previously reported. However histological studies of the testes of RH 5992 treated S. mauritia pupae failed to reveal any abnormalities in testicular development or spermatogenesis (Manogem and Nair, unpublished results). Testicular ecdysteroids also did not show any variation after RH 5992 treatment in the pupae. Testicular ecdysteroid titres were similar in the treated and control pupae on the days examined (day 2 and day 5). Haemolymph ecdysteroid titres however did not exhibit similar titres in the RH 5992 treated and control pupae. This is the first instance where testicular and haemolymph ecdysteroid titres differed in S. mauritia. It has been suggested that RH 5992 causes acceleration of spermatogenesis at the level of ecdysteroid receptors. RH 5992 attaches to and saturates a specific number of receptors and surplus of RH 5992 do not disrupt spermatogenesis and physiological function (Friedlander and Brown, 1995). Hence, RH 5992 treatment does not have any effect on spermatogenesis or testicular ecdysteroid titres during the pupal stage of S. mauritia.

160

4.5 SUMMARY

- 1. Testicular ecdysteroid profile of *S. mauritia* during the sixth (final) larval instar, pupal and adult stadia was determined by means of RIA. Effects of hormones and hormone mimics on the ecdysteroid content of the testes were studied by means of ligation and treatments of a JHA (hydroprene) and a non-steroidal ecdysteroid agonist (RH 5992). Control insects were treated with acetone. The function of testicular ecdysteroid titres was analysed by correlating it with known developmental stages of spermatogenesis in *S. mauritia*.
- 2. Testicular ecdysteroid profile of sixth instar larvae revealed two peaks, a small peak on day 3 and a large peak on day 5. The paired larval testes fused to form a single median organ on day 5 of the sixth larval instar. Ecdysteroid titres of the pupal testes exhibited a single large peak on day 1 of the pupal stadium. Ecdysteroid titres decreased considerably in the newly emerged adult.
- 3. Influence of brain factors and prothoracic gland secretions on testicular ecdysteroid titres were studied by means of neck/thorax ligation of day 1 and day 4 sixth instar larvae. The paired larval testes failed to undergo fusion in the neck-/thorax-ligated day 1 and day 4 larvae. Testicular ecdysteroid titres of the neck-/thorax-ligated day 1 larvae were significantly reduced compared to the control larvae. Titres of the neck-ligated day 4 larvae increased gradually to reach a peak on day 8 which was not significantly different from the ecdysteroid titre peak on day 5 (prepupae) of the control larvae while thorax ligation of day 4 larvae led to a significant decrease in testicular ecdysteroid titres.
- 4. The influence of JH on testicular ecdysteroid titres were investigated by topical application of JHA to unligated day 0 larvae and neck-/thorax-ligated day 1 and day 4 sixth instar larvae and freshly ecdysed intact pupae of S. mauritia.
- 5. Repetitive treatments of 20 µg JHA to intact sixth instar larvae induced a supernumerary larval moult on day 5. The testes of the supernumerary larvae remained separate. JHA treated larvae exhibited a single peak on day 4 which was significantly lower than the testicular ecdysteroid peak on day 5 of the control larvae. The smaller peak observed on day 3 of the control larvae also failed to appear in the JHA treated larvae.
- 6. Repetitive treatments of 5 μ g JHA to neck-/thorax-ligated day 1 larvae prevented metamorphic changes in the treated larvae. The testes

4.5 SUMMARY

- 1. Testicular ecdysteroid profile of *S. mauritia* during the sixth (final) larval instar, pupal and adult stadia was determined by means of RIA. Effects of hormones and hormone mimics on the ecdysteroid content of the testes were studied by means of ligation and treatments of a JHA (hydroprene) and a non-steroidal ecdysteroid agonist (RH 5992). Control insects were treated with acetone. The function of testicular ecdysteroid titres was analysed by correlating it with known developmental stages of spermatogenesis in *S. mauritia*.
- 2. Testicular ecdysteroid profile of sixth instar larvae revealed two peaks, a small peak on day 3 and a large peak on day 5. The paired larval testes fused to form a single median organ on day 5 of the sixth larval instar. Ecdysteroid titres of the pupal testes exhibited a single large peak on day 1 of the pupal stadium. Ecdysteroid titres decreased considerably in the newly emerged adult.
- 3. Influence of brain factors and prothoracic gland secretions on testicular ecdysteroid titres were studied by means of neck/thorax ligation of day 1 and day 4 sixth instar larvae. The paired larval testes failed to undergo fusion in the neck/thorax-ligated day 1 and day 4 larvae. Testicular ecdysteroid titres of the neck/thorax-ligated day 1 larvae were significantly reduced compared to the control larvae. Titres of the neck-ligated day 4 larvae increased gradually to reach a peak on day 8 which was not significantly different from the ecdysteroid titre peak on day 5 (prepupae) of the control larvae while thorax ligation of day 4 larvae led to a significant decrease in testicular ecdysteroid titres.
- 4. The influence of JH on testicular ecdysteroid titres were investigated by topical application of JHA to unligated day 0 larvae and neck-/thorax-ligated day 1 and day 4 sixth instar larvae and freshly ecdysed intact pupae of S. mauritia.
- 5. Repetitive treatments of 20 µg JHA to intact sixth instar larvae induced a supernumerary larval moult on day 5. The testes of the supernumerary larvae remained separate. JHA treated larvae exhibited a single peak on day 4 which was significantly lower than the testicular ecdysteroid peak on day 5 of the control larvae. The smaller peak observed on day 3 of the control larvae also failed to appear in the JHA treated larvae.
- 6. Repetitive treatments of 5 μ g JHA to neck-/thorax-ligated day 1 larvae prevented metamorphic changes in the treated larvae. The testes

remained separate in the JHA treated and control larvae and testes volumes decreased in the treated larvae. Testicular ecdysteroid titres of the neck-ligated, JHA treated larvae decreased significantly when compared to those of control larvae while titres of both thorax-ligated, JHA treated and control larvae were reduced to nil.

- 7. Treatment (single) of 5 µg JHA to neck-ligated day 4 larvae induced the formation of headless pupae with fused testes lobes while control larvae failed to pupate and the testes lobes remained separate. However, thorax-ligated, JHA treated and control larvae failed to pupate and the testes lobes remained separate. Testicular ecdysteroid titres of the headless pupae were significantly higher than those of control larvae. But testicular titres of the thorax-ligated, JHA treated and control larvae failed to control larvae. But testicular titres of the thorax-ligated, JHA treated and control larvae for the thorax-ligated, JHA treated and control larvae.
- 8. Treatment of different dosages of JHA (0.5 μg or 1 μg) to newly ecdysed pupae induced the formation of adultoids which failed to emerge. Testicular ecdysteroid titres of JHA treated pupae were significantly higher than those of control pupae on most days. Ecdysteroid titres increased greatly in the early pupal stage and declined in the latter period both in the JHA treated and control pupae. JHA appears to have a tropic effect on testicular ecdysteroid titres.
- 9. The influence of RH 5992 on testicular ecdysteroid titres was investigated by topical application of RH 5992 to neck-/thorax-ligated day 1 and day 4 sixth instar larvae and intact pupae of *S. mauritia*.
- 10. Treatment of RH 5992 (2 µg or 5 µg) to neck-/thorax-ligated day 1 larvae induced a precocious and lethal moult in the ligated larvae. The testes lobes remained separate in the treated and control larvae. However, the testes volumes increased in the treated larvae. Testicular ecdysteroid titres decreased further in the RH 5992 treated larvae but a significant but abrupt increase in titres occurred on day 3 in the 5 μ g RH 5992 treated neck-ligated larvae when compared to titres of the control larvae. Testicular ecdysteroid titres of RH 5992 treated, thoraxligated larvae were lower than those of controls on most days. The abrupt rise in titres which occurred in some of the treated larvae were not significant. RH 5992 acts at the (ecdysteroid) receptor level without a rise in ecdysteroid titres and hence the low titres on day 1 and day 2. The increase in titres on day 3 may be due to positive feedback effect of RH 5992 on steroid secreting cells of the testes or a reflection of the fluctuating haemolymph titres.

- Treatment of RH 5992 (2 µg or 5 µg or 10 µg) to neck-/thorax-ligated 11. day 4 larvae induced the development of larval-pupal intermediates with fused testes lobes. The control larvae remained unchanged and the testes lobes failed to undergo fusion. Testicular ecdysteroid titres of larval-pupal intermediates were significantly lower than those of controls in 5 μ g and 10 μ g RH 5992 treated larvae. Titres of the RH 5992 treated larvae were lower than those of the control larvae on most days while control larval titres exhibited an increasing trend reaching its maximum on day 7. Testicular ecdysteroid titres of the thorax-ligated, RH 5992 treated and control larvae decreased considerably. Testicular titres although lower in the RH 5992 treated larvae were not significantly different from those of controls. The decrease in ecdysteroid titres may be due to negative feedback effect of RH 5992 on the steroid secreting cells of the testes or a reflection of the fluctuating haemolymph titres.
- 12. RH 5992 treatment to pupae induced formation of adultoids. The pupal period got extended following treatment and mortality occurred towards the end of the pupal period. However testicular ecdysteroid titres of treated and control pupae did not show a significant difference on day 2 and day 5. Hence RH 5992 may not have any effect on the testicular ecdysteroid content of pupae.
- Ecdysteroid contents of the testes and haemolymph were compared 13. during larval-pupal and pupal-adult development. Testicular ecdysteroid levels during larval-pupal development increased on day 3 and showed a significant peak on day 5. Haemolymph ecdysteroid titres during larval-pupal development also exhibited a similar pattern of fluctuation. During pupal-adult development, testicular ecdysteroid titres exhibited a single large peak on day 1 and haemolymph titres peaked on day 2. Testicular and haemolymph ecdysteroid titres exhibited a single large and significant peak in the first half of pupal development and decreased in the latter half. Thus, pattern of fluctuation of testicular and haemolymph ecdysteroids were similar which suggest that testicular ecdysteroids may be sequestered from the haemolymph ecdysteroid pool. It is also possible that testicular and haemolymph ecdysteroid titres may be affected by other hormones in a similar manner.

REFERENCES

i p

REFERENCES

- Agui N. and Hiruma K. (1977a) In vitro activation of neurosecretory brain cells in Mamestra brassicae by β -ecdysone. Gen. Comp. Endocr. 33, 467-472.
- Agui N. and Hiruma K. (1977b) Ecdysone as a feedback regulator for the neurosecretory brain cells in *Mamestra brassicae*. J. Insect Physiol. 23, 1393-1396.
- Agui N. and Hiruma K. (1982) Ecdysteroid titre and its critical period during larval and pupal ecdysis in the cabbage armyworm, *Mamestra brassicae* L. (Lepidoptera, Noctuidae). Appl. Entomol. Zool. 17, 144-146.
- Agui N., Bollenbacher W.E., Granger N.A. and Gilbert L.I. (1980) Corpus allatum is release site for insect prothoracicotropic hormone. *Nature* 285, 669-670.
- Agui N., Granger N.A., Gilbert L.I. and Bollenbacher W.E. (1979) Cellular localization of the insect prothoracicotropic hormone: *In vitro* assay of a single neurosecretory cell. *Proc. Nat. Acad. Sci.* **76**, 5694-5698.
- Agui N., Kimura Y. and Fukaya M. (1972) Action of the prothoracic gland on the insect integument *in vitro*. Appl. Entomol. Zool. 7, 71-78.
- Aller H.E. and Ramsay J.R. (1988) RH 5849 A novel insect growth regulator with a new mode of action. BCPC – Pests and Diseases 5, 511.
- Baker F.C., Tsai L.W., Reuter C.C. and Schooley D.A. (1987) In vivo fluctuation of JH, JH acid and ecdysteroid titer, and JH esterase activity, during development of fifth stadium Manduca sexta. Insect Biochem. 17, 989-996.
- Balamani E. and Nair V.S.K. (1989a) Prothoracicotropic action of a juvenile hormone analogue on neck-ligated last instar larvae of *Spodoptera* mauritia Boisd. (Lepidoptera: Noctuidae). Entomon. 14, 283-288.
- Balamani E. and Nair V.S.K. (1989b) Effects of antijuvenile hormone agents and a juvenile hormone analogue on neck-ligated post-feeding last instar larvae of Spodoptera mauritia Boisd. (Lepidoptera: Noctuidae). Insect Sci. Appl. 10, 583-589.

- Balamani E. and Nair V.S.K. (1991). On the morphogenetic role of juvenile hormone analogue in the prepupal stage of Spodoptera mauritia Boisd. (Lepidoptera: Noctuidae) Inverteb. Reprod. Dev. 20, 175-179.
- Balamani E. and Nair V.S.K. (1992). Inhibitory effect of a juvenile hormone analogue on prothoracic gland activity in the penultimate and last larval development of Spodoptera mauritia Boisd. (Lepidoptera: Noctuidae). Zool. Anz. 228, 182-187.
- Balamani E. and Nair V.S.K. (1994) Ultrastructural changes in the prothoracic gland cells in relation to larval-pupal transformation of Spodoptera mauritia. Proc. Indian natn. Sci. Acad. B60: 6, 483-490.
- Balamani E. and Nair V.S.K. (1997) Cellular changes in relation to secretory activity of prothoracic glands during the larval development and metamorphosis of Spodoptera mauritia Boisd. (Lepidoptera: Noctuidae). J. Anim. Morphol. Physiol. 44: 1, 33-40.
- Beach R. (1980) Physiological changes governing the onset of sexual receptivity in male mosquitoes. J. Insect Physiol. 26, 245-252.
- Bean D.W., Beck S.D. and Goodman W.G. (1982) Juvenile hormone esterases in diapause and non-diapause larvae of European comborer, Ostrinia nubilalis. J. Insect Physiol. 28, 485-492.
- Beck S.D. (1980) Insect Photoperiodism. 2nd edition. Academic Press, New York.
- Benedeczky I., Mala J. and Sehnal F. (1980) Ultrastructural study on the innervation of prothoracic glands in *Galleria mellonella*. Gen. Comp. Endocr. 41, 400-407.
- Benny T.M. (2001) Studies on the spermatogenesis in Spodoptera mauritia Boisd. (Lepidoptera: Noctuidae). Ph.D. Thesis, University of Calicut.
- Benny T.M: and Nair V.S.K. (1999) Involvement of ecdysteroids in the fusion of testes lobes of Spodoptera mauritia Boisd. (Lepidoptera: Noctuidae). J. ent. Res.. 23(4), 343-345.
- Bergot B.J., Jamieson G.C., Ratcliff M.A. and Schooley D.A. (1980) JH zero: new naturally occurring insect juvenile hormone from developing embryos of the tobacco hornworm. *Science* 210, 336-338.
- Beydon P. and Lafont R. (1983) Feedback inhibition of ecdysone production by 20-hydroxyecdysone in *Pieris brassicae* pupae. J. Insect Physiol. 29, 529-533.

- Bidmon H.J. and Koolman J. (1989) Ecdysteroid receptors located in the central nervous system of an insect. *Experientia* 45, 106-109.
- Bidmon H.J. and Sliter T.J. (1990) The ecdysone receptor. Invert. Reprod. Devel. 18, 13-27.
- Birkenbeil H., Eckert M. and Gersch M. (1979) Electron microscopicalimmunocytochemical evidence of ecdysteroids in the prothoracic gland of *Galleria mellonella*. *Cell Tissue Res.* 200, 285.
- Bollenbacher W.E. and Bowen M.F. (1983) The prothoracicotropic hormone. In "Endocrinology of Insects" (Ed. by Downer R.G.H. and Laufer H.) pp.88-89. Alan R. Liss, New York.
- Bollenbacher W.E. and Gilbert L.I. (1981) Neuroendocrine control of postembryonic development in insects, the prothoracicotropic hormone. In "Neurosecretion: Molecules, Cells, Systems" (Ed. by Farner D.S. and Lederis K.) pp.361-370. Plenum Press, New York.
- Bollenbacher W.E. and Granger N.A. (1985) Endocrinology of the prothoracicotropic hormone. In "Comprehensive Insect Physiology, Biochemistry and Pharmacology" (Ed. by Kerkut G.A. and Gilbert L.I.) Vol. 7, pp.109-151. Pergamon Press, Oxford.
- Bollenbacher W.E., Agui N., Granger N.A. and Gilbert L.I. (1979) In vitro activation of insect prothoracic glands by the prothoracicotropic hormone. Proc. Natn. Acad. Sci. 76, 5148-5152.
- Bollenbacher W.E., Agui N., Granger N.A. and Gilbert L.I. (1980) Insect prothoracic glands in vitro: a system for studying the prothoracicotropic hormone. In "Invertenbrate Systems In vitro" (Ed. by Kurstak E., Maramorosch K. and Dubendorfer A.) pp.253-271, Elsevier, Amsterdam.
- Bollenbacher W.E., Goodman W., Vedeckis W.V. and Gilbert L.I. (1976) The *in vitro* synthesis and secretion of α -ecdysone by the ring glands of the fly Sarcophaga bullata. Steroids 27, 309-324.
- Bollenbacher W.E., Smith S.L., Goodman W. and Gilbert L.I. (1981) Ecdysteroid titer during larval-pupal-adult development of the tobacco hornworm *Manduca sexta. Gen. Comp. Endocr.* 44, 302-306.
- Bollenbacher W.E., Vedeckis W.V., Gilbert L.I. and O'Connor J.D. (1975) Ecdysone titers and prothoracic gland activity during the larval-pupal development of *Manduca sexta*. *Dev. Biol.* 44, 46-53.

- Bollenbacher W.E., Zvenko H., Kumaran A.K. and Gilbert L.I. (1978) Changes in ecdysone content during postembryonic development of the wax moth, Galleria mellonella: the role of the ovary. Gen. Comp. Endocr. 34, 169-179.
- Borst D.W. and Engelmann F. (1974) In vitro secretion of α-ecdysone by prothoracic glands of a hemimetabolous insect Leucophaea maderae (Blattaria). J. Exp. Zool. 189, 413-419.
- Borst D.W. and O'Connor J.D. (1974) Trace analysis of ecdysones, by gasliquid chromatography, radioimmunoassay and bioassay. *Steroids* 24, 637-655.
- Burdette W.J. (1962) Changes in titer of ecdysone in *Bombyx mori* during metamorphosis. *Science* 135, 432.
- Butenandt A. and Karlson P. (1954) Über die isolierung eines metamorphosehormone der inseckten in kristallisierten form. Z. Naturf. 9b, 389-391.
- Calvez B., Hirn M. and De Reggi M. (1976) Ecdysone changes in the haemolymph of two silkworms (Bombyx mori and Philosamia cynthia) during larval and pupal development. FEBS Letters 71, 57-61.
- Chino H., Sakurai S., Ohtaki T., Ikekawa N., Miyazaki H., Ishibashi M. and Abuki H. (1974) Biosynthesis of α-ecdysone by prothoracic glands in vitro. Science 183, 529-530.
- Chippendale G.M. and Turunen S. (1981) Hormonal and metabolic aspects of the larval diapause of the southwestern comborer, *Diatraea* grandiosella (Lepidoptera: Pyralidae). Ent. Gen. 7, 223-231.
- Chippendale G.M. and Yin C.M. (1979) Larval diapause of the European cornborer, Ostrinia nubilalis: further experiments examining its hormonal control. J. Insect Physiol. 25, 53-58.
- Ciancio M.J., Watson R.D. and Bollenbacher W.E. (1986) Competency of Manduca sexta prothoracic glands to synthesize ecdysone during development. Molec. Cell Endocr. 44, 171-178.
- Ciemior K.E., Sehnal F. and Schneiderman H.A. (1979) Moulting, growth and survival of *Galleria mellonella* L. treated with juvenoids. Z. ang. Ent. 88, 414-425.
- Clarke K.V. and Langley P.A. (1963) Studies on the initiation of growth and moulting in *Locusta migratoria migratoides* R. & F. IV. The

relationship between the stomatogastric nervous system and neurosecretion. J. Insect Physiol. 9, 423-430.

- Cymborowski B. and Stolarz G. (1979) The role of juvenile hormone during larval-pupal transformation of *Spodoptera littoralis*: switchover in the sensitivity of the prothoracic gland to juvenile hormone. J. Insect Physiol. 25, 939-942.
- Cymborowski B. and Zimowska G. (1984) Switchover in the sensitivity of the prothoracic glands to juvenile hormone in the cotton leafworm, Spodoptera littoralis. J. Insect Physiol. 30, 911-918.
- Dai J.D. and Gilbert L.I. (1997). Programmed cell death of the prothoracic glands of *Manduca sexta* during pupal-adult metamorphosis *Insect Biochem. Mol. Biol.* 27, 69-78.
- Dai J.D. and Gilbert L.I. (1998). Juvenile hormone prevents the onset of programmed cell death in the prothoracic glands of *Manduca sexta*. *Gen. Comp. Endocr.* 109, 155-165.
- Darvas B. and Varjas L. (1990). Insect growth regulators. In Armored Scale Insects, Their Biology, Natural Enemies and Control (Ed. by Rosen D.) Vol.B., pp.393-408. Elsevier Science Publishers, B.V. Amsterdam.
- Darvas B., Polgar L., Tag El-din M.H., Eross K. and Wing K.D. (1992) Developmental disturbances in different insect orders caused by an ecdysteroid agonist, RH-5849. J. Econ. Ent. 85, 2107-2112.
- Dean R., Bollenbacher W.E., Locke M., Smith S.L. and Gilbert L.I. (1980) Haemolymph ecdysteroid levels and cellular events in the intermoult/ moult sequence of *Calpodes ethlius*. J. Insect Physiol. 26, 267-280.
- Deb D.C. and Chakravorty (1981) Juvenoid-induced effects on the growth and differentiation of testis in the rice moth, Corcyra cephalonica. J. Insect Physiol. 27, 397-402.
- Delbecque J.P. and Slama K. (1980) Ecdysteroid titres during autonomous metamorphosis in a Dermistid beetle. Z. Naturf. 35C, 1066-1080.
- Delbecque J.P., Connat J.L. and Lafont R. (1988) Polar and apolar metabolites of ecdysteroids during the metamorphosis of *Tenebrio* molitor. J. Insect Physiol. 34, 619-623.
- Delbecque J.P., Weidner K. and Hoffmann K.H. (1990) Alternative sites for ecdysteroid production in insects. *Invert. Reprod. Devel.* 18, 29-42.

- Denlinger D.L. (1979) The physiology of pupal diapause in fleshflies. In "Current Topics in Insect Endocrinology and Nutrition" (Ed. by Bhaskaran G., Friedman S. and Rodriguez J.G.) pp.131-160. Plenum Press, New York.
- Denlinger D.L. (1985) Hormonal control of diapause. In "Comprehensive Insect Physiology, Biochemistry and Pharmacology" (Ed. by Kerkut G.A. and Gilbert L.I.) Vol. 8, pp.353-412. Pergamon Press, Oxford.
- Dumser J.B. (1980). The regulation of spermatogenesis in insects. Ann. Rev. Entomol. 25, 341-369.
- Dumser J.B. and Davey K.G. (1975). The *Rhodnius* testis: hormones, differentiation of germ cells and duration of the moulting cycle. *Can. J. Zool.* **52**, 1682-1689.
- Edwards J.P., Corbitt T.S., Mc Ardle H.F., Short J.E. and Weaver R.J. (1995)
 Endogenous levels of insect juvenile hormones in larval, pupal and adult stages of the tomato moth *Lacanobia oleracea*. J. Insect Physiol. 41, 641-651.
- Ephrussi B. and Beadle G.W. (1936). A technique of transplantation for Drosophila Amer. Nat. 70, 218-225.
- Fain M.J. and Riddiford L.M. (1975) Juvenile hormone titers in the haemolymph during late larval development of the tobacco hornworm, *Manduca sexta* (L.) *Biol. Bull.* 149, 506-512.
- Fain M.J. and Riddiford L.M. (1976) Reassessment of the critical period for prothoracicotropic hormone and juvenile hormone secretion in the larval molt of the tobacco hornworm, Manduca sexta. Gen. Comp. Endocr. 30, 131-141.
- Friedlander M. and Brown J.J. (1995). Tebufenozide (Mimic R) a nonsteroidal ecdysone agonist, induces spermatogenesis reinitiation in isolated abdomens of diapausing codling moth larvae (Cydia pomonella) J. Insect Physiol. 41, 403-411.
- Friedlander M. and Reynolds S.E. (1992) Intratesticular ecdysteroid titres and the arrest of sperm production during pupal diapause in the tobacco hornworm, *Manduca sexta. J. Insect Physiol.* 38, 693-703.
- Fugo H., Yamauchi M. and Dedos S.G. (1995). Testicular ecdysteroid level in the silk moth *Bombyx mori*, with special reference to heat treatment during the wandering stage. *Zoological Science*. **12**, 783-788.

- Fujishita M. and Ishizaki H. (1982) Temporal organization of the endocrine events with particular reference to the circadian clock in the larvalpupal development of the saturniid, Samia cynthia ricini. J. Insect Physiol. 28, 77-84.
- Fujishita M., Ohnishi E. and Ishizaki H. (1982). The role of ecdysteroids in the determination of gut-purge timing in the saturniid Samia cynthia ricini. J. Insect Physiol. 28, 961-967.
- Fukaya M. and Kobayashi M. (1966) Some inhibitory action of corpora allata in diapausing larvae of the rice stem borer, *Chilo suppressalis* Walker (Lepidoptera: Pyralidae). Appl. Ent. Zool. 1, 125-129.
- Fukaya M. and Mitsuhashi J. (1961) Larval diapause in the rice stem borer with special reference to its hormonal mechanism. Bull. Nat. Inst. Agri. Sci., 13, 1-30.
- Fukuda S. (1940) Induction of pupation in silkworm by transplanting the prothoracic gland. *Proc. Imp. Acad. Tokyo.* 16, 414-416.
- Fukuda S. (1944). The hormonal mechanism of larval moulting and metamorphosis in the silkworm. J. Fac. Sci. Univ. Tokyo. Sect. IV. Zool. 6, 478-532.
- Fukushima T. and Yagi S. (1975). Hormonal effect on cultivated insect tissues. III effects of α and β ecdysone or prothoracic glands on spermiogenesis in two noctuid insects *in vitro* (Lepidoptera: Noctuidae). *Appl. Entomol. Zool.* **10**, 220-225.
- Gelman D.B. and Borkovec, A.B. (1986). Head critical periods for controlling apyrene and eupyrene spermiogenesis in the European corn borer Ostrinia nubilalis. In "Insect Neurochemistry and Neurophysiology" (Ed. by Borkovec A.B. and Gelman D.B.) pp.371-374. Humana Press, Clifton, N.J.
- Gelman D.B. and Brents L.A. (1984) Hemolymph ecdysteroid levels in diapause- and nondiapause-bound fourth and fifth instars and in pupae of the European corn borer, Ostrinia nubilalis (Hübner). Comp. Biochem. Physiol. 78(A), 319.
- Gelman D.B. and Hayes D. (1982) Critical periods for the brain and prothoracic glands of 5th instars of the European cornborer, Ostrinia nubilalis Hübner. Comp. Biochem. Physiol. **73A**, 81-87.
- Gelman D.B. and Woods C.W. (1983) Haemolymph ecdysteroid titres of diapause- and nondiapause bound fifth instars and pupae of the

European corn borer, Ostrinia nubilalis (Hübner). Comp. Biochem. Physiol. 76(A), 367.

Gelman D.B., Khalidi A.A. and Loeb M.J. (1997) Improved techniques for the rapid radioimmunoassay of ecdysteroids and other metabolites. *Invert. Reprod. Devel.* 32: 2, 127-129.

- Gelman D.B., Woods C.W. and Borkovec A.B. (1985). Metabolism of ecdysone by testes of the European Corn borer, Ostrinia nubilalis (Hubner) Am. Zool. 25, 737.
- Gelman D.B., Woods C.W. and Borkovec A.B. (1988a) Ecdysteroid profiles for haemolymph and testes from larvae, pupae and pharate adults of European corn borer, Ostrinia nubilalis. Arch. Insect Biochem. Biophys, 7, 267-279.
- Gelman D.B., Woods C.W. and Borkovec A.B. (1988b) Effects of ecdysone and 20-hydroxy ecdysone on apyrene spermiogenesis in the European corn borer, Ostrinia nubilalis J. Insect Physiol. 34, 733-738.
- Gelman D.B., Woods C.W. and Borkovec A.B. (1989) Ecdysteroid synthesis by testes of sixth instars and pupae of the European corn borer Ostrinia nubilalis. Invertebr. Reprod. Dev. 15, 177-184.
- Gersch M. (1978) Die Klassiche Vorstellun der hormonalen steuerung der Insektenent wicklung in Lichte neuer experimenteller Tatsachen: Produktion von Haütungshormon anBerhalb der Prothoracaldrüse. Zool. Jb. Physiol. 82, 171-184.
- Gersch M. and Birkenbeil H. (1980) cAMP as a possible mediator of the neuro-hormonal influence on the membrane potential of prothoracic gland cells of *Galleria mellonella*. Zool. Jb. Physiol. 84, 58-66.
- Gibbs D. and Riddiford L.M. (1977) Prothoracicotropic hormone in Manduca sexta: Localization by a larval assay. J. Exp. Biol. 66, 255-266.

- Giebultowicz J.M., Feldlaufer M. and Gelman D.B. (1990) Role of ecdysteroids in the regulation of sperm release from the testis of the gypsy moth Lymantria dispar. J. Insect Physiol. 36, 567-571.
- Giebultowicz J.M., Loeb M.J. and Borkovec A.B. (1987) In vitro spermatogenesis in lepidopteran larvae. Role of the testis sheath. Int. J. Invertbr. Reprod. Dev. 11, 211-226.
- Gilbert L.I. (1962). Maintenance of the prothoracic gland by the juvenile hormone in insects. *Nature* 193, 1205-1207.
- Gilbert L.I., Bollenbacher W.E., Agui N. Granger N.A., Sedlak B.J., Gibbs D. and Buys C.M. (1981) The prothoracicotropes: source of the prothoracicotropic hormone. *Amer. Zool.* 21, 641-653.
- Gilbert L.I., Bollenbacher W.E., Goodman W., Smith S.L., Agui N., Granger N.A. and Sedlak B.J. (1980) Hormones controlling insect metamorphosis. Rec. Prog. Horm. Res. 36, 401-449.
- Gilbert L.I., Goodman W. and Bollenbacher W.E. (1977) Biochemistry of regulatory lipids and sterols in insects. In "Biochemistry of Lipids". Int. Rev. Biochem. 14, 1-50.
- Gilbert L.I., Rybczynski R. and Tobe S.B. (1996) Endocrine cascade in metamorphosis. In "Metamorphosis: Postembryonic Reprogramming of Gene Expression in Amphibian and Insect cells" (L.I. Gilbert, J.R. Tata and B.G. Atkinson Eds.) pp.59-107. Academic Press, San Diego.
- Girardie A. and De Reggi M.L. (1978). Moulting and ecdysone release in response to electrical stimulation of protocerebral neurosecretory cells in Locusta migratoria. J. Insect Physiol. 24, 797-802.
- Glitho I., Delbecque J.P. and Delachambre J. (1979) Prothoracic gland involution related to moulting hormone levels during the metamorphosis of *Tenebrio molitor L. J. Insect Physiol.* 25, 187-192.
- Goodwin T.W., Horn D.H.S., Karlson P., Koolman J., Nakanishi K., Robbins W.E., Siddall J.B. and Takemoto T. (1978) Ecdysteroids: a new generic term. Nature 272, 122.
- Granger N.A. and Bollenbacher W.E. (1981) Hormonal control of insect metamorphosis. In "Metamorphosis: A problem in Developmental Biology" 2nd ed. (Ed. by Gilbert L.I. and Frieden E.) pp.105-137. Plenum Press, New York.

- Gruetzmacher M.C., Gilbert L.I. and Bollenbacher W.E. (1984b) Indirect stimulation of the prothoracic glands of *Manduca sexta* by juvenile hormone: evidence for a fat body stimulatory factor. J. Insect Physiol. 30, 771-778.
- Gruetzmacher M.C., Gilbert L.I., Granger N.A., Goodman W. and Bollenbacher W.E. (1984a) The effect of juvenile hormone on prothoracic gland function during the larval-pupal development of Manduca sexta as in situ and in vitro analysis. J. Insect Physiol. 30, 331-340.
- Hammock B.D. (1985) Regulation of juvenile hormone titer: Degradation. In "Comprehensive Insect Physiology, Biochemistry and Physiology" (Ed. by Kerkut G.A. and Gilbert L.I.) Vol. 7, pp.431-472, Pergamon Press, Oxford.
- Hanaoka K. and Ohnishi E. (1974) Changes in ecdysone titre during pupaladult development in the silkworm, Bombyx mori. J. Insect Physiol. 20, 2375-2384.
- Hansen B.L., Hansen G.N. and Scharrer B. (1982) Immunoreactive material resembling vertebrate neuropeptides in the corpus cardiacum and corpus allatum of the insect *Leucophae maderae*. Cell Tiss. Res. 225, 319-329.
- Hatakoshi M., Agui N. and Nakayama I. (1986) 2-[1-methyl-2-(4-phenoxy phenoxy) ethoxyl pyridine as a new insect juvenile hormone analogue: Induction of supernumerary larvae in Spodoptera litura (Lepidoptera: Noctuidae). Appl. Ent. Zool. 21, 351-352.
- Hetru C. and Horn D.H.S. (1980) Phytoecdysteroids and zooecdysteroids. In "Progress in Ecdysone Research" (Ed. by Hoffmann J.A.) pp.13-28. Elsevier/North Holland Biomedical Press, Amsterdam.
- Hiruma K. (1980). Possible roles of juvenile hormone in the prepupal stage of Mamestra brassicae. Gen. Comp. Endocr. 41, 392-399.
- Hiruma K. (1986) Regulation of prothoracicotropic hormone release by juvenile hormone in the penultimate and last instar larvae of Mamestra brassicae. Gen. Comp. Endocrinol. 63 (1986) 201-211.
- Hiruma K., Shimada H. and Yagi S. (1978) Activation of prothoracic gland by juvenile hormone and prothoracicotropic hormone in *Mamestra* brassicae. J. Insect Physiol. 24, 215-220.

- Hodgetts R.B., Sage B.A. and O'Connor J.D. (1977) Ecdysone titres during post-embryonic development of *Drosophila melanogaster*. *Dev. Biol.* 60, 310-317.
- Hoffmann J.A., Lagauex M., Hetru C., Charlet M. and Goltzene F. (1980)
 Ecdysone in reproductively competent female adults and in embryos of insects. In "Progress in Ecdysone Research" (Ed. by Hoffmann J.A.)
 pp.431-466. Elsevier/ North-Holland Biomedical Press, Amsterdam.
- Hoffmann K.H. and Behrens W. (1982) Free ecdysteroids in adult male crickets, Gryllus bimaculatus. Physiol. Entromol. 7, 269-279.
- Hsiao T.H. and Hsiao C. (1977) Simultaneous determination of moulting and juvenile hormone titers of the greater wax moth. J. Insect Physiol. 23, 89-93.

Ishibashi J., Kataoka H., Isogai A., Kawakami A., Saegusa H., Yagi Y., Mizoguchi A., Shizaki H. and Suzuki A. (1994) Assignment of disulfide bond location in prothoracicotropic hormone of the silkworm, *Bombyx mori*: a homodimeric protein. *Biochem.* 33, 5912-5919.

- Ishizaki H. and Ichikawa M. (1967) Purification of the brain hormone of the silkworm *Bombyx mori. Biol. Bull.* 133, 355-368.
- Jagannadh V. and Nair V.S.K. (1992) Azadirachtin-induced effects on larvalpupal ransformation of Spodoptera mauritia. Physiol. Entomol. 17, 56-61.
- Jarvis T.D., Earley F.G.P. and Rees H.H. (1994) Ecdysteroid biosynthesis in the larval testes of Spodoptera littoralis. Insect Biochem. Mol. Biol. 24, 531-535.
- Jones D., Jones G. and Hammock B.D. (1981) Growth parameters associated with endocrine events in larval Trichoplusia ni (Hübner) and timing of these events with developmental markers. J. Insect Physiol. 27, 779-788.
- Jones G. and Hammock B.D. (1985). Critical roles for prepupal juvenile hormone and its esterase. Archs. Insect Biochem. Physiol. 2, 397-404.
- Jones G., Click A., Malleezewen V.R., Loeb M.J., Brandt E.P. and Woods C.W. (1986) Haemolymph ecdysteroid titre in larvae of the cabbage looper *Trichoplusia ni*, and its modification by juvenile hormone. J. Insect Physiol. 32, 561-566.
- Jones G., Hanzlik T., Hammock B.D., Schooley D.A., Miller C.A., Tsai L.W. and Baker F.C. (1990). The Juvenile Hormone titre during the penultimate and ultimate larval stadia of *Trichoplusia ni. J. Insect Physiol.* **36**, 77-83.

- Kallenborn H.G. and Mosbacher G.C. (1983) The ecdysteroid titres during the last-larval instar of *Ephestia kuhniella* Z. (Lepidoptera: Pyralidae). J. Insect Physiol. 29, 749-753.
- Kambysellis M.P. and Williams C.M. (1971b) In vitro development of insect tissues. II The role of ecdysone in the spermatogenesis of silkworms. Biol. Bull. 141, 541-552.
- Kambysellis M.P. and Williams C.M. (1972) Spermatogenesis in the cultured testes of the *cynthia* silk worm: effects of ecdysone and prothoracic glands. *Science*, 175, 769-770.
- Kaplanis J.N., Thompson M.J., Yamamoto R.T. and Louloudes S.J. (1966a) Ecdysones from the pupa of the tobacco hornworm, *Manduca sexta* (Johannson). Steroids 8, 605-623.
- Karlson P. (1956) Biochemical studies on insect hormones. Vit. Horm. 14, 227-266.
- Karlson P. (1980) Ecdysone in retrospect and prospect. In "Progress in Ecdysone Research" (Ed. by Hoffmann J.A.) pp.1-11, Elsevier, Amsterdam.
- Kawakami A., Kataoka H., Oka T., Mizoguchi A., Kimura-Kawakami M., Adachi T., Iwami M., Nagasawa H., Suzuki A. and Ishizaki H. (1990) Molecular cloning of the Bombyx mori prothoracicotropic hormone. Science 247, 1333-1335.
- Kawasaki H. (1995) Ecdysteroid concentration inducing cell proliferation brings about the imaginal differentiation in the wing disc of *Bombyx* mori in vitro. Deve., Growth Differ. 37, 575-580.
- Kelly T.J., Marler E.P., Thyagaraja B.S., Bell R.A. and Borkovec A.B. (1986) Prothoracicotropic hormone stimulation of ecdysone synthesis by the prothoracic glands of the gypsy moth, Lymantria dispar. In "Insect Neurochemistry and Neurophysiology" (Ed. by Borkovec A.B. and Gelman D.B.) pp.327-330. Humana Press, Clifton, NJ.
- Kelly T.J., Woods C.W., Redfern, R.E. and Borkovec A.B. (1981) Makisterone A. The molting hormone of larval Oncopeltus. J. Exp. Zool. 218, 127-132.

- Khan M.A., Koopmanschap A.B. and De Kort C.A.D. (1982) The effects of juvenile hormone, 20-hydroxyecdysone and precocene II on activity of corpora allata and the mode of negative feedback regulation of these glands in the adult Colorado potato beetle. J. Insect Physiol. 28, 995-1001.
- Kiguchi K. (1983) Time-table for the development of the silkworm, Bombyx mori JARC (Tropical, Agricultural Research Centre, Japan) 17, 42-46.
- Kiguchi K. and Agui N. (1981) Ecdysteroid levels and developmental events during larval moulting in the silkworm, *Bombyx mori. J. Insect Physiol.* 27, 805-812.
- Kiguchi K., Agui N., Kawasaki H. and Kobayashi K. (1985) Developmental time-table for the last larval and pharate pupal stages in the silkworm, *Bombyx mori*, with special reference to the correlation between the developmental events and hemolymph ecdysteroid levels. *Bull. Seric. Exp. Sta.*, 30, 83-100.
- King D.S. (1972) Metabolism of α-ecdysone and possible immediate precursors by insects in vivo and in vitro. Gen. Comp. Endocr. Suppl. 3, 221-227.
- King D.S., Bollenbacher W.E., Borst D.E., Vedeckis N.V., O'Connor J.D., Ittycheriah P.I. and Gilbert L.I. (1974) The secretion of α-ecdysone by the prothoracic glands of Manduca sexta in vitro. Proc. Natn. Acad. Sci. 71, 793-796.
- Kiuchi M. (1990) Effects of an insect growth regulator, RH 5849, on the larval molt and larval-pupal metamorphosis of the silkworm, Bombyx mori. Abstract Seventh International Congress of Pesticide Chemistry, Hamburg, August 1990.
- Koeppe J.K., Fuchs M., Chen ,T.T., Hunt L.M., Kovalick G.E. and Briers T. (1985) "The role of juvenile hormone in Reproduction". In "Comprehensive Insect Physiology, Biochemistry and Pharmacology". (Ed. by Kerkut G.A. and Gilbert L.I.). Vol. 8, pp.165-203, Pergamon Press, New York.
- Koolman J. (1978) Ecdysone oxidase in insects. Hoppe-Seyler's Z. Physiol. Chem. 359, 1315-1321.
- Koolman J., Scheller K. and Bodenstein D. (1979). Ecdysteroids in the adult male blowfly Calliphora vicina. Experientia, 35, 134-135.

- Kopec S. (1917) Experiments on metamorphosis of insects. Boll. Int. Acad. Cracovie B. 57-60.
- Kopec S. (1922) Studies on the necessity of the brain for the inception of insect metamorphosis. *Biol. Bull.* 42, 322-342.
- Kubo I., Klocke A. and Asano S. (1983) Effects of ingested phytoecdysteroids on the growth and development of two lepidopterous larvae. J. Insect Physiol. 29, 307-313.
- Lafont R., Mauchamp B., Blais C. and Pennetier J.L. (1977b) Ecdysones and imaginal disc development during the last larval instar of *Pieris* brassicae. J. Insect Physiol. 23, 277-283.
- Lafont R., Mouchamp B., Boulay G. and Tarroux P. (1975) Developmental studies in *Pieris brassicae* (Lepidoptera) – I. Growth of various tissues during the last larval instar. *Comp. Biochem. Physiol.* **51B**, 439-444.
- Lagaeux M., Perron J.-M. and Hoffmann J.A. (1976) Ecdysone metabolism and endogenous moulting hormone titer during larval-pupal development in *Choristoneura fumiferana*. J. Insect Physiol. 22, 57-62.
- Lanzrein B. (1975) Programming, induction, or prevention of the breakdown of the prothoracic gland in the cockroach, Naphoeta cinera. J. Insect Physiol. 21, 367-389.
- Leclerq-Smekens M. (1974) Organogenese et differenciation des voies genitales males internes des lepidopteres sur la base d'observations effectuees chez *Euproctis chrysorrhea* (Lymantriidae). Ann. Soc. r. Zool. Belg. 104, 131-154.
- Leviatan R. and Friedlander M. (1979) Eupyrene-apyrene dichtomous spermatogenesis of Lepidoptera. Dev. Biol. 68, 515-524.
- Loeb M. (1982) Diapause and development in the tobacco budworm, Heliothis virescens: A comparison of haemolyph ecdysteroid titres. J. Insect Physiol. 28, 667-673.
- Loeb M.J. (1986) Ecdysteroids in Testis sheaths of *Heliothis virescens* larvae: An immunocytological study. Arch. Insect Biochem. Physiol., 3, 173-180.
- Loeb M.J. (1991b) Effect of calcium ions on ecdysteroid secretion by testes of *Heliothis virescens*. Comp. Biochem. Physiol. [B] 100, 303-308.
- Loeb M.J. (1994). Characterization of genital tract growth factor like activity from testis sheath and fat body of *Heliothis virescens*. Arch. Insect Biochem. Physiol. 26, 263-277.

- Loeb M.J. and Birnbaum M.J. (1981) The relationship of hemolymph osmotic pressure to spermatogenesis in the tobacco budworm, *Heliothis virescens. Int. J. Invert. Reprod.* 4, 67-79.
- Loeb M.J. and Hakim R.S. (1991) Development of genital imaginal discs of *Heliothis virescens* cultured *in vitro* with 20-hydroxyecdysone and fat body or testis sheaths. *Invert. Reprod. Dev.* 20, 181-191.
- Loeb M.J. and Woods C.W. (1989) Metabolism of ecdysteroids by testes of the Tobacco budworm, *Heliothis virescens Arch. Insect Biochem. Biophys.* 10, 83-92.
- Loeb M.J., Brandt E.P. and Birnbaum M.J. (1984) Ecdysteroid production by the testes of Tobacco budworm, *Heliothis virescens*, from last larval instar to adult. J. Insect Physiol. 30, 375-381.
- Loeb M.J., Brandt E.P., Giebultowicz J.M. and Woods C.W. (1986) Endocrine regulation of the testes and of spermatogenesis in insects. In "Advances in Invertebrate Reproduction". (Ed. by Andries J.C. and Dhainaut A.) Vol.4, pp.93-100. Elsevier, Amsterdam.
- Loeb M.J., Brandt E.P., Woods C.W. and Bell R.A. (1988) Secretion of ecdysteroid by sheaths of testes of the Gypsy moth, Lymantria dispar and its regulation by testes ecdysiotropin. J. exp. Zool. 248, 94-100.
- Loeb M.J., Brandt E.P., Woods C.W. and Borkovec A.B. (1987) An ecdysiotropic factor from brain of *Heliothis virescens* induces testes to produce immunodetectable ecdysteroid *in vitro*. J. Exp. Zool. 243, 275-282.
- Loeb M.J., Wagner R.M., Gelman D.B., Kochansky J.P., Lusby W.R., Woods C.W. and Rao K.R. (1994b) Testis ecdysiotropin: sequence and mode of action. In "Insect Neurobiology and Neurophysiology 1993" (Ed. by Borkovec A.B. and Loeb M.J.) pp. 277-280. Bloca Raton, FL: CRC Press.
- Loeb M.J., Wagner R.M., Woods C.W. Gelman D.B., Harrison D., Bell R.A. (1997) Naturally occurring analogues of Lymantria testis ecdysiotropin, a gonadotropin isolated from brains of Lymantria dispar pupae. Arch. Insect Biochem. Physiol. 36, 37-50.
- Loeb M.J., Woods C.W., Brandt E.P. and Borkovec A.B. (1982) Larval testes of tobacco budworm. A new source of insect ecdysteroids. *Science*. 218, 896-898.
- Mala J. and Sehnal F. (1978) Role of the nerve cord in the control of prothoracic glands in Galleria mellonella. Experientia 34, 1233-1234.

- Mala J., Granger N.A. and Sehnal F. (1977) Control of prothoracic gland activity in larvae of Galleria mellonella. J. Insect Physiol. 23, 309-316.
- Marks E.P., Ittycheriah P.I. and Leloup A.M. (1972) The effect of β -ecdysone on insect neurosecretion in vivo. J. Insect Physiol. 18, 847-850.
- Maroy P. and Tarnoy K. (1978) Moulting hormone level during the last instar of Galleria mellonella larva. J. Insect Physiol. 24, 325-327.
- Masner P., Hangartner W. and Suchy M. (1975) Reduced titers of ecdysone following juvenile hormone treatment in the German cockroach, Blatella germanica. J. Insect Physiol. 21, 1755-1762.
- Meola S.M. and Loeb M.J. (1995) Unique intertesticular tissue complex in larvae of Heliothis virescens (F) (Lepidoptera: Noctuidae). Int. J. Insect Morphol. Embryol. 24, 443-457.
- Meyer A.S., Schneiderman H.A., Hanzymann E. and Ko J.H. (1968) Two juvenile hormones from the cecropia silkmoth. *Proc. Nat. Acad. Sci.* USA 60, 853-860.
- Mikitani K. (1996) Ecdysteroid receptor binding activity and ecdysteroid agonist activity at the level of gene expression are correlated with the activity of dibenzoyl hydrazines in larvae of *Bombyx mori*. J. Insect Physiol. 42, 937-941.
- Mitsui T. and Riddiford L.M. (1978) Hormonal requirements for the larvalpupal transformation of the epidermis of *Manduca sexta in vitro*. *Devel. Biol.* 62, 193-205.
- Mizoguchi A. (1995) Measurement of the titers of Bombyx prothoracicotropic hormone and bombyxin in the hemolymph of Bombyx mori by timeresolved fluorimmunoassay. In "Molecular Mechanisms of Insect Metamorphosis and Diapause" (Ed. by Suzuki A., Kataoka H. and Matsumoto S.) pp.129-137. Industrial Publishing and Consulting Inc., Tokyo.
- Mizoguchi A. and Ishizaki H. (1982) Prothoracic glands of the saturniid moth Samia cynthia ricini possess a circadian clock controlling gut purge timing. Proc. Natn. Acad. Sci., 79, 2716-2730.
- Mizoguchi A., Oka T., Kataoka H., Nagasawa H., Suzuki A. and Ishizaki H. (1990) Immunohistochemical localization of prothoracicotropic hormone-producing cells in the brain of Bombyx mori. Dev. Growth Differ. 32, 591-598.

- Monthean C. and Potter D.A. (1992) Effects of RH 5849, a novel insect growth regulator, on Japanese beetle (Coleoptera: Scarabaeidae) and fall armyworm (Lepidoptera: Noctuidae) in turfgrass. J. Econ. Entomol. 85, 507.
- Nair V.S.K. (1980) Moult inhibition in the insect Dysdercus cingulatus (Insecta: Heteroptera) by the cerebral glands of the millipede Jonespeltis splendidus (Myriapoda:Diplopoda). Experientia 36, 607-608.
- Nair V.S.K. (1981) Hormonal control on the mating behaviour in the females of Spodoptera mauritia (Lepidoptera: Noctuidae). Curr. Sci. 50, 690-691.
- Nair V.S.K. (1993) Applied potential of insect hormone research: perspectives and prospects. In "Chemical Ecology of Phytophagous Insects" (Ed. Ananthakrishanan T.N. and Raman A. pp.89-104. Oxford - IBH Publ. Co., New Delhi.
- Nair V.S.K. and Rajalekshmi E. (1989) Effects of fluoromevalonate on penultimate and last instar larvae of Spodoptera mauritia Boisd. Indian J. Exp. Biol. 27, 170-173.
- Nakanishi K., Koreeda M., Sasaki S., Chang M.I. and Hsu H.Y. (1966) Insect Hormones: The structure of panosterone A, an insect moulting hormone from the leaves of Podocarpus nakaii Hay. Chem. Commun. 915-917.
- Nijhout H.F. (1975) Dynamics of juvenile hormone action in the larvae of the tobacco hornworm *Manduca sexta*. *Biol. Bull.* **149**, 568-579.
- Nijhout H.F. (1979) Stretch-induced molting in Oncopeltus fasciatus. J. Insect Physiol. 25, 277-281.
- Nijhout H.F. and Williams C.M. (1974) Control of moulting and metamorphosis in the tobacco hornworm, *Mandua sexta*: cessation of juvenile hormone secretion as a trigger for pupation. J. Exp. Biol. 61, 493-501.
- Nishiitsutsuji-Uwo J. (1961) Further studies on the developmental features of the male germ cells in the prospective diapausing and non-diapausing swallow tails, *Papilio xuthus*. Annotnes Zoo. Jap. 34, 30-37.
- Nowock J. (1972). Induction of imaginal differentiation by ecdysone in the testes of *Ephestia kuhniella J. Insect Physiol.* 18, 1699-1704.

Nowock J. (1973). Growth and metamorphosis in the testes of *Ephestia* Kuhniella in vitro. J. Insect Physiol. 19, 941-949.

- Okuda M., Sakurai S. and Ohtaki T. (1985) Activity of the prothoracic gland and its sensitivity to prothoracicotropic hormone in the penultimate and last-larval instar of *Bombyx mori*. J. Insect Physiol. **31**, 455-461.
- Palli S.R., Primavera M., Tomkins W., Lambert D. and Retnakaran A. (1995)
 Age specific effects of a nonsteroidal ecdysone agonist, RH-5992, on the spruce budworm Choristoneura fumiferana (Lepidoptera: Tortricidae)
 Eur. J. Entomol., 92, 325-332.
- Pradeep A.R. and Nair V.S.K. (1989) Morphogenetic effects of precocene II in the brown planthopper *Nilaparvata lugens* Stäl. (Homoptera: Delphacidae). *Proc. Indian natn. Sci. Acad.* **B55**, 381-386.
- Pratt G.E., Tobe S.S., Weaver R.J. and Finney J.R. (1975a) Spontaneous synthesis and release of C₁₆ juvenile hormone by isolated corpora allata of female locust Schistocerca gregaria and female cockroach Periplaneta americana. Gen. Comp. Endocr. 26, 478-484.
- Retnakaran A., Granett, J. and Ennis, T. (1985). Insect Growth Regulators. In "Comprehensive Insect Physiology, Biochemistry and Pharmacology" (Ed. by Kerkut G.A. and Gilbert L.I.) Vol.12, pp.529-601. Pergamon Press, Oxford.
- Retnakaran A., Hiruma K., Palli S.R. and Riddiford L.M. (1995) Molecular analysis of the mode of action of RH-5992, a lepidopteran-specific, non-steroidal ecdysteroid agonist. *Insect Biochem. Mol. Biol.* 25, 109-117.
- Retnakaran A., Mac Donald A., Tomkins W.L., Davis C.N., Brownright A.J. and Palli S.R. (1997) Ultrastructural effects of a non-steroidal ecdysone agonist, RH-5992, on the sixth instar larva of the spruce budworm, *Choristoneura fumiferana*. J. Insect Physiol. 43, 55-68.
- Richards G. (1981) The radioimmune assay of ecdysteroid titres in Drosophila melanogaster. Molec. Cell. Endocr. 21, 181-197.
- Riddiford L.M. (1976) Juvenile hormone control of epidermal commitment in vivo and in vitro. In "The Juvenile Hormones" (Ed. by Gilbert L.I.) pp.198-219. Plenum Press, New York.
- Riddiford L.M. (1978) Ecdysone-induced change in cellular commitment of the epidermis of the tobacco hornworm, *Manduca sexta*, at the initiation of metamorphosis. *Gen. Comp. Endocr.* 34, 438-446.
- Riddiford L.M. (1980) Interaction of ecdysteroids and juvenile hormone in the regulation of larval growth and metamorphosis

- Riddiford L.M. (1981) Hormonal control of epidermal cell development Amer. Zool. 21, 751-762.
- Riddiford L.M. (1985) Hormone action at cellular level. In "Comprehensive Insect Physiology, Biochemistry and Pharmacology" (Ed. by Kerkut G.A. and Gilbert L.I.) Vol. 8, pp.37-84. Pergamon Press, Oxford.
- Riddiford L.M. (1994) Cellular and molecular actions of juvenile hormone I. General considerations and premetamorphic actions. Adv. Insect Physiol. 24, 213-274.
- Riddiford L.M. (1996) Juvenile hormone: the status of its "status quo" action. Arch. Insect Biochem. Physiol. 32, 271-286.
- Riddiford L.M. and Truman J.W. (1978) Biochemistry of insect hormones and insect growth regulators. In "Biochemistry of Insects" (Ed. by Rockstein M.) pp.307-357. Academic Press, New York.
- Röller H., Dahm K.H., Sweeley C.C. and Trost B.M. (1967) The structure of the juvenile hormone. Angew. Chem. Int. Edit. 6, 179-180.
- Rountree D.B. and Bollenbacher W.E. (1986) The release of the prothoracicotropic hormone in the tobacco hornworm *Manduca sexta* is controlled intrinsically by juvenile hormone. J. Exp. Biol. 120, 41-58.
- Rountree D.B., Combest W.L. and Gilbert L.I. (1987) Protein phosphorylation in the prothoracic glands as a cellular model for juvenile hormone prothoracicotropic hormone interactions. *Insect Biochem.* 17, 943-948.
- Safranek L., Cymborowski B. and Williams C.M. (1980) Effects of juvenile hormone on ecdysone-dependent development in the tobacco hornworm, *Manduca sexta. Biol. Bull.* 158, 248-256.
- Sakunthala C. and Nair V.S.K. (1995) Effects of ecdysone agonist RH 5849 on embryonic and post embryonic development of Spodoptera mauritia Boisd. (Lepidoptera: Noctuidae) Indian. J. Exp. Biol. 33, 690-693.
- Sakurai S. (1983) Temporal organization of endocrine events underlying larval-larval ecdysis in the silkworm Bombyx mori. J. Insect Physiol. 29, 919-932.
- Sakurai S. (1984) Temporal organization of endocrine events underlying larval-pupal metamorphosis in the silkworm, Bombyx mori. J. Insect Physiol. 30, 657-664.
- Sakurai S. and Gilbert L.I. (1990) Biosynthesis and secretion of ecdysteroids by the prothoracic glands. In "Molting and Metamorphosis" (Ed. by

Ohnishi E. and Ishizaki H.) pp. 83-106. Japan Science Society Press, Tokyo.

- Sakurai S. and Williams C.M. (1989) Short-loop negative and positive feedback on ecdysone secretion by prothoracic glands in the tobacco hornworm, *Manduca sexta. Gen. Comp. Endocr.* **75**, 204-216.
- Sakurai S., Kaya M. and Satake S.I. (1998) Hemolymph ecdysteroid titer and ecdysteroid-dependent developmental events in the last-larval stadium of the silkworm, *Bombyx mori*: role of low ecdysteroid titer in larvalpupal metamorphosis and a reappraisal of the head critical period. J. Insect Physiol. 44, 867-881.
- Sakurai S., Okuda M. and Ohtaki T. (1989) Juvenile hormone inhibits ecdysone secretion and responsiveness to prothoracicotropic hormone in prothoracic glands of *Bombyx mori. Gen. Comp. Endocr.* **75**, 222-230.
- Sam Mathai and Nair V.S.K. (1983) Effects of precocene II on last instar larvae of Spodoptera mauritia (Lepidoptera: Noctuidae). Curr. Sci. 52, 376-377.
- Sam Mathai and Nair V.S.K. (1984a). Treatment of larvae with repeated doses of precocene II induces precocious metamorphic changes in Spodoptera mauritia (Lepidoptera: Noctuidae) Arch. Insect. Biochem. Physiol. 1, 199-203.
- Sam Mathai and Nair V.S.K. (1984b) Effects of precocene II on ovarian follicle development and vitellogenesis in Spodoptera mauritia Boisd. (Lepidoptera: Noctuidae) abst. XVII. Internt. Cong. Entomol p.21.
- Sam Mathai and Nair V.S.K. (1984c) Morphogenetic effects of precocene II on the larval stages of Spodoptera mauritia Boisd. (Lepidoptera: Noctuidae). In Proceedings of Fifth All India Symposium on Developmental Biology. (Ed. by Aggarwal S.K.and Goel S.C.) pp.59-66.
- Sam Mathai and Nair V.S.K. (1984d) Effects of a juvenile hormone analogue on acid and alkaline phosphatase activities in the pupae of *Spodoptera mauritia* (Lepidoptera: Noctuidae). Comp. Physiol. Ecol. 9, 253-256.
- Sam Mathai, Santha P.C. and Nair V.S.K. (1989) Antigonadotropic effects of precocene II in Spodoptera mauritia Boisd. (Lepidoptera: Noctuidae). Proc. Indian natn. Sci. Acad. B55, 91-96.
- Santha P.C. and Nair V.S.K. (1986) Age-dependent responses of last instar larvae of Spodoptera mauritia to precocene II. Physiol-Entomol. 11, 335-341.

- Santha P.C. and Nair V.S.K. (1987) Larval-pupal transformation of *Spodoptera mauritia* (Lepidoptera: Noctuidae). Effects of a juvenile hormone analogue *Zool. Anz.* **218**, 379-384.
- Santha P.C. and Nair V.S.K. (1988) Precocene induced abnormalities in F1 generation of Spodoptera mauritia Zool. Anz. 220, 104-110.
- Santha P.C. and Nair V.S.K. (1991) Precocene-induced changes in the cerebral neurosecretory cells of *Spodoptera mauritia* Boisd. (Lepidoptera: Noctuidae) *Zool. Anz.* 226, 163-173.
- Santha P.C., Sankaranarayanan P. and Nair V.S.K. (1987) Precoceneinduced effects on pupal-adult development of *Spodoptera mauritia* Boisd. (Lepidoptera: Noctuidae) *Geobios New Reports* 6, 35-38.
- Sbrenna G., Leis M., Bonora M. and Sbrenna- Micciarelli A. (1988) Hormonal control of metamorphosis in the hemimetabolous insect *Schistocerca gregaria* (Orthoptera: Acrididae). In "Proceedings of the 15th Italian National Entomology Congress, L. Aquela, 747-754.
- Schmidt E.L. and Williams C.M. (1953) Physiology of insect diapause. V. Assays of the growth and differentiation hormone of lepidoptera by the method of tissue culture. Biol. Bull. Mar. Biol. Lab. Woods Hole. 105, 174-187.
- Schooley D.A., Judy K.I., Bergot B.I., Hall M.S. and Jennings R.C. (1976) Determination of the physiological levels of juvenile hormones in several insects and biosynthesis of the carbon skeletons of the juvenile hormones. In "The Juvenile Hormones" (Ed. by Gilbert L.I.) pp.101-117. Plenum Press, New York.
- Schooneveld H., Kramer S.J., Privee H. and Van Huis A. (1979) Evidence of controlled corpus allatum activity in the adult Colorado potato beetle. J. Insect Physiol. 25, 449-453.
- Scott D. (1982) Communication to Insect Hormone Discussion Group, Workshop 1982. Royal Entomological Society, London cited in "Comprehensive Insect Physiology, Biochemistry and Pharmacology" (Ed. by Kerkut G.A. and Gilbert L.I.) Vol. 7, p.319. Pergamon Press, Oxford.
- Sedlak B.J., Marchione L., Devokin B. and Davino R. (1983) Correlations between endocrine gland ultrastructure and hormone titers in the fifth larval instar of *Manduca sexta*. *Gen. Comp. Endocr.* **52**, 291-310.
- Sehnal F. (1983) Juvenile hormone analogues. In "Endocrinology of Insects" (Ed. by Downer R.G.H. and Laufer H.) pp.657-672. Alan R. Liss, New York.

- Sehnal F. (1985) Growth and life cycle. In "Comprehensive Insect Physiology, Biochemistry and Pharmacology" (Ed. by Kerkut G.A. and Gilbert L.I.), Vol. 2, pp.1-86, Pergamon, Oxford.
- Sehnal F., Delbecque J.P., Maroy P. and Mala J. (1986). Ecdysteroid titers during larval life and metamorphosis of *Galleria mellonella*. Insect Biochem. 16, 157-162.
- Sehnal F., Maroy P. and Mala J. (1981) Regulation and significance of ecdysteroid titre fluctuation in lepidopterous larvae and pupae. J. Insect Physiol. 27, 535-544.
- Sehnal F., Metwally M.M. and Gelbic I. (1976) Reactions of immature stages of noctuid moths to juvenoids. Z. ang. Ent. 81, 85-102.
- Shaaya E. and Karlson P. (1965a) Der ecdysontiter während der insektenentwicklung II. Die postembryonale entwicklung der schmeissfliege Calliphora erythrocephala Meig. J. Insect Physiol. 11, 65-69.
- Shaaya E. and Karlson P. (1965b) Der ecdysontiter während der insektentwicklung IV. Die entwicklung der lepidopteren Bombyx mori L. and Cerura vinula L. Devel. Biol. 11, 424-432.
- Shimizu T., Moribayashi A. and Agui N. (1985). Biosynthesis and metabolism of ecdysteroids and methods of isolation and identification of the free and conjugated compounds. *Meth. Enzymol.* III B, 377-410.
- Shimizu T., Yagi S. and Agui N. (1989) The relationship of testicular and haemolymph ecdysteroid titer to spermiogenesis in the common armyworm, *Leucania separata*. Entomol. Exp. Appl. 50, 195-198.
- Shinyaeva L.I. (1981) Effect of juvenile hormone analogues on spermatogenesis in *Eurygaster intergricepts* dring the period of prediapause and diapause formation. *Zool. Zh.* **60**, 1198-1204.
- Shirai Y., Aizono Y., Iwasaki T., Yanagida A., Mori H., Sumida M. and Matsubara F. (1993) Prothoracicotropic hormone is released five times in 5th larval instar of the silkworm, Bombyx mori. J. Insect Physiol. 39, 83-88.
- Sieber R. and Benz G. (1977) Juvenile hormone in larval diapause of the codling moth, Laspeyresia pomonella L. (Lepidoptera: Tortricidae). Experientia 33, 1598-1599.

- Sieber R. and Benz G. (1980) The hormonal regulation of the larval diapause in the codling moth, *Laspeyresia pomonella* (Lepidoptera: Tortricidae). J. Insect Physiol. 26, 213-218.
- Siew Y.C. and Gilbert L.I. (1971) Effects of moulting hormone and juvenile hormone on insect endocrine gland activity. J. Insect Physiol. 17, 2095-2104.
- Silhacek D.L., Oberlander H. and Porcheron P. (1990) Action of RH 5849, a non-steroidal ecdysteoid mimic on *Plodia interpunctella* (Hübner) in vivo and in vitro. Arch. Insect Biochem. Physiol. 15, 201.
- Singh H.H. and Sehnal F. (1979) Lack of specific neurones in the ventral nerve cord for the control of prothoracic glands. *Experientia* 35, 1117-1119.
- Slama K. (1971) Insect juvenile hormone analogues. Annu. Rev. Biochem. 49, 1079-1102.
- Slama K. and Mala J. (1984) Growth and respiratory enzyme activity in the prothoracic glands of *Galleria*. J. Insect Physiol. 30, 877-882.
- Slama K., Ramanuk M. and Sorm F. (1974) Insect Hormones and Bioanalogues. Berlin/Heidelberg. New York: Springer.
- Smagghe G. and Degheele D. (1992) Effects of RH 5849, the first nonsteroidal ecdysteroid agonist, on larvae of *Spodoptera littoralis* (Boisd.) (Lepidoptera: Noctuidae) Arch. Insect Biochem. Physiol. 21, 119.
- Smagghe G. and Degheele D. (1993) Toxicity, pharmacokinetics and metabolism of the first nonsteroidal ecdysteroid agonist RH 5849, on Spodoptera exempta (Walker), Spodoptera exigua (Hübner), and Leptionotarsa decemlineata (Say). Pestic. Biochem. Physiol. 46, 149-160.
- Smagghe G. and Degheele D. (1994a) Action of the nonsteroidal ecdysteroid mimic RH 5849 on larval development and adult reproduction of insects of different orders. *Invert. Reprod. Dev.* 25, 227-236.
- Smagghe G. and Degheele D. (1994b) Action of a novel nonsteroidal ecdysteroid mimic, tebufenozide (RH 5992), on insects of different orders. *Pestic. Sci.* 42, 85-92.

- Smagghe G., Degheele D., Vinuela E. and Budia F. (1994) Potency and ultrastructural effects of tebufenozide on Spodoptera exigua. Proc. BCPC – Pests and Diseases, 339-340.
- Smagghe G. and Degheele D. (1997) Comparative toxicity and tolerance for the ecdysteroid mimic tebufenozide in a laboratory and field strain of cotton leafworm (Lepidoptera: Noctuidae). J. Econ. Entomol. 90, 278-282.
- Smith S.L. (1985) Regulation of ecdysteroid titer: synthesis. In "Comprehensive Insect Physiology, Biochemistry and Pharmacology" Vol. 7 (Ed. by Kerkut G.A. and Gilbert L.I.) pp.295-341. Pergamon Press, New York.
- Smith S.L., Bollenbacher W.E. and Gilbert L.I. (1980) Studies on the biosynthesis of ecdysone and 20-hydroxyecdysone in the tobacco hornworm, *Manduca sexta*. In "Progress in Ecdysone Research" (Ed. by Hoffmann J.A.) pp.139-162, Elsevier, Amsterdam.
- Smith W.A. and Nijhout H.F. (1983) In vitro stimulation of cell death in the moulting glands on Oncopeltus fasciatus by 20-hydroxyecdysone. J. Insect Physiol. 29, 169-176.
- Staal G.B. (1986). Antijuvenile hormone agents. Ann. Rev. Entomol. 31, 391-429.
- Steel C.G.H. (1975) A neuroendocrine feedback mechanism in the insect moulting cycle. *Nature* 253, 267-269.
- Steel C.G.H. (1978) Some functions of identified neurosecretory cells in the brain of the aphid, *Megoura viciae. Gen. Comp. Endocr.* 34, 219-228.
- Steel C.G.H. and Davey K.G. (1985) Integration in the insect endocrine system. In "Comprehensive Insect Physiology, Bicohemistry and Pharmacology" (Ed. by Kerkut G.A. and Gilbert L.I.) Vol. 8, pp.1-35. Pergamon Press, Oxford.
- Steel C.G.H., Bollenbacher W.E., Smith S.L. and Gilbert L.I. (1982) Haemolymph ecdysteroid titres during larval-adult development in *Rhodnius prolixus*: Correlations with moulting hormone action and brain neurosecretory cell activity. J. Insect Physiol. 28, 519-525.
- Szibbo C.M. and Tobe S.S. (1981a) Cellular and volumetric changes in relation to the activity cycle in the corpora allata of *Diploptera* punctata. J. Insect Physiol. 27, 655-665.
- Szollozi A. (1982). Relationship between germ and somatic cells in the testes of locusts and moths. In: "Insect Ultrastructure" (Ed. by King R.C. Akai H.) pp.32-60. Plenum, New York.

- Szollozi A., Riemann J. and Marcaillou C. (1980) Localization of the blood testis barrier in the testis of the moth, Anagasta kuehniella. J. Ultrastruct. Res. 72, 189.
- Takeda N. (1972) Effect of ecdysone on spermatogenesis in the diapausing slug moth pharate pupa Monema flavescens. J. Insect Physiol. 571-580.
- Takeda N. (1978) Hormonal control of prepupal diapause in Monema flavescens (Lepidoptera). Gen. Comp. Endocr. 34, 123-131.
- Takeuchi S. (1969). Endocrinological studies of spermatogenesis in the silk worm Bombyx mori. Dev. Growth. Differ. 11, 8-28.
- Talbot W.S., Swyrd E.A. and Hogness D.S. (1993) Drosophila tissues with different metamorphic responses to ecdysone express different ecdysone receptor isoforms. Cell 73, 1323-1337.
- Tauber M.J. and Tauber C.A. (1976) Insect seasonality: diapause maintenance, termination and post-diapause development. Ann. Devel. Ent. 21, 81-107.
- Thorson B.J. and Riemann J.G. (1982). Effects of 20-hydroxy ecdysone on sperm release from the testes of the mediterranean flour moth Anagasta kuehniella (Zeller) J. Insect Physiol. 28, 1013-1019.
- Tobe S.S. and Pratt G.E. (1974b) Dependence of juvenile hormone release from corpus allatum on intraglandular content. *Nature* 252, 474-476.
- Tobe S.S. and Stay B. (1977) Corpus allatum activity in vitro during the reproductive cycle of the viviparous cockroach, *Diploptera punctata* (Eschsholtz). Gen. Comp. Endocr. **31**, 138-147.
- Tobe S.S. and Stay B. (1980) Control of juvenile hormone biosynthesis during the reproductive cycle of a viviparous cockroach III. Effects of denervation and age on compensation with unilateral allatectomy and supernumerary corpora allata. *Gen. Comp. Endocr.* 40, 89-98.
- Tobe S.S. and Stay B. (1985) Structure and regulation of the corpus allatum. Adv. Insect Physiol. 18, 305-432.
- Truman J.W. (1971) Physiology of insect ecdysis-I. The eclosion behaviour of saturniid moths and its hormone release. J. Exp. Biol. 54, 805.
- Truman J.W. (1972) Physiology of insect rhythms. I. Circadian organization of the endocrine events underlying the moulting cycle of larval tobacco hornworms. J. Exp. Biol. 57, 805-820.

- Truman J.W. and Riddiford L.M. (1974) Physiology of insect rhythms. III. The temporal organization of the endocrine events underlying pupation of the tobacco hornworm. J. Exp. Biol. 60, 371-382.
- Vafopoulou X. and Steel C.G. (1996) The insect neuropeptide prothoracicotropic hormone is released with a daily rhythm: reevaluation of its role in development. *Proc. Natn. Acad. Sci. USA.* 93, 3368-3372.
- Varjas L., Paquia P. and De Wilde J. (1976) Juvenile hormone titers in penultimate and last instar larvae of *Pieris brassicae* and *Barathra brassicae*, in relation to the effect of juvenoid application. *Experientia* 32, 249-251.
- Vedeckis W.V., Bollenbacher W.E. and Gilbert L.I. (1976) Insect prothoracic glands: a role for cyclic AMP in the stimulation of α -ecdysone secretion. *Molec. Cell Endocr.* 5, 81-88.
- Venugopalan A.K. (1995). Effects of hormones, hormone analogues and antihormones on the spermatogenesis of Spodoptera mauritia Boisd. (Lepidoptera: Noctuidae). Ph.D. Thesis, University of Calicut.
- Venugopalan A.K., Benny T.M. and Nair V.S.K. (1994) Testicular development and spermatogenesis in Spodoptera mauritia Boisd. (Lepidoptera: Noctuidae) J. Anim. Morphol. Physiol. 41, 125-134.
- Wagner R.M., Loeb M.J., Kochansky J.P., Gelman D.B., Lusby N.R. and Bell R.A. (1997) Identification and characterization of an ecdysiotropic peptide from brain extracts of the gypsy moth Lymantria dispar. Arch. Insect Biochem. Physiol. 34, 175-189.
- Warren J. and Hetru C. (1990) Ecdysone biosynthesis: pathways, enzymes and the early steps problem. *Invert. Reprod. Devel.* 18: 1-2, 91-99.
- Watson R.D. and Bollenbacher W.E. (1988). Juvenile hormone regulates the steroidogenic competence of Manduca sexta prothoracic glands. Molec. Cell. Endocrinol. 57, 251-259.
- Watson R.D., Agui N., Haire M.E. and Bollenbacher W.E. (1987b) Juvenile hormone co-ordinates the regulation of the haemolymph ecdysteroid titer during pupal commitment in *Manduca sexta*. Insect Biochem. 17, 955-959.
- Watson R.D., Williams T.K. and Bollenbacher W.E. (1987a) Regulation of ecdysone biosynthesis in the tobacco hornworm, *Manduca sexta*: titre of

the haemolymph stimulatory factor during the last larval instar. J. Exp. Biol. 128, 159-173.

- Wigglesworth V.B. (1933) The physiology of the cuticle and of ecdysis in *Rhodnius prolixus* (Triatomidae, Hemiptera); with special reference to the function of the oenocytes and of dermal glands. *Quart. J. Mic. Sci.* 76, 296-318.
- Wigglesworth V.B. (1934) The physiology of ecdysis in *Rhodnius prolixus* (Hemiptera) II. Factors controlling moulting and metamorphosis. *Quart. J. micr. Sci.* 77, 101-222.
- Wigglesworth V.B. (1936). The function of the corpus allatum in the growth and reproduction of *Rhodnius prolixus* (Hemiptera). Quart. J. Micr. Sci. 79, 91-121.
- Wigglesworth V.B. (1940) The determination of characters at metamorphosis in *Rhodnius prolixus* (Hemiptera). J. Exp. Biol. 17, 201-222.
- Wigglesworth V.B. (1955). The breakdown of the thoracic gland in the adult insect, *Rhodnius prolixus J. Exp. Biol.* 32, 485-491.
- Williams C.M. (1948) Physiology of insect diapause. III. The prothoracic glands in the *Cecropia* silkworm with special reference to their significance in embryonic development. *Biol. Bull.* 94, 60-65.
- Williams C.M. (1952) Physiology of insect diapause. IV. The brain and prothoracic glands as an endocrine system in the *Cecropia* silkworm. *Biol. Bull.* 103, 120-138.
- Williams C.M. (1959) The juvenile hormone. I. Endocrine activity of the corpora allata of the adult *Cecropia* silkworm. *Biol. Bull.* 116, 323-338.
- Williams C.M. (1961) The juvenile hormone II. Its role in the endocrine control of moulting, pupation and adult development in the *Cecropia* silkworm. *Biol. Bull.* 121, 572-585.
- Williams C.M. (1967) The third generation pesticides. Sci. Amer. 217, 13-17.
- Wing K.D. (1988) RH-5849, a nonsteroidal ecdysone agonist; Effects on a *Drosophila* cell line *Science*, 241, 467-469.
- Wing K.D., Slawecki R.A. and Carlson G.R. (1988) RH-5849, a non-steroidal ecdysone agonist: effects on larval Lepidoptera. *Science*, 241, 470-472.

- Wolfgang W.J. and Riddiford L.M. (1986) Larval cuticular morphogenesis in the tobacco hornworm, *Manduca sexta*, and its hormonal regulation. *Dev. Biol.* 113, 305-316.
- Yagi S. (1976) The role of juvenile hormone in diapause and phase variation in some lepidopterous insects. In "The Juvenile Hormones" (Ed. by Gilbert L.I.) pp.288-300, Plenum Press, New York.
- Yagi S. and Fukaya M. (1974) Juvenile hormone as a key factor regulating larval diapause of the rice stemborer, *Chilo suppressalis* (Lepidoptera: Pyralidae). *Appl. Entomol. Zool.* 9, 247-255.
- Yagi S. and Kuranochi K. (1976) The role of juvenile hormone in larval duration and spermiogenesis in relation to phase variation in the tobacco cutworm Spodoptera litura (Lepidoptera: Noctuidae). Appl. Entomol. Zool. 11, 133-138.
- Yagi S., Kondo E. and Fukaya M. (1969) Hormonal effect on cultivated insect tissues I. Effect of ecdysterone on cultivated testes of diapausing rice stemborer larvae. Appl. Ent. Zool. 4, 70-78.
- Yin C.-M. and Chaw W.-D. (1984) Haemolymph ecdysteroid titre and epidermal cell commitment of the female southwestern corn borer larvae. J. Insect Physiol. 30, 257-263.
- Yin C.M. and Chippendale G.M. (1973a) Endocrine system of mature, diapause and non-diapause larvae of the southwestern comborer Diatraea grandiosella. Ann. Entomol. Soc. Am. 66, 943-947.
- Yin C.M. and Chippendale G.M. (1973b) Juvenile hormone regulation of the larval diapause of the south-western cornborer, *Diatraea grandiosella*. J. Insect Physiol. 19, 2403-2420.
- Zimowska G., Handler A.M. and Cymborowski B. (1985) Cellular events in the prothoracic glands and ecdysteroid titres during the last larval instar of Spodoptera littoralis. J. Insect Physiol. 31, 3311-340.